AGREEMENT BETWEEN

STANFORD HEALTH CARE

AND

COMMITTEE FOR RECOGNITION OF

NURSING ACHIEVEMENT

(CRONA)

May 1, 2022 – March 31, 2025

DocuSign Envelope ID: 6405CD24-BF36-4C1A-A507-FE50D20B458A

CRONA offices are located at:

499 Seaport Court Suite 202 Redwood City, California 94063 Telephone: 650-366-0156 Email: crona@crona.org

TABLE OF CONTENTS

				Page
SECTION 1	RECO	GNITION	AND COVERAGE	1
SECTION 2	NO DIS	CRIMINA	TION	2
SECTION 3	EXCLU	SIVE AG	REEMENT, AMENDMENTS, TERM OF AGREEMENT	2
SECTION 4			MUNICATIONS, REQUESTS FOR INFORMATION AND	2
SECTION 5	CRONA	SECUR	ITY AND DUES DEDUCTIONS	3
5.1	Current	Employe	es	3
5.2	New Er	nployees		3
5.3	Payroll	Deductio	ns of CRONA Dues and Service Fees	4
SECTION 6			N AND COMPENSATION PROCEDURES FOR SES	6
6.1	New Hi	res		6
6.2	Transfe	ers After C	Completion of Trial Period - Regular and Relief	7
6.3	Reclass	sifications		7
	6.3.1	Volunta	ry	7
	6.3.2	Involunt	tary – Non-Transfer	7
6.4	Rehires	s – Regula	ar Nurses	7
SECTION 7	COMPI	ENSATIO	N	8
7.1	Salary	Placemen	t	8
7.2	Part-tim	ne Nurses		10
7.3	Resour	ce Nurse		10
7.4	Work C	utside Nu	Irse Classifications	10
7.5	Shift Di	fferentials		11
	7.5.1	Purpose	э	11
	7.5.2	Shift De	finitions (for Shift Differential Purposes)	11
		a. Ev	/ening Shift	11
		b. Ni	ght Shift	11
	7.5.3	Paymer	nt	11
	7.5.4	Weeker	nd Differential	11
	7.5.5	Part-tim	e Shifts	11
	7.5.6	Combin	ation of Shifts	11
		a. Ev	/ening/Night	11
		b. Da	ay/Evening or Night	11

	7.5.7	Diffe	erential for PTO Hours	12
	7.5.8		tiguous Overtime (immediately preceding or following a Ilar shift)	12
	7.5.9	Nor	-contiguous Overtime	12
7.6	Overtin	ne Col	npensation	12
	7.6.4	Dou	ble Back Premium	13
7.7	On-Cal	l Syste	ems	13
	7.7.1	Res	tricted On-Call	13
	7.7.2	Unr	estricted On-Call	13
	7.7.3	Slee	ep Room	13
7.8	Short N	lotice	Compensation	14
7.9	Staff ar	nd Cor	nmittee Meetings	14
	7.9.1	Staf	f Meetings	14
	7.9.2	Con	nmittee Meetings	15
7.10	Precep	tor Pro	ogram Differential	15
7.11	Certific	ation I	⁵ ay	16
SECTION 8	CLASSIFICATION AND COMPENSATION PROCEDURES FOR TRAINING PROGRAMS			
8.1	Extend	ed Ori	entation Training Programs	17
8.2	Tuition	Based	d Specialty Training Programs	18
SECTION 9	BENEF	ITS		18
9.1	Eligibili	ty		18
9.2	Change	es in E	mployer Provided Benefits	18
	9.2.1	Ann	ual Review of Health Plans	19
	9.2.2	RFF	of Health Plans	19
9.3	Benefit	Plan	Documents	19
	9.3.1	Мес	lical Coverage	19
		e.	Hospitalization	19
	9.3.2	Den	tal Plan	20
		a.	Limits	20
		b.	Optional Coverage	20
		C.	Payment of Premiums	20
		d.	Detailed Information	20
	9.3.3	Visi	on Plan	20
9.4	Long T	erm D	isability Insurance	20

9.5	Basic Li	fe Ins	urance	20	
9.6	Retiree	Medic	al Insurance	21	
9.7	Retiree	Health	n Reimbursement Account	21	
9.8	Back Up	Back Up Care			
9.9	Suppler	nental	Long Term Disability Insurance	21	
9.10	Suppler	nental	Short Term Disability Insurance	21	
9.11	Suppler	nental	Group Life Insurance	21	
9.12	Suppler	nental	Accidental Death and Dismemberment Insurance	22	
9.13	Malprac	tice a	nd General Liability Insurance	22	
9.14	Legal C	are Pl	an	22	
9.15	Flexible	Spen	ding and Health Savings Account Programs	22	
9.16	Employ	ee As	sistance Program	22	
9.17	Mental I	Health	and Wellness Programs	23	
9.18	Nurse F	eedba	ack on Benefit Programs	23	
SECTION 10	PAID TI	ME O	FF (PTO) AND EXTENDED SICK LEAVE (ESL)	23	
10.1	Paid Tin	ne Off	(PTO)	23	
	10.1.1	Purp	ose and Rate of Pay	23	
	10.1.2	Eligi	bility	23	
	10.1.3	Accr	ual	23	
	10.1.4	Use	of PTO	24	
		I.	Cash Out PTO Account	26	
		m.	Voluntary PTO Cash Out	27	
	10.1.5	Holic	lays	27	
10.2	Extende	ed Sicl	<pre>< Leave (ESL)</pre>	28	
	10.2.1	Purp	ose and Rate of Pay	28	
	10.2.2	Eligi	bility	28	
	10.2.3	Αςςι	imulation of ESL	28	
	10.2.4	Use	of ESL	28	
	10.2.5	Integ	ration of ESL	28	
SECTION 11	GOVER		NT PROGRAMS	28	
11.1	State Di	isabilit	y Insurance (SDI)	28	
11.2	Workers	s' Con	pensation	29	
11.3	State U	nempl	oyment Insurance	29	
11.4	COBRA			29	

SECTION 12	EMPLOY	R RETIREMENT PI	_AN	29
12.1	Employe	Retirement Plan		29
	12.1.1	etirement Plan		29
	12.1.3	erms and Condition	S	29
		. Basic Contribut	on	
		. Retirement Mat	ch	
12.2	Tax-Defe	ed Annuity Program		31
SECTION 13	PRE-PL/	EMENT AND ANN	JAL PHYSICAL EXAMINATIONS	31
13.1	Pre-plac	nent and Annual Phy	vsical Examinations	31
SECTION 14	EDUCAT	ONAL ASSISTANCE	AND PROFESSIONAL ENRICHMENT	31
14.1	Purpose			31
14.2	Eligibility			31
14.3	Reimbur	ment		32
	14.3.3	eimbursement Sect	ion	32
	14.3.4	ontinuing Education	Hours	32
14.4	Paid Edu	ational Hours		32
	14.4.2	rocedure		33
	14.4.3	early Basis		
	14.4.4	ccumulation		
	14.4.5	ayment for Educatio	nal Hours	34
	14.4.6	ome Study		34
14.5	Educatio	al Training and Clas	ses Required by the Employer	34
14.6	Shift Diff	ential for Educationa	al Classes	35
14.7	Program	Course, and/or Exan	n Qualifications	35
	14.7.1	riteria		35
14.8	Applicati	Procedure		35
	14.8.1		am Approval (other than Continuing	35
	14.8.2	eimbursement		
14.9	Specialty	Continuing Education	٦	
14.10	Sabbatic	Assignment		
	14.10.1	urpose		
	14.10.2	ligibility and Qualific	ations	37
	14.10.3	erms of Sabbatical	Assignments	37

	14.10.4 Selection	37
	14.10.5 Pay and Benefits	37
	14.10.6 Number of Sabbatical Assignments	37
SECTION 15	RELIEF NURSE PROGRAM	37
15.1	Summary of Responsibilities	37
15.2	New Hires	37
15.3	Qualifications for Employment	38
15.4	Movement from Regular Nurse Employment	38
15.5	Rehires – Relief Nurses	39
15.6	Transfers to Different Units or Areas	39
15.7	Movement to Regular Nurse Employment During the Trial Period	39
15.8	Movement to Regular Nurse Employment After the Trial Period	39
15.10	Reclassifications of Relief Nurses	39
15.11	Performance Evaluation	39
15.12	Employment Commitment and Status	40
15.13	Compensation and Benefits	41
15.14	Scheduling	42
	15.14.9 Educational Assistance	43
SECTION 16	15.14.9 Educational Assistance TEMPORARY REASSIGNMENTS/FLOATING	
SECTION 16 16.2		43
	TEMPORARY REASSIGNMENTS/FLOATING	43 43
16.2	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region	43 43 44
16.2 16.4	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment	43 43 44 45
16.2 16.4 SECTION 17	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment WEEKEND STAFFING	43 43 44 45 45
16.2 16.4 SECTION 17 17.1	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment WEEKEND STAFFING Weekend Defined	43 43 44 45 45 45
16.2 16.4 SECTION 17 17.1 17.2	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment WEEKEND STAFFING Weekend Defined Guarantee of Weekends Off	43 43 44 45 45 45 45 45
16.2 16.4 SECTION 17 17.1 17.2 17.3	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment WEEKEND STAFFING Weekend Defined Guarantee of Weekends Off Exceptions	43 43 44 45 45 45 45 45 45
16.2 16.4 SECTION 17 17.1 17.2 17.3 SECTION 18	TEMPORARY REASSIGNMENTS/FLOATING	43 43 44 45 45 45 45 46 46
16.2 16.4 SECTION 17 17.1 17.2 17.3 SECTION 18 18.1	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment WEEKEND STAFFING Weekend Defined Guarantee of Weekends Off Exceptions SHIFTS AND SHIFT ROTATION – REGULAR NURSES Shift Rotation	43 43 44 45 45 45 45 46 46 46
16.2 16.4 SECTION 17 17.1 17.2 17.3 SECTION 18 18.1 18.2	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment WEEKEND STAFFING Weekend Defined Guarantee of Weekends Off Exceptions SHIFTS AND SHIFT ROTATION – REGULAR NURSES Shift Rotation Regular Shifts	43 43 44 45 45 45 45 46 46 46 46
16.2 16.4 SECTION 17 17.1 17.2 17.3 SECTION 18 18.1 18.2 18.3	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment WEEKEND STAFFING Weekend Defined Guarantee of Weekends Off Exceptions SHIFTS AND SHIFT ROTATION – REGULAR NURSES Shift Rotation Regular Shifts Discussions with CRONA	43 43 44 45 45 45 45 46 46 46 46 46 46
16.2 16.4 SECTION 17 17.1 17.2 17.3 SECTION 18 18.1 18.2 18.3 18.4	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment WEEKEND STAFFING Weekend Defined Guarantee of Weekends Off Exceptions SHIFTS AND SHIFT ROTATION – REGULAR NURSES Shift Rotation Regular Shifts Discussions with CRONA Schedules will be posted for four (4) week periods	43 43 44 45 45 45 45 45 46 46 46 46 46 46
16.2 16.4 SECTION 17 17.1 17.2 17.3 SECTION 18 18.1 18.2 18.3 18.4 18.5	TEMPORARY REASSIGNMENTS/FLOATING Voluntary Commitment - Floating Outside Region Floating to Different Work Locations/Multi-Location Assignment WEEKEND STAFFING Weekend Defined Guarantee of Weekends Off Exceptions SHIFTS AND SHIFT ROTATION – REGULAR NURSES Shift Rotation Regular Shifts Discussions with CRONA Schedules will be posted for four (4) week periods Breaks	43 43 44 45 45 45 45 45 46 46 46 46 46 46 46 46

18.9	End of Established Staffing Patterns	47		
SECTION 19	LEAVES OF ABSENCE – REGULAR NURSES			
19.1	Eligibility			
19.2	Bereavement Leave	48		
19.3	Leave Categories	48		
	19.3.1 Educational Leave	48		
	19.3.2 State and Federal Family and Medical Leave	49		
	19.3.3 Medical Leaves of Absence (excluding pregnancy related disability)	49		
	19.3.4 Pregnancy Related Disability Leave	50		
	19.3.5 Parental Leave of Absence (Non-Disability)	50		
	19.3.6 Personal Leave of Absence	50		
	19.3.7 Military Training Leaves	50		
19.4	Duration of Leave	50		
19.5	Combinations of Leaves of Absences	51		
19.6	Reinstatement Rights	51		
19.7	Working Leave Status	51		
19.8	Procedures	51		
19.12	Return from Leave	52		
	19.12.1 Advance Notice	52		
	19.12.2 Failure to Return	52		
SECTION 20	CHRISTMAS AND NEW YEAR'S HOLIDAY	52		
SECTION 21	JURY DUTY AND LEGAL APPEARANCES	53		
SECTION 22	TIME OFF FOR CRONA OFFICERS	54		
SECTION 23	VACANCIES AND INTERNAL TRANSFERS	55		
SECTION 24	REDUCED COMMITMENTS	56		
24.1	Voluntary Requested Reduction in Scheduled Commitment	56		
SECTION 25	PAYROLL PROCEDURES	57		
25.1	Direct Deposit	57		
25.2	Late and Missing Time Cards	57		
25.3	Correction of Payroll Errors	58		
25.4	Paycheck Questions	58		
SECTION 26	NURSE INFORMATION CENTERS	58		
SECTION 27	PERFORMANCE EVALUATION	58		

SECTIO	N 28	WRITTE	N WA	RNINGS AND DISCIPLINE	59			
2	8.1	Written \	Written Warning (excludes other Disciplinary Notices)					
2	8.2	Tempora	Temporary Relief of Duty Pending Investigation60					
2	8.3	Disciplin	e		61			
		28.3.1	Durir	ng the Trial Period of Employment	61			
		28.3.2	After	the Trial Period of Employment	62			
			a.	Discipline for Cause – Regular Nurses	62			
			b.	Discipline for Cause – Relief Nurses	62			
		28.3.3		nination Primarily for Unsatisfactory Performance – Regular Relief	62			
			a.	Probation for Non-Trial Period Nurses	62			
			b.	Termination for Cause – Regular and Relief	62			
			C.	Opportunity to Transfer After Trial Period – Regular Nurses	63			
			d.	Opportunity to Transfer After Trial Period – Relief Nurses	63			
SECTIO	N 29	GRIEVA	NCE	AND ARBITRATION	63			
2	9.1	Purpose			63			
2	9.2	Grievano	ces		63			
2	9.3	Grievano	ce Pro	ocedure	63			
		29.3.1	Step	One – Informal Discussions	63			
		29.3.2	Step	Two – Submission of Formal Grievance	64			
			a.	Content of Formal Grievance	64			
			b.	Timeliness	64			
		29.3.3	Step	Three – Resolution or Referral for Review	64			
			a.	Review Meeting	64			
			b.	Resolution	64			
		29.3.4	Step	Four (Optional) – Mediation	65			
		29.3.5	Step	Five – Arbitration	65			
			b.	Hearing	65			
			C.	Resolution	66			
			d.	Expense	66			
			e.	Expedited Arbitration	66			
		29.3.6	Righ	t to Representation	66			
			a.	Representation at Grievance Meetings	66			
			b.	Representation at Investigatory Interview	66			

		C.	Representation at Disciplinary Meeting	67
		d.	Definition of Discipline	67
	29.3.7	Adh	erence to Time Limits	67
		a.	Processing of Grievances	67
		b.	Limit of Liability	67
	29.3.8	Grie	vance Pending	67
	29.3.9	Infor	mal Settlement Discussions	68
SECTION 30	NO STF	RIKES	OR LOCKOUT	68
SECTION 31	NOTICE	E OF F	PERSONNEL GUIDELINES AND PROCEDURE CHANGES	68
SECTION 32	JOINT (CONF	ERENCES	68
SECTION 33	NURSE	STAF	FING	69
33.1	Patient	Class	fication System and Committee	69
33.2	Staffing	Ratio	s	70
33.3	Staffing	Plans	5	70
33.4	Dispute	Reso	lution	70
SECTION 34	NURSE	PRA	CTICE COMMITTEE	70
34.1	Compos	sition.		70
34.2	Purpose)		71
34.3	Respon	sibiliti	es	71
34.4	Procedu	ires		71
SECTION 35	SPECIA	L CO	NFERENCES	72
SECTION 36	HEALTH	H, WE	LFARE AND SAFETY	72
SECTION 37	SENIOF	RITY A	ACCRUAL AND APPLICATION	74
37.1	Seniorit	y		74
SECTION 38	STAFF	REDL	JCTIONS AND LAYOFF	75
38.1	Staffing			75
38.2	Tempor	ary St	aff Reductions	75
	38.2.3	"A" ⁻	Гіme Credit	76
38.3	Perman	ent or	Indefinite Layoff	77
	38.3.1	Noti	ce	77
	38.3.2	Sele	ction of Those to be Laid-Off	77
	38.3.3	Plac	ement Alternative to Layoff	77
	38.3.4	Seve	erance Allowance	77
	38.3.5	-	ayment of Severance	70

	38.3.6 Continuation of Benefits	78					
	38.3.7 Re-Employment	78					
	38.3.8 Re-Employment Preference	79					
SECTION 39	HOME HEALTH CARE	79					
SECTION 40	RESERVE RIGHTS OF STANFORD HEALTH CARE	79					
SECTION 41	PROFESSIONAL NURSE DEVELOPMENT PROGRAM	79					
SECTION 42	SEPARABILITY AND SAVINGS CLAUSE	81					
SECTION 43	SUPERVISORY STATUS OF COVERED NURSES	82					
SIGNATURE	PAGE	83					
APPENDIX "A	" WAGE SCHEDULE	84					
GLOSSARY		88					
SIDE LETTER	R Between Stanford Health Care and Committee for Recognition of Nursing Achievement (CRONA) 2010	89					
Α.	Parking	89					
В.	Staffing	89					
C.	Relief Nurses	89					
SIDE LETTER	R Between Stanford Health Care and Committee for Recognition of Nursing Achievement (CRONA) 2010	90					
Α.	Weekend Differential	90					
В.	16-Hours	91					
C.	Vacation and Holiday Time	91					
SIDE LETTER	R Between Stanford Health Care and CRONA Re On Call and Overtime Issues 2016	92					
Α.	On Call Shift Selection/Assignment Procedures	92					
В.	On Call Schedules	92					
C.	Call Back Contiguous With a Scheduled Shift	92					
D.	Mandatory Overtime	93					
E.	Unavoidable Conflicts and Unforeseen Emergencies	93					
SIDE LETTER	R Between Stanford Hospital and CRONA Re Resource Nurse Positions 2004	95					
SIDE LETTER	SIDE LETTER Between Stanford Health Care and CRONA Re Use of Term "Regular" in 2011 – 2013 Agreement						
SIDE LETTER	R Between Stanford Health Care and CRONA Re Use of Travelers 2022	98					
SIDE LETTER	R Between Stanford Health Care and CRONA Re Attendance and Pre- Approved Vacation and Education Days Policies 2022	99					
	R Between Stanford Health Care and CRONA Re 500 Pasteur Dr. 2019	100					

SIDE LETT	ER Between Stanford Health Care And CRONA Re Temporary Relief	101
	Position While Pursuing Education 2019	101
Α.	Purpose	101
В.	Eligibility and Qualifications	101
C.	Number of Temporary Relief Positions	101
D.	Assignment to Temporary Relief Position	101
E.	Selection	101
F.	No Trial Period	101
G.	Return to Open Position	101
Н.	Documentation	101
I.	End of Temporary Relief Nurse Position	102
SIDE LETT	ER Between Stanford Health Care and CRONA Re Student Loan Repayment Program 2022	103
SIDE LETT	ER Between Stanford Health Care and CRONA Re Strategic Recruitmer and Retention Program 2022	
SIDE LETT	ER Between Stanford Health Care and CRONA Re: Transfer Center Nur 2021	
SIDE LETT	ER Between Stanford Health Care and CRONA Re Weekend Premium F	⁵ ay110
INDEX		111

BETWEEN

STANFORD HEALTH CARE AND COMMITTEE FOR RECOGNITION OF NURSING ACHIEVEMENT (CRONA)

MAY 1, 2022 - MARCH 31, 2025

PREAMBLE

This Agreement is entered into by and between Stanford Health Care, hereinafter sometimes referred to as, "Employer", "Hospital", or "Management", and the Committee for Recognition of Nursing Achievement, hereinafter referred to as "CRONA".

SECTION 1 RECOGNITION AND COVERAGE

- 1.1 Stanford Health Care hereby recognizes CRONA as the exclusive collective bargaining representative for purposes of collective bargaining concerning wages, hours and working conditions for Registered Nurses, currently licensed in California and employed by Stanford Health Care in the following classifications: Clinical Nurse I, II, III, and IV; and Relief Clinical Nurse I, II, III, and IV.
- 1.2 The term "Nurses" as used in this Agreement refers to all members of the bargaining unit described in 1.1 unless a provision specifically differentiates between "Regular Nurses" and "Relief Nurses." The term "Clinical Nurses" as used in this Agreement refers to Clinical Nurses I, II, III and IV. The term "Regular Nurses" as used in this Agreement refers to full and part-time Clinical Nurses I, II, III, and IV. The term "Relief Nurses" as used in this Agreement refers to full and refers to Relief Clinical Nurses I, II, III, and IV.
- 1.3 In the event the Employer determines that the content of a Nurse's job covered by this Agreement is such that the job should no longer be in the bargaining unit represented by CRONA, the Employer will notify CRONA in writing. If CRONA does not agree that the job should be removed from the bargaining unit, CRONA will notify the Director of Employee and Labor Relations or designee, within ten (10) calendar days of receipt of the Employer's notice that it wishes to challenge the removal of the job. The Employer shall arrange a meeting to include the Nurse whose job is in question, a CRONA representative and an Employer representative to review the content of the Nurse's job. Except to the extent specified above, removal of positions from the bargaining unit will not be reviewable under this Agreement.
- 1.4 The Employer agrees to notify CRONA in writing of any change in ownership, partnership or merger of Stanford Health Care or a change of the entity that operates Stanford Health Care where work is performed by Nurses covered by this Agreement. If the Employer retains control or majority ownership of the new owner or such changed entity, the terms and conditions of this Agreement shall be applicable to such work. In all other cases, upon CRONA's written request, the Employer will meet with CRONA to discuss CRONA's recommendations and suggestions concerning the effect of the change on CRONA Nurses, with the objective of ensuring the terms of the Agreement are kept in effect. The Employer shall not use any sale, transfer, or other mechanism for the purpose of evading the terms of the Agreement. The Employer is obligated to notify in writing any new owner, buyer or partner of the contract between the Employer and CRONA.

The Employer will notify CRONA in writing as soon as practicable but no later than thirty (30) days prior to any elimination of any unit or area of practice in which Nurses covered by this Agreement are employed. Upon written request, the Employer will meet to discuss any proposed elimination and the effects of such elimination if any, upon the bargaining unit. The Employer shall not eliminate any unit or area of practice for the purpose of evading the terms of the Agreement.

SECTION 2 NO DISCRIMINATION

- 2.1 There shall be no discrimination by the Employer or by CRONA against any Nurse or applicant for a position as a Nurse on account of membership or non-membership in, and activity or non-activity on behalf of CRONA, or on account of sex, age, race, religion, creed, color, national origin, ancestry, physical or mental disability, sexual orientation, and veteran status as provided by law.
- 2.2 The Hospital shall provide at least biennial anti-discrimination and anti-harassment training for each unit or area of practice in which all Nurses, patient care managers, and assistant patient care managers will participate. The training shall address at a minimum (1) implicit bias in the workplace, and (2) effective techniques for bystander intervention when any such conduct is observed in the workplace. The training shall be interactive and may be provided by a third party.

SECTION 3 EXCLUSIVE AGREEMENT, AMENDMENTS, TERM OF AGREEMENT

- 3.1 This Agreement and its appendices and any side letters executed herewith constitute the exclusive record of agreement between CRONA and the Employer on all matters relating to wages, hours, and working conditions.
- 3.2 The Employer and CRONA may mutually agree to amend or add to any provision of this Agreement, its appendices or side letters. Any such amendment or modification must be in writing executed by the duly authorized representatives of each party and any verbal modification or amendment shall be of no force or effect.
- 3.3 This Agreement shall be effective from May 1, 2022 and shall continue in effect up to and including March 31, 2025 and from year to year thereafter, unless, at least ninety (90) days prior to March 31, 2025 or at least ninety (90) days prior to any subsequent anniversary date thereafter, either party gives written notice to the other of its desire to terminate or make changes in the Agreement.

SECTION 4

OFFICIAL COMMUNICATIONS, REQUESTS FOR INFORMATION AND MEETINGS

- 4.1 Any official communications under this Agreement shall be sent to the CRONA office at an address supplied by CRONA as "President, CRONA" and to the Employer as "Director of Employee and Labor Relations", or designee.
- 4.2 To the extent that CRONA may request information or data other than that provided by the terms of this Agreement, all such requests must be in writing by the CRONA President and directed to the Director of Employee and Labor Relations, or designee of the Employer. The Employer shall provide information requested to the extent relevant and necessary to CRONA's representation responsibilities under this Agreement. If all or part of any information requested by CRONA is considered confidential by the Employer, then CRONA

and the Employer will discuss and mutually agree upon ways to safeguard the confidentiality of any information provided. The Employer may charge a fee for those requests which require special or extraordinary processing or staff time.

- 4.3 The Employer shall in good faith attempt to provide monthly to CRONA the following information for each bargaining unit Nurse:
 - 1. Name
 - 2. Mailing Address
 - 3. Employee Identification Number
 - 4. Retirement Plan
 - 5. CRONA Dues Status, including last four digits of Social Security Number
 - 6. Telephone Number, if available
 - 7. Cost Center Assignment
 - 8. Commitment "Regular" or "Relief" and commitment level
 - 9. Status
 - 10. Nurse Level
 - 11. Adjusted Hire Date.
- 4.4 In addition the Employer shall provide CRONA with a monthly list of the names, last four digits of Social Security Number, and dates of employment (including the adjusted hire date if any), classification and cost center assignment of all newly hired bargaining unit Regular and Relief Nurses, and the names of all bargaining unit Nurses who have resigned or been terminated.
- 4.5 The Employer's meeting facilities shall be available to CRONA on an equal basis with other voluntary organizations.

SECTION 5 CRONA SECURITY AND DUES DEDUCTIONS

- 5.1 <u>Current Employees</u>. All Nurses hired before July 1, 1978 may voluntarily remain or become members of CRONA or comply with the provisions of paragraph 5.2.1.a or 5.2.1.b. However, the Employer shall have no obligation to discharge or otherwise discipline any such Nurse who does not desire to do so.
- 5.2 <u>New Employees</u>. It shall be a condition of employment that all Nurses hired on or after July 1, 1978, shall, within thirty (30) days after the beginning date of such employment, either become and remain members of CRONA or comply with the provisions of 5.2.1.
 - 5.2.1 Any Nurse covered by this Agreement who, for personal or religious reasons, does not choose to conform to the membership requirements of 5.2, may petition CRONA in writing within thirty (30) days after the effective date of her/his employment. Such petition shall request, and by such request shall allow the Nurse to cease membership in or not become a member of CRONA and to do either of the following on an annual basis:
 - a. Through payroll deduction or by biweekly payments which correspond to the end of the Employer's pay periods, to remit to CRONA a service fee as a contribution toward the administration of this Agreement in an amount equal to the regular dues of CRONA; or

- b. If such Nurse's choice is based upon religious reasons, to contribute in a manner agreeable to CRONA an amount equal to the regular dues of CRONA to a non-religious charity agreed to by CRONA. CRONA agrees that it will not unreasonably withhold its agreement as to the manner of such payments or the identity of any such charity.
- 5.2.2 Any Nurse who falls within the provisions of 5.2.1 and who fails to comply with the provisions of such paragraph shall, upon notice of such failure in writing from CRONA to the Employer and after counseling of such Nurse by the Employer and CRONA, be discharged or allowed to resign by the Employer no later than fifteen (15) days following receipt of written request from CRONA to terminate such Nurse.

5.3 Payroll Deductions of CRONA Dues and Service Fees

- 5.3.1 If authorized in writing by a Nurse covered by this Agreement, the Employer will deduct from the wages of said Nurse, in the manner and to the extent described below, the sum certified by CRONA as the periodic dues uniformly required by CRONA as a condition of membership, or as the service fee for administration of the Agreement, for the term of this Agreement.
- 5.3.2 No such authorization shall be effective until thirty (30) days after CRONA has filed with the Employer's Human Resources Department the certification which is signed and dated by CRONA's President and Treasurer and states the amount of pay period dues uniformly required as a condition of membership, or service fee for administration of the Agreement. The certification previously filed shall be effective for the duration of this Agreement, unless CRONA files a new certification with the Employer at least thirty (30) calendar days in advance of a change in pay period dues.
- 5.3.3 All pay period deductions pursuant to this Section shall be deducted from each paycheck and shall be paid to CRONA within three (3) working days. Such remittances to CRONA shall be accompanied by pay period itemized statements showing the employee name, the amount of dues deducted, and cost center assignments. The deduction shall not be made for any Nurse whose paycheck for the applicable pay period, after all other mandatory deductions have been made, is less than the sum certified in accordance with 5.3.2 of this Section, and, if such paycheck is less than the sum so certified, the Employer shall have no future obligation to deduct the amount from any succeeding paycheck of the Nurse. A valid authorization, once revoked by a Nurse, may not be reinstated within sixty (60) days of the date on which revocation was received by the Employer.
- 5.3.4 CRONA shall undertake to advise all newly hired Regular and Relief Nurses of their obligations under this Section. The Employer agrees to supply newly hired Regular and Relief Nurses with an informational packet which CRONA shall provide containing a CRONA Dues or Service Fee Deduction Authorization Form and a copy of this Agreement. The Employer at the time of hiring new bargaining unit Nurses shall advise those Nurses that the Employer recognizes CRONA as the exclusive bargaining representative on all matters related to wages, hours and working conditions. The Employer will afford each new bargaining unit Nurse with an opportunity to sign the CRONA Dues or Service Fee Deduction Form at the time the CRONA informational packet is provided to the Nurse.

- 5.3.5 CRONA shall indemnify, defend and save the Employer harmless against any and all claims, demands, suits or other forms of liability that shall arise out of or by reason of action taken or not taken by the Employer in the deduction and remittance of CRONA dues as described in this section.
- 5.3.6 CRONA Membership Dues Or Service Fee Deductions
- To: Stanford Health Care:

For each pay period during which I work for Stanford Health Care while this Authorization is in effect, I hereby direct that, from my earnings now or hereafter payable to me from the Employer, there be paid to CRONA my CRONA membership dues or an equivalent service fee as indicated below, and I hereby authorize and direct you to deduct such sums from my earnings and pay the same for my account to CRONA. You are hereby authorized to deduct such sums from my earnings payable each pay period. I hereby authorize that this deduction of CRONA dues or equivalent service fee be given priority after all State and Federal benefit deductions required by law and medical benefit deductions have been met. The money deducted from my earnings shall be paid by the Employer to CRONA in such a manner as from time to time agreed upon between CRONA and the Employer.

This authorization shall be automatically canceled at such time as I am no longer employed in the bargaining unit represented by CRONA but shall otherwise remain in effect unless I revoke it by sending written notice to the Employer by registered mail, said revocation to become effective thirty (30) days after the revocation is received by the Employer. This authorization is entirely voluntary on my part.

- □ Membership Dues
- □ Service Fees

Print Name

Address

Signature

Last 4 digits of Social Security Number Date

Personal/Home email/address (please print) (Optional)

SECTION 6 CLASSIFICATION AND COMPENSATION PROCEDURES FOR REGULAR NURSES

- 6.1 <u>New Hires</u>.
 - 6.1.1 All newly hired Regular Clinical Nurses will serve a Trial Period of six (6) months, unless extended pursuant to Section 28 of this Agreement except current employees who have already served a trial period. During the trial period or extension thereof a Regular Clinical Nurse may be terminated at any time at the Employer's sole discretion and the Employer's discretion shall not be subject to review under any provision of this Agreement.
 - 6.1.2 Current employees of the Employer who have completed a trial period and are newly hired into a Regular or Relief Clinical Nurse position will serve a six (6) month probation during which s/he can be terminated for just cause. If in the judgment of the Patient Care Manager, the Clinical Nurse is not meeting all the Clinical Nurse criteria at the end of the six (6) months but is making significant progress toward the Clinical Nurse criteria and overall job performance, the Patient Care Manager has the option to extend the probation period for an additional three (3) months.
 - 6.1.3 The Employer will place newly hired Regular Clinical Nurses with six (6) months or less experience at the Clinical Nurse I level. Newly hired Regular Clinical Nurses with more than six (6) months experience will be placed as a Clinical Nurse II at the appropriate step based upon the Employer's assessment of the Nurse's experience, at its sole discretion, provided that:
 - a. Experience must be within the past two (2) years.
 - b. Experience must be applicable to the unit or area(s) in which the Clinical Nurse is hired, as determined by the Patient Care Manager.
 - 6.1.4 After completion of not more than three (3) months of employment, the Employer will give the Regular Clinical Nurse an informal, interim evaluation. If the Clinical Nurse is having difficulty in meeting expectations during this three (3) month period, the evaluation will be in writing. This provision does not limit in any way the right of the Employer to terminate a Regular Clinical Nurse without cause during the trial period; nor does it give rise to any claim concerning such terminations under the Grievance and Arbitration provisions.
 - 6.1.5 If the Regular Clinical Nurse is not performing at the appropriate level after the initial six (6) months of employment, the Employer will determine which of the following actions it deems to be appropriate and act accordingly.
 - a. Extension of trial period pursuant to Section 28 up to a maximum of three (3) months, with postponement of any salary increase. The Regular Clinical Nurse will be evaluated at the end of the extension period and reclassified at the Clinical Nurse II level or if the Employer determines that performance is still unsatisfactory, terminated.
 - b. If the Employer determines that extension of the trial period is not appropriate, the Regular Clinical Nurse will be terminated.

c. The Employer may determine, at its sole discretion, that the Regular Clinical Nurse be transferred (to be restricted to one transfer) to a more suitable position. If a transfer is deemed appropriate, the trial period will be extended pursuant to Section 28 up to a maximum of three (3) months. The Regular Clinical Nurse will be evaluated at the end of the extension period and if the Employer determines that performance is still unsatisfactory, terminated.

6.2 Transfers After Completion of Trial Period - Regular and Relief.

6.2.1 A Nurse of the Hospital who has completed the trial period shall be eligible for transfer to a position represented under this Agreement. Such transfer shall be in accordance with applicable transfer procedures of this Agreement.

A Nurse who transfers to a position represented under this Agreement shall maintain her/his adjusted hire date, and shall be eligible for benefits and other terms in accordance with this Agreement based on the employee's adjusted hire date, if any.

Nothing in this Agreement shall permit the involuntary transfer of a Nurse from a position represented by CRONA to a position not represented by CRONA.

- 6.2.2 Transfers of Clinical Nurse IIIs and IVs will be governed by Section XV of the Professional Nurse Development Program.
- 6.3 <u>Reclassifications</u>.
 - 6.3.1 <u>Voluntary</u>. Regular Clinical Nurse IIIs and Clinical Nurse IVs have the option to request reclassification to a Clinical Nurse II level. Written documentation of this choice is to be placed in the personnel file of the individual Nurse.
 - 6.3.2 <u>Involuntary Non-Transfer</u>. Regular Clinical Nurse IIIs and IVs will be reclassified as provided in the Professional Nurse Development Program Section VIII.

6.4 <u>Rehires – Regular Nurses</u>.

- 6.4.1 If rehired into the same or similar unit within twelve (12) months of resignation, a Regular Nurse will be placed as a Clinical Nurse II level at the step that had been attained at the time of resignation. The Nurse's date of eligibility for the next step increase if rehired into a regular position will be adjusted for the period of resignation, layoff or termination and the Nurse will be eligible for a step after serving twelve (12) months of employment on the step. Eligibility for longevity steps will be according to Appendix "A". To the extent operationally feasible the Nurse may return to the same shift rotation held at the time of resignation. The Employer will determine whether a unit is similar based on its review of the clinical skills and experience required. If hiring into a different unit, the Nurse will be treated as a new hire for purposes of Clinical Nurse level and step.
- 6.4.2 In the case of the rehiring of a Regular Nurse who had been employed as a Regular Clinical Nurse III or IV or Staff Nurse III or IV, the rehired Nurse shall be rehired as a Clinical Nurse II and may reapply for Clinical Nurse III or IV status at the next available application period.

- 6.4.3 If rehired after twelve (12) months from resignation with no intervening employment, a Regular Nurse will be considered as a new hire. The Employer will evaluate any intervening employment to determine appropriate step placement on the Clinical Nurse II salary scale.
- 6.4.4 If a Regular Nurse is rehired into the same unit within two (2) years from resignation, and has had continuous clinically comparable employment as determined by the Patient Care Manager, the Nurse is placed in the step on the Clinical Nurse II salary scale that had been attained at the time of resignation. If hiring into a different unit the Nurse will be treated as a new hire.
- 6.5 Classification and compensation procedures for Relief Nurses are set forth in Section 15 of this agreement.

SECTION 7 COMPENSATION

7.1 Salary Placement.

- 7.1.1 Regular Clinical Nurse I will be hired at the Regular Clinical Nurse I (Step 1) rate. After six (6) months, provided performance is satisfactory, the Regular Clinical Nurse will move to the Regular Clinical Nurse II, step 2 rate. Thereafter, step increases will occur effective the pay period during which the step increase date occurs in accordance with applicable procedures of Section 7.1.2. (Salary step progression for Relief Nurses is specified in Section 15.)
- 7.1.2 Salary step increases for Steps 3 through 6 shall occur each year effective the pay period during which the Regular Nurse's step increase date occurs, providing performance is satisfactory. Regular Nurses will progress to Step 7 upon two (2) years completed at Step 6, provided performance is satisfactory. The Employer will initiate payroll action for salary step increases on a timely basis providing that performance is satisfactory, unless the Nurse has not timely completed her/his responsibilities under performance evaluation procedures, outlined in Section 27.
- 7.1.3 Step increases for Regular Nurses will be implemented as follows:

Step 2: After 6 months at Step 1 (Clinical Nurse I).

Step 3: After 1 year at Step 2.

Step 4: After 1 year at Step 3.

Step 5: After 1 year at Step 4.

Step 6: After 1 year at Step 5.

Step 7: After 2 years at Step 6.

Step 8: After 10.5 years of continuous Employer service (this will not be effective until the Nurse has met the requirements of Step 7).

Step 9: After 15 years of continuous Employer service (this will not be effective until the Nurse has met the requirements of Step 7).

Step 10: After 20 years of continuous Employer service (this will not be effective until the Nurse has met the requirements of Step 7).

Step 11: After 25 years of continuous Employer service (this will not be effective until the Nurse has met the requirements of Step 7).

Step 12: After 30 years of continuous Employer service (this will not be effective until the Nurse has met the requirements of Step 7).

The longevity step increases for Steps 7 through 12 for eligible Nurses shall occur at the beginning of the pay period in which the Adjusted Hire Date falls.

- 7.1.4 There shall be no adjustment of the step increase date as a result of any interim evaluation or change in Regular Nurse classification. Definitions of qualifications of Regular Nurse classifications (experience, educational preparation, ability, etc.) shall be in accord with classification guidelines established by the Employer including the Professional Nurse Development Program for Clinical Nurse III and IV.
- 7.1.5 Prior to revising or establishing new qualifications and Clinical Performance Criteria for Clinical Nurse II, the Employer shall notify CRONA. Upon the request of CRONA, the Nurse Practice Committee shall be given the opportunity to participate in a mutual effort to revise or establish such qualifications or criteria. Disputes regarding revised or new gualifications or criteria that are not resolved within fortyfive (45) days of notification of CRONA may be submitted by CRONA to mediation. pursuant to the terms of Section 29.2.4, but such disputes shall not be subject to arbitration under this Agreement. Mediation shall occur within thirty (30) days of CRONA's submission of the dispute to mediation. Disputes not resolved at the Nurse Practice Committee or submitted to but not resolved in mediation may be resolved by the CNO. Unless agreed to by the Nurse Practice Committee or in mediation, the revised or new qualifications or criteria for Clinical Nurse II shall not be implemented by the CNO earlier than at least forty-five (45) days after the Employer has provided notice to CRONA of such changes if CRONA has not requested mediation during that time, or ninety (90) days after the Employer has provided notice to CRONA of such changes if CRONA has timely requested mediation. The Employer shall make the final determination of the definition of qualifications, recognizing that the Employer holds the ultimate responsibility and accountability for nursing practice and standards.
- 7.1.6 Any Regular Clinical Nurse hired during the term of this Agreement will be initially compensated in accordance with the procedures set forth in Section 6 of this Agreement. Upon completion of one (1) year of employment, such Nurse may be eligible to apply for a Clinical Nurse III or IV position in accordance with the procedures and criteria established in the Professional Nurse Development Program.
- 7.1.7 A Clinical Nurse promoted to the Clinical Nurse II classification will be paid in accordance with the compensation schedule for that classification effective the pay period in which the reclassification was made. The promotion of a Clinical Nurse to the Clinical Nurse III and IV classifications will be governed by the terms of the Professional Nurse Development Program.
- 7.1.8 Procedures for annual performance evaluations are set forth in Section 27 of this Agreement.

- 7.2 <u>Part-time Nurses</u>. Part-time Regular Nurses shall be paid in accordance with the salary schedules in Appendix "A" on an all hours worked basis. Such Regular Nurses may work a portion of their actual commitment in more than one unit with the written approval of the Patient Care Managers of the units involved. In such cases the Personnel Action Form will reflect the total actual commitment worked.
- 7.3 <u>Resource Nurse</u>. Nothing shall restrict the Employer's right to determine when a Resource Nurse should be appointed or to designate Nurses to assume the responsibilities of Resource Nurse.
 - 7.3.1 The Employer shall seek volunteers first, by notifying the Nurses in the unit by email (via Outlook or a similar Hospital-wide email system) of the Employer's decision to appoint an additional Resource Nurse or additional Resource Nurses in the unit. The Employer shall appoint the Resource Nurses from among the volunteers provided that, in the Employer's judgment, an adequate number of the volunteering Nurses possess the requisite skills and abilities to perform the Resource role. Whenever there is an excess of Nurses volunteering for the role who possess the requisite skills and abilities and who, in the determination of the Employer, are equal in their skills and abilities, seniority shall be the tie-breaker in making the selection decision.
 - 7.3.2 If an inadequate number of qualifying Nurses volunteer for the Resource Nurse role, the Manager may appoint Nurses to assume the responsibilities of Resource Nurses. The Manager shall give appropriate consideration to a Nurse's request not to be appointed as a Resource Nurse or to be relieved of Resource Nurse duties for good faith reasons.
 - 7.3.3 The Employer will provide Nurses who are appointed to be a Resource Nurse orientation, education and training, which shall be paid for by the Employer.
 - a. Prior to being assigned to perform the duties of a Resource Nurse, the Nurse will receive precepting from an experienced Resource Nurse for at least three (3) shifts. In addition, beginning on April 1, 2014, the Nurse will be offered a Resource Nurse course prior to being assigned to perform the duties of a Resource Nurse. A newly appointed Resource Nurse will complete an initial competency prior to being assigned Resource Nurse responsibilities and a competency validation after six (6) months of being appointed as a Resource Nurse.
 - b. On an ongoing basis, Resource Nurses will attend regular Resource Nurse meetings. Resource Nurse competency shall be completed annually. Individual training needs will be identified annually by the Manager through completion of the Resource Nurse competency.
 - 7.3.4 Nurses designated as Resource Nurse will be paid in accordance with the salary schedules set forth in Appendix A and will receive in addition a differential of seven and one-half percent (7.5%) per hour for which the Nurse is designated and acts as Resource. Nurses designated and acting in the capacity of Assistant Patient Care Manager (APCM) or Patient Care Manager (PCM) pursuant to 7.4 do not receive the Resource Nurse differential.
- 7.4 <u>Work Outside Nurse Classifications</u>. The Employer has the right to designate Nurses to work temporarily in positions outside the bargaining unit classifications described in Section 1 of

this Agreement. Any Nurse who is designated, and acts as and in lieu of, an APCM or PCM, for a continuous period of fifteen (15) calendar days shall be designated in writing as an APCM or PCM and shall receive a premium of 7.5% of the Nurse's base rate for each shift for which the Nurse is so designated and acts in this capacity. Such written designations shall automatically terminate on the end date specified on the initial designation unless extended in writing. Any temporary designation as an acting APCM or PCM or other non-unit position shall not result in any reclassification of a Nurse or in the loss of any seniority or change of any CRONA benefits during the time such position is held. A Nurse's temporary assignment outside the CRONA bargaining unit shall not otherwise be subject to review under this Agreement.

7.5 <u>Shift Differentials</u>.

- 7.5.1 <u>Purpose</u>. Shift differentials are intended to provide additional compensation for the evening and night shift.
- 7.5.2 Shift Definitions (for Shift Differential Purposes).
 - a. <u>Evening Shift</u>. The evening shift shall include all scheduled shifts where the majority of hours occur after 2:00 p.m.
 - b. <u>Night Shift</u>. The night shift shall include all scheduled shifts where the majority of hours occur after 10:00 p.m.
- 7.5.3 <u>Payment</u>. A differential of ten percent (10%) will be added to the base wage rate for all regularly scheduled "evening" shifts worked; a differential of eighteen percent (18%) will be added to the base wage rate for all regularly scheduled "night" shifts worked.
- 7.5.4 <u>Weekend Differential</u>. Any Nurse assigned to work on a weekend will receive a differential based upon five percent (5%) of the Nurse's base wage rate.
- 7.5.5 <u>Part-time Shifts</u>. The applicable shift differential will be paid for shifts of less than eight (8) hours only if at least half of the hours worked are between the hours of 6:00 PM and 6:00 AM.
- 7.5.6 <u>Combination of Shifts</u>.
 - <u>Evening/Night</u>. If a Nurse works a combination of evening shift and night shift hours, and a majority of the hours are before 10:00 PM, the evening rate will be paid; if a majority of the hours are after 10:00 PM, the night rate applies. (Example: A Nurse works from 9:00 PM to 5:30 AM. The night shift rate applies.) If an equal number of hours is worked both before and after 10:00 PM, the night shift rate applies.
 - b. Day/Evening or Night.
 - (1) If a Nurse works an eight hour shift which is a combination of day shift and either evening or night shift hours, the applicable shift differential will be paid only if at least half of the hours worked are between the hours of 2:00 PM and 6:00 AM.

- (2) If a Nurse works a ten hour shift which is a combination of day shift and either evening or night shift hours, the applicable shift differential will be paid only if at least half of the hours worked are between the hours of 3:00 PM and 7:00 AM.
- (3) If a Nurse works a twelve hour shift which is a combination of day shift and either evening or night shift hours, the applicable shift differential will be paid only if at least half of the hours worked are between the hours of 3:45 PM and 7:15 AM.
- 7.5.7 <u>Differential for PTO Hours</u>. A differential of ten percent (10%) or eighteen percent (18%) respectively will be paid to Regular Nurses claiming PTO if they are permanently assigned to evenings or nights. Permanent assignment constitutes an anticipated or realized six (6) months of assignment. This assignment is agreed upon by the Regular Nurse and department and indicated on a signed Personnel Action Form.
- 7.5.8 <u>Contiguous Overtime (immediately preceding or following a regular shift)</u>. A Nurse working overtime hours which are contiguous to a regular shift will receive the higher shift differential of the two (2) shifts worked. A Nurse working hours in accordance with the Short Notice Compensation provision which are contiguous to a regular shift, will receive the highest shift differential for the Short Notice Compensation hours worked.
- 7.5.9 <u>Non-contiguous Overtime</u>. A Nurse working overtime hours which are not contiguous to a regular shift will be paid shift differential at the rate applicable to the shift in which the overtime hours are worked.
- 7.6 Overtime Compensation.
 - 7.6.1 Except as otherwise provided in this Agreement, Nurses will be paid overtime as follows:
 - a. For Nurses regularly assigned to work twelve (12) hour shifts, overtime will be paid for all hours worked in excess of twelve (12) in the work day at the rate of two times (2x) the Nurse's regular rate;
 - b. For Nurses regularly assigned to work shifts of eight (8) or more hours, but less than twelve (12) hours, overtime premium will be paid for all hours worked in the work day in excess of the regularly scheduled shift (e.g. 8, 9, 10, or 11) up to twelve (12) total hours worked at the rate of one and one-half (1-1/2) times the Nurse's regular rate, and for all hours worked in the work day in excess of twelve (12) hours at the rate of two times (2x) the Nurse's regular rate;
 - c. Nurses will also be paid overtime premium for all hours worked in excess of forty (40) hours in the workweek, or, in the case of Nurses, who regularly work eight (8) hour shifts pursuant to the 8/80 pay plan, for all hours worked in excess of eighty (80) hours in the fourteen (14) day work period. Unless the parties agree otherwise, the fourteen (14) day work period for Nurses working the 8/80 pay plan shall coincide with the pay period.

- 7.6.2 For purposes of overtime calculation, hours worked will include time actually worked, but will not include any time for which the time and one-half (1-1/2) premium or the double time premium has been paid.
- 7.6.3 The Employer will make reasonable efforts to secure volunteers from the staff on the premises, including Nurses not covered by this Agreement, and from resources outside the hospital, to replace a Nurse who is unable to work overtime.

A Nurse who is too fatigued or ill to perform his/her duties safely will inform the Manager and be relieved of duty.

7.6.4 <u>Double Back Premium</u>. Any Nurse assigned to rotating shifts who is scheduled, unless by her/his request, to work within eight (8) hours after the end of a scheduled shift shall be paid overtime rates for the entire second shift.

When a Nurse is required to stay involuntarily past the scheduled end of her/his shift due to the Employer's business, and if the Nurse does not receive a break of at least eight (8) hours, any shift begun during this eight (8) hour period will be paid at the applicable overtime rate for the entire shift.

- 7.6.5 Except as explicitly stated in 7.6.1, nothing in this Agreement shall be applied to require compensation greater than one and one-half (1-1/2) times a Nurse's regular hourly rate of pay for any period worked.
- 7.7 <u>On-Call Systems</u>. There are two types of On-Call, titled "Restricted" and "Unrestricted".
 - 7.7.1 <u>Restricted On-Call</u>. This type of On-Call is considered to be hours worked and will be compensated accordingly. The Nurse is required to remain on-call on the Employer's premises or at some designated place (within a restricted geographical area.)
 - 7.7.2 <u>Unrestricted On-Call</u>. This type of On-Call is not considered to be hours worked. The Nurse is not required to remain on the Employer premises or any other particular place, but is required to leave word where s/he may be reached by telephone. The Employer will compensate a Nurse on unrestricted on-call at onehalf the hourly base wage rate for scheduled unrestricted on-call hours. If a Nurse is called into work while on unrestricted on-call status, s/he will be paid at the rate of one and one-half (1-1/2) times the regular hourly rate of pay for actual hours worked and will be guaranteed a minimum of two (2) hours pay at this rate, unless, with supervision's approval the Nurse wishes to return to her/his home to resume on-call status. A Nurse called in to work under this paragraph will be compensated according to subsection 7.6 for any overtime hours worked.
 - 7.7.3 <u>Sleep Room</u>. The Employer shall maintain a list of a sufficient number of appropriate locations to sleep, which may include off-site lodging paid for by the Employer. The Patient Care Manager, Administrative Nursing Supervisor, or their designee will identify from that list a location to sleep for each Nurse placed on-call who has worked a minimum of twelve (12) consecutive hours or who has less than eight (8) hours before her or his next scheduled shift begins.

From time to time, the Nurse Practice Committee shall meet and review the list of pre-designated appropriate sleep locations to discuss potential modifications or expansions to the list.

- 7.7.4 Effective August 28, 2016, a Nurse may not be assigned call in an amount exceeding twelve (12) hours per week.
- 7.8 <u>Short Notice Compensation</u>. To meet a staffing need a Nurse may be called on short notice to report for work. If the Nurse called for work is not in an on-call status on an established on-call system, the following shall apply, except in situations where a Nurse previously scheduled to work and canceled has been called in to work within one (1) hour of the initial cancellation.
 - 7.8.1 A Nurse requested to report to work prior to the normal starting time of her/his assigned shift will be paid one and one-half (1-1/2) times the regular hourly rate for all early hours worked. Once the Nurse's regularly scheduled shift begins, the Nurse will be paid at the regular hourly rate for the first eight (8) hours (excluding the early hours), one and a half (1-1/2) times the regular hourly rate for hours in excess of eight (8) but less than twelve (12), and double the regular hourly rate for hours in excess of twelve (12) in that workday.
 - a. A Nurse who is scheduled for and works a twelve (12) hour shift who is requested to work prior to the normal starting time of her/his assigned shift will be paid at double time for all early hours worked. A Nurse working overtime contiguous to a regular shift will receive the higher shift differential of the two (2) shifts worked for the overtime hours. (Cross reference section 18.6.7)
 - 7.8.2 A Nurse requested to report back to work on a day in which he or she has already worked will be provided one and one-half (1-1/2) times the regular hourly rate for each hour in that workday in excess of eight (8) but less than twelve (12), and double the regular hourly rate for each hour in excess of twelve (12). When the Nurse's twenty-four (24) hour work day ends a Nurse who continues to work a regularly scheduled shift will be paid at the regular hourly rate regardless of overtime hours worked in the previous workday.
 - 7.8.3 A Nurse requested to report back to work on a scheduled day off, without receiving at least twenty-four (24) hours advance notice, will be paid one and a half (1-1/2) times the regular hourly rate of pay, with a guaranteed minimum of four (4) hours, for all hours through the twelfth (12th) hour, and double the Nurse's regular hourly rate of pay for all hours in excess of twelve in the workday. When the Nurse's regular scheduled workday begins, the Nurse will be paid at regular rates, regardless of the number of overtime hours worked the previous day except as otherwise provided in section 7.6.4.
 - a. A Nurse requested to report back to work on a scheduled day off by prearrangement (more than twenty-four (24) hours notice) is compensated at regular hourly rates unless overtimes rates are applicable because the Nurse is working more than eighty (80) hours in a pay period or eight (8) hours in a day.

7.9 <u>Staff and Committee Meetings</u>.

7.9.1 <u>Staff Meetings</u>. Nurses will be paid for attending staff meetings outside of the Nurse's regularly scheduled shift. If the meeting is contiguous with the Nurse's shift, the pay rate will include the Nurse's regular shift differential, if any, and overtime premium, if applicable.

7.9.2 <u>Committee Meetings</u>. Nurses will be paid for time spent attending and performing required duties as a member of a committee created, recognized and sponsored by the Employer. The Employer and the Nurse must agree on the appointment to, or participation in, such a committee and the extent to which the Nurse is authorized to perform duties other than attendance at the meeting. If the meeting is contiguous with the Nurse's shift, the pay rate will include applicable shift differential, if any, and overtime premium, if applicable.

7.10 Preceptor Program Differential.

- 7.10.1 A Nurse assigned by the Employer as a Preceptor pursuant to this section will be paid a premium of three dollars (\$3.00) per hour for all hours worked as an assigned Preceptor. To be eligible for this differential, a Preceptor must have been assigned in writing by the Preceptor's manager or designee pursuant to this Section to function in the role for a specific Nurse.
- 7.10.2 For purposes of this section, a Preceptor is an experienced nurse assigned to function as a role model, teacher, and evaluator for a specific nurse. During the hours the Preceptor is so assigned, the nurse receiving training from the Preceptor is not considered in the count in the unit under applicable staffing laws and rules.
- 7.10.3 For purposes of this section, a Preceptor is responsible for planning, organizing, and evaluating the knowledge and skill development of a nurse in a formalized training program such as, but not limited to, the New Graduate/New Resident Training Program and the Specialty Training Programs (e.g. Critical Care, OR, ED, AAU). In addition, a Preceptor may be assigned to provide formalized training for newly hired experienced Nurses and for current Nurses who need specific and formalized training in connection with a relocation to a different unit of the Hospital or to a different role within the same unit. A Preceptor may also be assigned to precept a Nurse in the same unit who is learning a new specialty skill, such as advanced surgical procedure, ECMO, CRRT, IABP, and VAD.
- 7.10.4 The length of a formalized training program and the amount of formalized training shall be determined by the Employer.
- 7.10.5 The decision of a Nurse to become eligible for Preceptor assignments is voluntary. A Nurse who has volunteered to act as a Preceptor shall receive education and training for the role, complete an initial competency validation and thereafter may be assigned Preceptor duties so long as the Nurse's competency is current.
- 7.10.6 The Employer shall provide a Nurse the opportunity to complete a Preceptor Training Program prior to serving as a Preceptor. The Employer will pay Nurses for the time spent and expenses incurred in participating in the Preceptor Training Program and will use best efforts to provide Nurses an opportunity to participate in the Preceptor Training Program within the Nurse's regular commitment.
- 7.10.7 Preceptor competency shall be completed annually. Individual training needs will be identified annually by the Manager or designee through completion of the Preceptor competency.
- 7.10.8 A qualified Preceptor may request to be removed from the Preceptor program or to be relieved temporarily of Preceptor duties by making a request to her or his Manager and such request shall not be unreasonably denied, taking into

consideration the staffing and training needs of the unit and patient care. A manager who denies a request shall provide an estimate in writing of the time period by which the Nurse's request may be granted.

- 7.10.9 Each unit will maintain a list of qualified preceptors. In the event that a unit does not or will not have a sufficient number of qualified Preceptors, the matter will be discussed at Nurse Practice Committee. CRONA and the Employer agree to work collaboratively to identify Nurses who are interested in becoming Preceptors.
- 7.10.10 A Nurse assigned to orient a Traveler Nurse shall, for purposes of Section 7.10 only, be deemed to have been assigned to precept the Traveler Nurse while the Traveler Nurse is not in the count under applicable staffing laws and rules.
- 7.10.11 Pursuant to Section 7.10.10, a Nurse assigned to orient a Traveler Nurse shall be eligible to receive the precepting differential (i.e., \$3.00 per hour) for all hours orienting the Traveler Nurse while the Traveler Nurse is not in the count under applicable staffing laws and rules.

7.11 Certification Pay

- 7.11.1 A Nurse who possesses or obtains a national certification or recertification that is either the most applicable certification in the Nurse's area of specialty or a certification that supports the basic clinical practice in the Nurse's area of work shall receive a semiannual payment of seven hundred fifty dollars (\$750) (equal to one thousand five hundred dollars (\$1,500) annually), subject to the provisions below:
 - a. To be eligible for the incentive payment, the certification or recertification must be one that has been approved by the ANCC or one which the Employer's Chief Nursing Officer, in her or his discretion, has approved and added to a list of approved certifications. CRONA's input will be considered before any such determination is made. Non-ANCC-approved certifications must reflect a discrete body of knowledge and skills with respect to a particular subject matter or scope of practice, and substantiate advanced contributions to nursing practice.
 - b. To be eligible for an incentive payment, the Nurse must upload into the Healthstream or equivalent system proof of a valid certification, which shall be subject to verification, at least thirty (30) days prior to the beginning of the month in which the payment shall be made, except that a valid certification that has been entered into the Healthstream or equivalent system and has been verified, does not need to be re-entered in subsequent periods so long as the certification remains valid. The Employer shall rely on the uploaded certification documents that are submitted by the deadline for purposes of validating certifications or recertifications for an incentive payment.
 - c. The first semiannual payment shall be made in the first full pay period of October 2022. Thereafter, the payments shall be made in the first full pay period of the month every six months.
 - d. Regardless of the number of certifications or recertifications obtained or held by a Nurse, the Nurse will not receive more than the annual amount of one thousand five hundred dollars (\$1,500).

SECTION 8 CLASSIFICATION AND COMPENSATION PROCEDURES FOR TRAINING PROGRAMS

The Employer supports job oriented training and professional enrichment through the provision of selected Employer programs offering expanded specialty training opportunities. The Employer has the responsibility to make the final determination of the need for and definition of such programs at its sole discretion.

8.1 Extended Orientation Training Programs.

- 8.1.1 To the extent the Employer provides in house Extended Orientation Training Programs in certain specialty areas participation is open, subject to enrollment limitations, to all current Regular Nurses of the Employer who have successfully completed their trial period and to other external Registered Nurse applicants deemed qualified by the Employer. Successful applicants will be eligible to fill available positions within the specialty area upon successful completion of the program.
 - a. Current Regular Nurses of the Employer who apply and are selected for an Extended Orientation Training Program will continue to be compensated at the Nurse level and step attained at the time of entry into the program. External Registered Nurse applicants who are selected for such programs will be hired into the program at the Regular Clinical Nurse II level except in the O.R. where external participants may be hired at the Regular Clinical Nurse I level.
 - b. The terms and conditions for completion of Extended Orientation Training Programs are set at the sole discretion of the Employer and not reviewable under the terms of this Agreement. Following successful completion of the program the Nurse will be evaluated by the Employer to determine appropriate Nurse level within the specialty area. A non-trial period Nurse will be given up to a maximum of six (6) months (including the time spent in the training program) to meet the Clinical Performance Criteria for the new specialty area. If the Nurse does not meet the criteria at the end of the six (6) month period s/he will be reclassified downward to the appropriate Nurse level. Salary decrease will be processed as necessary three (3) months from date of reclassification. The Patient Care Manager has the option to extend the above referenced period for any period not to exceed an additional six (6) months if, in her/his judgment, the Nurse is making significant progress toward the clinical performance criteria and overall job performance. Extension under these circumstances shall not be unreasonably withheld. If, at the end of the extended period the Nurse is not meeting the performance level, reclassification downward will occur to the appropriate Nurse level with salary decrease processed immediately as necessary. A downward reclassification is not considered to be an involuntary reclassification under Section 6.3.2 of the Agreement.
- 8.1.2 Current Nurses who apply to Extended Orientation Training Programs will be given preference, if qualified, over external applicants unless in the judgment of the Employer the external applicant is better qualified. Selection for such training programs is not otherwise reviewable under the terms of this Agreement.

- 8.1.3 A current Regular Nurse accepted into an Extended Orientation Training Program will maintain her/his appropriate Employer hire date and will suffer no loss of related Nurse benefits during the period of time in the program.
- 8.2 <u>Tuition Based Specialty Training Programs</u>. To the extent the Employer provides Tuition Based Specialty Training Programs, participation in such programs is open, subject to enrollment limitations, to all qualified Registered Regular Nurse applicants.
 - 8.2.1 Current Regular Nurses who apply to Tuition Based Programs will be given preference, if qualified, over external applicants unless in the judgment of the Employer the external applicant is better qualified. Selection for such training programs is not otherwise reviewable under the terms of this Agreement.
 - 8.2.2 Participants in Tuition Based Programs will be considered student/trainees and not employees of Stanford Health Care during the time spent in the program and as such not included in the bargaining unit represented by CRONA. The terms and conditions for completion of such programs are set at the sole discretion of the Employer and not reviewable under the terms of this Agreement.
 - 8.2.3 Regular Nurses accepted into such programs will be considered to be on an approved Educational Leave of Absence as defined in Section 19 of this Agreement for the period of time spent in the program.
 - 8.2.4 Successful graduates of tuition based training programs are eligible for hire into the specialty area covered by the program to the extent vacancies exist. External applicant program graduates will be considered new hires and evaluated as such pursuant to Section 6 of the Agreement.
 - 8.2.5 Regular Nurses successfully completing the program will be qualified over external applicant graduates unless, in the judgment of the Employer the external applicant is better qualified. Nurses will be evaluated as transfers pursuant to Article 6.2 of the Agreement. Selection for hire is not otherwise reviewable under the terms of this Agreement.

SECTION 9 BENEFITS

- 9.1 <u>Eligibility</u>. All Nurses covered by this Agreement are eligible to participate in the benefit programs enumerated in this Section as the terms for the programs are set forth below, provided that the Nurse meets the eligibility requirements for each plan. The Employer may alter the criteria for eligibility provided that no Nurse who is eligible for a benefit upon the execution of this Agreement becomes ineligible because of the changes. Specific eligibility requirements for Relief Nurses are set forth in Section 15 of this Agreement.
- 9.2 <u>Changes in Employer Provided Benefits</u>. The Employer shall have the right to alter the benefits or carrier of any of the plans enumerated in this Section. Sixty (60) days prior to the implementation of any such changes the Employer shall notify CRONA of any such proposed changes and shall provide, in Excel or a similar electronic format, to CRONA relevant information that is not confidential or privileged upon which the Employer has based its decision to make the proposed changes. Upon CRONA's request, the Employer shall meet with CRONA to review the proposed changes in an attempt to reach mutual agreement regarding modifications. If mutual agreement is not reached, the Employer shall maintain substantially equivalent benefits and plans as enumerated hereunder during the term of this

Agreement. In the event the Employer makes changes which include improvements for unrepresented employees in the benefits described in Section 9 which are common to both the CRONA unit and such unrepresented employees, CRONA will be offered the opportunity to accept those changes; provided that CRONA's acceptance is given in writing and is based on the changes in their entirety.

- 9.2.1 <u>Annual Review of Health Plans</u>. The Employer shall meet annually with CRONA at least ninety (90) days prior to plan renewal to discuss problems regarding claims, administration and access issues of the health plan.
- 9.2.2 <u>RFP of Health Plans</u>. If the health plan is put out for bid, the Employer agrees to meet with CRONA to review the criteria to be placed in the Request For Proposal. In addition, the Employer agrees to meet with CRONA prior to the selection of a carrier. In advance of the meeting regarding the selection of a carrier, the Employer shall provide, in Excel or a similar electronic format, to CRONA relevant information that is not confidential or privileged considered by the Employer as a basis for carrier selection.
- 9.3 <u>Benefit Plan Documents</u>. The Employer provides the following benefits to all Regular Nurses who are eligible in accordance with the terms of this Agreement. The various plans are summarized herein. Eligibility and coverage available for Relief Nurses is specified in Section 15 of this Agreement. CRONA understands and agrees that the descriptions below do not purport to recite completely the coverage or eligibility requirements for the plan, the details of which are subject to periodic review and modification by the Employer.
 - 9.3.1 <u>Medical Coverage</u>.
 - a. The Employer shall provide as one option a basic and major medical plan with no charge for self and children coverage, which it may change from time to time, provided that such changes shall not involve distinction based upon membership or non- membership in CRONA. This coverage is provided to all Nurses covered by this Agreement except that specific eligibility requirements for Relief Nurses are set forth in Section 15.
 - b. For Health insurance plans currently in effect, increases in the employee's contribution cannot be greater than the percentage increase in the premium of that plan.
 - c. Unless precluded by law, the Employer shall provide at least one plan option for self, spouse/eligible domestic partner, children, and family coverage that includes a Health Savings Account ("HSA") or other similar account that is owned by the enrolled Nurse and rolls over from year to year (the "HSA Plan"). The HSA Plan shall have no more than the minimum deductible(s) required by law.
 - d. For all medical plans, coverage shall remain in effect until the last day of the month in which employment terminates or in which the employee is no longer eligible for coverage.
 - e. <u>Hospitalization</u>: Hospital charges (excluding physician charges) for employees covered by the PPO will be waived if a plan participant is hospitalized at SHC and/or LPCH. Hospitalization includes same day surgery and ATU. This does

not include charges when the plan participant is not hospitalized, e.g., outpatient charges for MRI.

- 9.3.2 <u>Dental Plan</u>. Eligible Nurses and their dependents will be covered by a dental plan in accordance with the following. When both parents are employed by Stanford Health Care, dependent children may be covered under each parent's dental plan.
 - a. <u>Limits</u>.
 - (1) Orthodontic services are provided only to eligible dependent children and limited to fifty percent (50%) coverage and a maximum reimbursement of One Thousand Dollars (\$1,000.00) per child.
 - (2) Other dental services are provided to eighty percent (80%) with a maximum dollar amount paid by the plan per calendar year of One Thousand Five Hundred Dollars (\$1,500.00) per individual covered or as otherwise covered by the plan summary document.
 - b. <u>Optional Coverage</u>. The Nurse's spouse/domestic partner may be enrolled in the plan at group rates, at the option of the Nurse. The spouse/domestic partner must be enrolled within one month of the Nurse's eligibility or enrollment may be delayed to dates established by the insurance carrier.
 - c. Payment of Premiums.
 - (1) Premiums for the dental insurance provided are paid by the Employer for the covered Nurse and all eligible dependent children enrolled in the program.
 - (2) Premiums for spouse/domestic partner coverage elected by the Nurse shall be paid through automatic paycheck deductions upon the employee's written authorization.
 - d. <u>Detailed Information</u>. Plan details shall be maintained in the Human Resources Department and made available to employees upon request.
- 9.3.3 <u>Vision Plan</u>. The Employer shall provide a vision care plan.
- 9.4 <u>Long Term Disability Insurance</u>. All Regular Nurses are covered under the Employer's Long Term Disability Insurance Plan that is provided to non-represented employees. Relief Nurses will be entitled to coverage as specified in Section 15.

The Employer will pay the full cost of providing long-term disability insurance for Regular Nurses. Coverage begins on the first day of the month.

The plan pays fifty percent (50%) of the eligible Nurse's monthly base wage (to a limit of Eight Thousand Dollars (\$8,000)), reduced by other available income benefits, commencing at the end of a six (6) month qualifying period of continuous disability. Benefit payments continue for the duration of total disability to age 65.

9.5 <u>Basic Life Insurance</u>. The Employer pays the total premium cost to provide all Regular Nurses and eligible Relief Nurses as specified in Section 15 with one (1) times annual salary to a maximum of Fifty Thousand Dollars (\$50,000.00) of Group Life Insurance. If the Nurse

enrolls at the time of hire or when newly eligible, coverage becomes effective on the first (1st) day of the month following that date.

- 9.6 <u>Retiree Medical Insurance</u>. Benefit eligible Nurses will be able to participate in the Employer's Retiree Medical Plan, based on their age and years of service (adjusted hire date) as described under the terms of the plan. Effective January 1, 2017, any benefit eligible nurse who is a Group B or Group C retiree will have a one-time opportunity to elect the Group D Retiree Health Reimbursement Account benefit described in Section 9.7 in lieu of such Group B or Group C benefit. Failure to so elect such Group D benefit when eligible to do so will result in the retiree receiving the Group B or Group C benefit, as applicable, to which they are entitled under the Plan.
- 9.7 <u>Retiree Health Reimbursement Account</u>. Nurses will participate in the Employer's Retiree Health Reimbursement Account based on their age and years of service (adjusted hire date) as described under the terms of the plan. For Nurses eligible to participate in the RHRA:
 - 9.7.1 Effective January 1, 2023, the Group D Retiree Health Reimbursement Account benefit will be increased by fifty percent (50%), except that the Group D benefit shall be increased by seventy-five percent (75%) for Nurses who retire at ages 55 through 64 (before age 65) and who have at least 25 years of continuous service as defined in the Retiree Medical Plan as of the date of the Nurse's retirement.
- 9.8 <u>Back Up Care</u>. Effective August 5, 2011, Regular Nurses will be eligible to utilize the Employer's Back Up Care Program in accordance with its terms and provisions.
- 9.9 <u>Supplemental Long Term Disability Insurance</u>. Regular Nurses and eligible Relief Nurses as specified in Section 15 are eligible to purchase, through payroll deduction, Supplemental Long Term Disability Insurance within 31 days of hire or within 31 days of becoming eligible. The plan increases the percentage of pay replaced, described in Section 15 up to sixty-six and two-thirds (66-2/3%) percent, as described under the terms of the plan. If the Nurse wishes to enroll after the thirty-one (31) day period, a statement of Physical Condition Form is required and must be approved by the insurance company. If approved, coverage begins on the date determined by the insurance company and appropriate payroll deductions will be made.
- 9.10 Supplemental Short Term Disability Insurance. Regular Nurses and eligible Relief Nurses as specified in Section 15 are eligible to purchase supplemental short term disability insurance within 31 days of hire. The short term disability plan pays sixty (60%) percent of the pre-disability base wage rate to a maximum weekly benefit of \$1,846 less disability payments from any state-mandated disability plan and Social Security, less any other employment earnings. Effective January 1, 2023, the maximum weekly benefit shall be increased to \$3,000, subject to the same terms and conditions as the previous maximum weekly benefit. The benefits begin eight (8) days after the Nurse becomes disabled and ends when the Nurse is no longer disabled or after 52 weeks, whichever comes first.
- 9.11 <u>Supplemental Group Life Insurance</u>. Eligible Regular Nurses and eligible Relief Nurses may purchase Supplemental Group Life Insurance in accordance with the applicable plan provisions contained in the Staff Benefits Handbook and/or on file in the Benefits Office.

New eligible Nurses are given the opportunity to enroll during orientation or when they are newly eligible for the benefit. If the eligible Nurse enrolls at that time, coverage becomes effective on the first (1st) day of the month following date of hire or change in status. If the eligible Nurse wishes to enroll after the thirty one (31) day period, a Statement of Physical

Condition Form is required and must be approved by the insurance company. If approved, coverage begins on the date determined by the insurance company and appropriate payroll deductions will be made.

Eligible Nurses may purchase up to six (6) times their annual base salary (to a maximum of \$3,000,000 combined with basic life). Evidence of good health is required for amounts over three (3) times annual salary. After the initial thirty-one (31) day enrollment period, the Nurse may enroll or change supplemental life insurance at any time by completing a regular enrollment/change form and meeting all requirements for changing coverage including providing evidence of good health. If the Nurse has supplemental life insurance, he/she may purchase insurance for spouse and/or children subject to any insurance company requirements including evidence of good health.

9.12 <u>Supplemental Accidental Death and Dismemberment Insurance</u>. Eligible Regular and Relief Nurses may purchase Supplemental Accidental Death and Dismemberment Insurance, through payroll deduction, for themselves only or for themselves, spouse and dependent children. Nurses may elect to purchase one of a number of plans, with the principal sum ranging from Ten Thousand Dollars (\$10,000.00) to Five Hundred Thousand Dollars (\$500,000.00) or ten (10) times base annual salary, whichever is lower. The total premium is paid by the Nurse through payroll deduction.

New eligible Nurses are given the opportunity to enroll during orientation. A Regular Nurse electing not to participate at that time may enroll for coverage, to be effective the first day of any month, by submitting the application during the previous month.

- 9.13 <u>Malpractice and General Liability Insurance</u>. The Employer pays the total cost of a group General Liability insurance policy which automatically covers all Nurses who are acting within the scope of their professional duties.
- 9.14 <u>Legal Care Plan</u>. Eligible Nurses may purchase a legal care plan through appropriate payroll deduction.
- 9.15 <u>Flexible Spending and Health Savings Account Programs</u>. The Employer will provide the opportunity for Regular Nurses and "C" and "D" Relief Nurses to use pre-tax earnings for payment of health care premiums and to place pre-tax earnings into the following reimbursement accounts up to the maximum allowed by law and in accordance with the Employer's summary plan document, applicable law, and IRS regulations for these programs:
 - 1. Premiums for health benefits.
 - 2. Dependent day care expenses.
 - 3. IRS allowed medical spending account (effective beginning of 1999 plan year).
 - 4. Health Savings Account Programs (if Employee is eligible to contribute under Internal Revenue Code and IRS guidance, the maximum allowable annual Employee contributions will be reduced by any Employer contributions).
- 9.16 Employee Assistance Program.
 - 9.16.1 The Employer shall provide Nurses access to the Employee Assistance Program (EAP) offered to the Employer's nonrepresented employees. The EAP shall include

confidential mental health counseling appointments at no cost to Nurses and the availability of crisis counseling.

- 9.16.2 Nurses shall have access to the Stanford Faculty Staff Help Center to seek mental health counseling appointments and crisis counseling, as long as this center remains available to the Employer's employees.
- 9.16.3 The Employer shall consult with CRONA regarding the results of the Request for Proposal for the EAP vendor effective on January 1, 2023, before the vendor is selected by the Employer. The Employer shall treat RFPs for the EAP vendor after January 1, 2023, as covered by Section 9.2.2.
- 9.17 <u>Mental Health and Wellness Programs</u>: The Employer shall maintain a centralized online resource that informs Nurses of programs made available through the Employer to support the mental health and wellness of Nurses.
- 9.18 <u>Nurse Feedback on Benefit Programs</u>. The Employer shall provide a means by which Nurses can provide feedback at any time on any of the benefit programs offered by the Employer, including but not limited to the EAP. The feedback received by the Employer from Nurses shall be provided to CRONA upon CRONA's request and discussed during Joint Conference.

SECTION 10 PAID TIME OFF (PTO) AND EXTENDED SICK LEAVE (ESL)

- 10.1 Paid Time Off (PTO).
 - 10.1.1 <u>Purpose and Rate of Pay</u>. Paid Time Off (PTO) allows more flexibility and individual management of time off subject to the Employer's reserved right to determine scheduling and ensure the continuity of patient care. PTO compensates Regular Nurses at their hourly base wage rate of pay when they are absent from work for such purposes as vacation, illness, holidays, family emergencies, religious observances, preventive health and dental care, and other excused elective absences. A Regular Nurse permanently assigned to evenings or nights at the time of payment will receive a differential of ten percent (10%) or eighteen percent (18%), respectively, based upon the hourly base wage rate. Permanent assignment constitutes an anticipated or realized six (6) months of assignment. This assignment is agreed upon by the Nurse and department and indicated on a signed Personnel Action Form.
 - 10.1.2 <u>Eligibility</u>. All Regular Nurses (excludes Relief) who are categorized as regular or trial period whether full-time or part-time are eligible for PTO.
 - 10.1.3 <u>Accrual</u>.
 - a. PTO is accrued on the basis of paid hours, on a maximum of eighty (80) hours in a pay period, excluding missed meal period penalties, PTO cashouts and on call pay. As an exception, eight (8) hour double shifts worked in a workday may result in an accrual of PTO over eighty (80) hours in a pay period.

- b. PTO accrual will continue for all hours off on Jury Duty and legal appearances as defined in Section 21, for hours the Regular Nurse would normally have been scheduled to work.
- c. The following accrual rates are illustrative for Regular Nurses who work full time.

Accrual Rates for Full-Time 8-hour shift Nurse

	Time Accrued	Projected Annual	Projected Annual
	Per Hour	Hours For Full-Time	Days For Full-Time
Years of Service	Worked	8-Hr Employee	8-Hr Employee
First year of employment	.1000	208	26
2 nd through 3 rd year	.1193	248	31
4 th through 9 th year	.1385	288	36
10 th & subsequent years	.1500	312	39

Accrual Rates for Full-Time 12-hour shift Nurse

Years of Service	Time Accrued Per Hour Worked	Projected Annual Hours For Full-Time 12-Hr Employee	Projected Annual Days For Full-Time 12-Hr Employee
First year of employment	.1000	187.2	15.6
2nd through 3 rd year	.1193	223.33	18.6
4 th through 9 th year	.1385	259.27	21.6
10th and subsequent years	.1500	280.80	23.4

10.1.4 <u>Use of PTO</u>.

- a. PTO may be used as soon as it is earned. PTO may not be used in advance and may not be used on regularly scheduled days off.
- b. PTO, as with all other time off, such as absent time, must, except in unusual circumstances, be requested in writing in advance of the time off desired, and approved in writing by the Supervisor, except for an emergency or illness. Approval will be based upon the Employer's determination of its staffing needs. When time off is requested without prior approval due to an emergency or illness (either physical or mental), a reason for the request is to be given and accrued PTO time must be used. When making the request, the employee is not required to disclose any medical information. A Regular or Relief Nurse's time off due to their mental health needs or condition will be treated as time off for the Nurse's own illness for purposes of federal, state, or local sick leave laws, provided the Nurse is eligible under such law and has applicable sick leave time accrued. A Nurse requiring time off without prior approval must call at least two (2) hours before the start of the assigned shift. If the department is not open before the start of the shift, the employee has sixty (60) minutes after the opening of the office within which to call. If the Nurse does not have approval for each day of absence, it shall be considered
an unpaid, unexcused absence. Such absences can become cause for disciplinary action. The Employer may request a doctor's certificate of illness, to be documented by a treating licensed healthcare provider and/or require the Nurse to be seen by the Employer's designated Physician or Nurse Practitioner if the amount of time off due to illness is deemed excessive by the Employer.

- c. The Employer shall use its best efforts to provide Regular Nurses rotating off the night shift to the day shift with a schedule which allows two (2) consecutive unpaid twenty-four (24) hour periods off between shifts. In the event two (2) consecutive unpaid twenty-four (24) hour periods off are not scheduled, the Regular Nurse may request, at her/his option, either Absent Days or PTO. Such scheduling and such requests shall be given preferential treatment over other PTO requests. Such requests shall not affect the Regular Nurse's rights to receive premium pay for weekends when appropriate under the terms of this Agreement.
- d. PTO or an absent day may be offered to and accepted by a Regular Nurse if such Nurse is extra on her/his respective unit and cannot in the determination of the Employer be utilized elsewhere to meet a staffing need.
- e. In units where there is prolonged low census, those Regular Nurses who have depleted their PTO will be eligible, depending on operational needs, for up to two weeks of vacation annually without pay. CRONA and Nursing Administration will mutually agree on the definition of a period of prolonged low census.
- f. PTO shall be granted to Regular Nurses having Herpes who are assigned to specialty units (Delivery Room, Nurseries), and cannot be utilized in other areas.
- g. Nurses not allowed to work because of on-the-job exposure to a communicable disease are entitled to be compensated under the terms of the workers' compensation policy provided by the Employer for the duration of the incubation period or until such time that the Employer determines is appropriate for protecting the health and safety of the Nurse, coworkersworkers, and patients. Accrued PTO may be used during the waiting period for workers' compensation until the Regular Nurse becomes eligible for ESL, and may be used to supplement any such workers' compensation payments to equal the Nurse's regular earnings (at regular hourly rate of pay), after ESL has been exhausted.
- h. If a holiday falls within an orientation period and a new hire is not sufficiently oriented to be assigned work for the day, the new hire will be scheduled for PTO if desired, if sufficient PTO has been accrued. Otherwise the new hire will be scheduled for absent days.
- i. Accrued PTO may be used during any waiting period for State Disability Insurance or Workers' Compensation for which ESL is not available and may be used to supplement any such disability payments during a period of disability to equal the Nurse's regular earnings (at regular hourly rate of pay) after ESL has been exhausted. A Regular Nurse will be eligible to take prescheduled vacation, even if the Nurse has insufficient PTO, if the

insufficient PTO is a result of the use of PTO during a waiting period for Workers' Compensation that occurred following the scheduling of the vacation.

- j. If work is not available on a holiday, the Nurse may use either PTO or Absent time.
- k. Regular Nurses will accrue PTO hours up to five hundred twenty (520) hours. If a Nurse's PTO accrual reaches five hundred twenty (520) hours, further accrual will cease until the Nurse utilizes PTO hours and reduces the accrual to below five hundred twenty (520) hours. As an exception, if a Nurse who accrues a balance of four hundred eighty (480) hours of PTO has made three (3) or more attempts within a calendar year to schedule PTO in advance, in writing and in accordance with departmental procedures, including alternative date requests, and all of the Nurse's requests for PTO within that same calendar year were denied, then upon the Nurse's prompt request, he/she will be granted PTO in an amount equal to the greatest amount of PTO requested in any one of the previously denied requests, up to eighty (80) hours. The PTO will be granted within ninety (90) days, but at a time consistent with operational needs to the extent possible, and not during the period between December 10th and January 5th. The Nurse and his or her manager will work together to attempt to schedule PTO in a manner that accommodates the interests of both the Nurse and the Hospital.
- Ι. Cash Out PTO Account. Thirty (30) days prior to implementation of the Cash Out PTO Account, the Employer will distribute to Nurses a written explanation. written in a manner intended to be readily understandable, of the accrual cap and the Cash Out PTO Account how they will work, how to check PTO and Cash Out PTO balances, and where to call with questions. Any Nurse who, as of the one hundred twentieth (120th) day following ratification of this Agreement, has accrued but unused PTO in an amount in excess of two hundred sixty (260) hours, will have the accrued but unused PTO hours that exceed two hundred sixty (260) hours transferred into a Cash Out PTO Account. The hours so transferred shall thereafter be paid out to the Nurse annually, on the last payday of August, in an amount up to one hundred (100) hours until the Cash Out PTO Account has been exhausted, provided that if such amount would not be exhausted by the sixth (6th) such payment of one hundred (100) hours, the entire remaining balance will be paid out in the sixth (6th) annual payment, and the one hundred (100) hour limit shall not apply. Payments made in lieu of paid time off will be at the Nurse's base wage rate and will not include shift differential. Whenever the regular PTO accrual account of a Nurse with a Cash Out PTO Account reaches a balance of forty (40) hours or less, PTO time from his/her PTO Cash Out Account will be transferred to his/her regular PTO account in an amount equal to eighty (80) hours, or equal to the entire Cash Out PTO Account balance if said balance is less than eighty (80) hours. Once the PTO is transferred from the Cash Out PTO Account to the Nurse's regular PTO account, the PTO hours are available for any purpose for which PTO may or must be utilized. Such transfers of PTO from the Nurse's Cash Out PTO Account will not reduce the amount of the annual cash out except to the extent that utilization of PTO from the Cash Out PTO Account reduces the balance to below one hundred (100) hours. The Nurse's Cash Out PTO balance will be shown separately from the Nurse's regular PTO accrual balance on e-Connect, and the Employer will

advise the Nurse of the annual Cash Out payment in advance of the actual payment.

- Voluntary PTO Cash Out. During the annual benefit open enrollment period m. (typically held in November), a Nurse who as of the beginning of the payroll period immediately before the start of the open enrollment period has four hundred forty (440) hours or more of accrued but unused PTO in his/her regular PTO Accrual Account, and no accrued but unused hours in a Cash Out PTO Account, may pre-elect to receive payment in lieu of paid time off for up to eighty (80) hours of PTO to be accrued in the following calendar year, which cash out payment will be made in November of the following calendar year. In order to receive such payment in lieu of paid time off, the Nurse must have the accrued but unused PTO necessary to cover the payment at the time the payment is to be made. If the Nurse does not have enough accrued but unused PTO at the time the payment is to be made, the Nurse will be paid for such portion of the elected amount as his/her accrued but unused PTO will cover, if any. Payments made in lieu of paid time off will be at the Nurse's base wage rate and will not include shift differential. The Employer will notify the Nurse that the payment has been made.
- n. All Regular Nurses shall be permitted to cash out hours in the event of an "unforeseeable emergency" within the meaning of the Internal Revenue Service rules or guidelines. Nurses must submit a request for approval form to the Vice President of Patient Care with the required supporting documentation. If approved by the Director of Employee and Labor Relations, payment will be made to the Nurse. Employer decisions to approve or deny requests for cash outs shall be final and binding and not reviewable under the terms of this Agreement.
- A Nurse changing status from Regular to Relief will receive an immediate cash-out of all PTO hours accrued at her/his hourly base wage rate, in his/her PTO Account, and in his/her Cashout PTO Account, if applicable, (not including any applicable relief per diem).
- p. The Employer will make a good faith effort to grant each Nurse two (2) consecutive weeks of vacation per year, if requested, without requiring the Nurse to find coverage.
- 10.1.5 <u>Holidays</u>. The following holidays are built-in components of the PTO accrual rates: New Year's Day, Martin Luther King's Birthday, Memorial Day, Independence Day, Labor Day, Thanksgiving Day, Christmas Day, and the Regular Nurse's Birthday or alternatively, any other work day mutually agreed upon by the Regular Nurse and her/his supervisor within the year following the Nurse's birthday. However, Regular Nurses required to work on any actual holiday, except their birthday, will be paid 1-1/2 times the regular hourly rate of pay including any applicable shift differential for all time worked between 10:45 p.m. on the eve of the holiday and 11:15 p.m. on the day of the holiday. In addition to the one and one-half (1-1/2) times pay, the Regular Nurse may claim eight (8) hours of accrued PTO to be paid at the hourly base wage rate plus shift differential if applicable in accordance with Section 7.5.7.
- 10.1.6 Upon termination from the Employer, a Nurse will receive a cash-out of all accrued but unused PTO hours from the employee's regular PTO accrual account, and from

his/her Cash Out PTO Account, if applicable, at the Nurse's current base hourly wage rate, not including any shift differential rate.

- 10.2 Extended Sick Leave (ESL).
 - 10.2.1 <u>Purpose and Rate of Pay</u>. Extended sick leave compensates Regular Nurses in the case of their own extended illness or injury, and is utilized for absences beginning the fourth (4th) consecutive day of illness or injury, or, from the first (1st) day if the employee is hospitalized on that day. A Nurse may not take unpaid time off for any absence for which ESL is intended, if the employee has accumulated ESL available, Nurses with anticipated or realized six (6) months of assignment to an evening or night shift in a calendar year will receive shift differential for ESL hours used when the Nurse is scheduled to work a shift for which a differential is paid at the time the ESL hours are used.
 - 10.2.2 <u>Eligibility</u>. All Regular Nurses are eligible to accumulate ESL.
 - 10.2.3 <u>Accumulation of ESL</u>. Effective the beginning of the pay period following ratification of this Agreement, Nurses will begin accumulating ESL hours for use in the case of extended illness or injury. Nurses will accumulate ESL at the rate of .0116 hours per hour worked (the equivalent of 24 hours per year for a full-time employee). ESL will be accumulated on the basis of productive hours worked, including overtime, on PTO and ESL hours taken, and for paid jury duty and paid bereavement leave hours, to a combined maximum of eighty (80) such hours in a biweekly pay period. There is no limit on the accumulation of ESL.
 - 10.2.4 <u>Use of ESL</u>. ESL must be used for any absence for which its use is intended, until exhausted. ESL hours for a full day's absence will be taken in the same manner as the Nurse is normally scheduled to work (e.g., in 8 hour increments for 8 hour shifts, in 12 hour increments for 12 hour shifts, etc.) and will include applicable shift differential.
 - a. A Nurse changing status from Regular to Relief status will lose all accumulated but unused ESL.
 - b. Accumulated but unused ESL is not paid out upon a Nurse's termination of employment.
 - 10.2.5 <u>Integration of ESL</u>. In cases where a Nurse is receiving disability benefit payments, ESL will be integrated (e.g., SDI Workers' Compensation, or Supplemental Short Term Disability). To the extent disability payments do not equal the Nurse's regular wages, ESL will be used in an amount equal to, but not exceeding, the Nurse's scheduled hours at his/her straight time hourly rate and any shift differential to which (s)he would be entitled.

SECTION 11 GOVERNMENT PROGRAMS

11.1 <u>State Disability Insurance (SDI)</u>. SDI provides partial income to Nurses with non- workrelated illnesses or injuries who meet eligibility requirements set by the State of California. Premiums are deducted from the Nurse's salary as required by law. Nurses who sustain work-related injuries are not covered by SDI but may be eligible for Workers' Compensation as specified in this Section.

Benefits.

- a. Benefits become payable on the eighth (8th) day of the disability. The maximum period covered is fifty-two (52) weeks.
- b. A pregnant Nurse may be eligible for benefits for both normal and abnormal pregnancies, if she meets the eligibility requirements and her doctor certifies that she is unable to do her regular work.
- c. Claims must be submitted directly to the California Employment Development Department.
- 11.2 <u>Workers' Compensation</u>. The Employer provides Workers' Compensation for Nurses sustaining on-the-job or demonstrably work-related injuries. The Employer pays the entire cost of this program. The Employer will investigate and attempt to resolve complaints regarding the mishandling of workers compensation investigations.
- 11.3 <u>State Unemployment Insurance</u>. Certain former Nurses may be eligible for State Unemployment Insurance. The State of California establishes requirements and determines eligibility and benefits.
- 11.4 <u>COBRA</u>. The Employer provides continuation of coverage of benefits as defined in the Consolidated Omnibus Budget Reconciliation Act of 1985 (COBRA), as amended.

SECTION 12 EMPLOYER RETIREMENT PLAN

12.1 Employer Retirement Plan.

- 12.1.1 <u>Retirement Plan</u>. The Employer's Retirement Plan for eligible Nurses covered by this Agreement is described herein. It is understood that the description does not purport to recite completely the coverage, eligibility requirements, carrier or other details. In the event the Employer makes changes which include improvements for unrepresented employees in the Employer's Retirement Plan which is common to both the CRONA unit and such unrepresented employees, CRONA will be offered the opportunity to accept those changes; provided that CRONA's acceptance is given in writing and is based on the changes in their entirety.
- 12.1.2 The Employer will make regular deposits as set forth in 12.1.3.b with the Section 403(b) custodian. The Employer will make regular deposits as soon as practicable after payday, but within no more than seven (7) days, except in circumstances outside the Employer's control.
- 12.1.3 <u>Terms and Conditions</u>.
 - a. To be eligible to participate in the Employer Retirement Plan, a Nurse must be employed in a Regular position requiring at least forty (40) hours work per pay period with the duration of such employment expected by the Employer to exceed twelve (12) months of consecutive employment. Participation begins after twelve (12) months of continuous employment. [This change applies only to Nurses hired on or after April 1, 2007.]

- b. <u>Basic Contribution</u>. The Employer's total annual contribution to the retirement plan shall be five percent (5%) of the individual Regular Nurse's annual pension eligible earnings.
- c. <u>Retirement Match</u>. In addition a Regular Nurse who makes a voluntary contribution of one percent (1%), two percent (2%), three percent (3%) or four percent (4%) of pension eligible earnings will receive an additional matching contribution from the Employer equal to the Regular Nurse's contribution to the Employer's Retirement Plan.
 - (1) Effective the pay period beginning May 6, 2007, the Employer will match the voluntary contribution of a Regular Nurse with fifteen (15) years of retirement eligible service up to six percent (6%) of pensioneligible earnings. Eligibility will begin the pay period following the date on which the Regular Nurse reaches fifteen (15) years of retirement eligible service.
 - (2) Effective at the start of the payroll period after June 6, 2011, the Employer will match the voluntary contribution of a Regular Nurse with ten (10) years of retirement eligible service up to five percent (5%) of pension-eligible earnings.
 - (3) Effective January 1, 2017, the Employer will match the voluntary contribution of a Regular Nurse with between five (5) and ten (10) years of retirement eligible service up to five percent (5%) of pension-eligible earnings, and the Employer will match the voluntary contribution of a Regular Nurse with ten (10) or more years of retirement eligible service up to seven percent (7%) of pension-eligible earnings.
 - (4) Effective the first pay period of 2022 (pay period beginning 12/26/2021), the Employer will match the voluntary contribution of a Regular Nurse with fewer than five (5) years of retirement eligible service up to five percent (5%) of pension-eligible earnings, the Employer will match the voluntary contribution of a Regular Nurse with between five (5) and ten (10) years of retirement eligible service up to six percent (6%) of pension-eligible earnings, and the Employer will match the voluntary contribution of a Regular Nurse with ten (10) or more years of retirement eligible service up to eight percent (8%) of pension-eligible earnings.
- d. Nurses in Relief status C or D shall be eligible for the five percent (5%) Employer contribution only, provided they satisfy the eligibility requirement of one (1) year of employment in which they work at least one thousand (1000) hours. After the Nurse has satisfied the eligibility requirement, for each subsequent calendar year in which the Nurse works at least one thousand (1000) hours, and provided the Nurse is employed on the last day of the calendar year, the Employer will make the required contribution in April of the following year. The one (1) year eligibility requirement shall be waived for Relief C and D Nurses who are employed on the date of ratification of the 2010-2013 Agreement, nor shall it apply to a Nurse who changes from regular status to Relief status C or D, provided that Nurse has been employed for one (1) year prior to conversion to relief status.

- e. Pension eligible earnings include all earnings at the Nurse's regular hourly rate including evening and night shift differential and the straight time portion of overtime and double time, but does not include any other allowances or premiums or differentials.
- 12.1.4 A Regular Nurse, Relief Nurse C or Relief Nurse D who changes to Limited Relief Nurse, or Relief A or Relief B, will not be eligible to participate in the Stanford Health Care Retirement Plan after the end of the pay period in which his/her status changes.
- 12.2 <u>Tax-Deferred Annuity Program</u>. The Employer provides the opportunity for all Nurses covered by this Agreement to contribute up to the allowable limit of earnings to the Employer's Tax-Deferred Annuity Plan, under IRS Regulation 403(b).

SECTION 13 PRE-PLACEMENT AND ANNUAL PHYSICAL EXAMINATIONS

13.1 <u>Pre-placement and Annual Physical Examinations</u>. All Nurses must be cleared by an Employer designated Physician or Nurse Practitioner before being scheduled to work at Stanford Health Care. The examination will include those elements which the Physician or Nurse Practitioner determines are appropriate in evaluating a Nurse's health relative to employment by the Employer, e.g., personal and family medical history, an examination, laboratory studies, immunizations and other relevant procedures. At the end of one (1) year of continuous employment and annually thereafter, a Nurse may obtain a physical examination by an Employer designated Physician or Nurse Practitioner without cost to the Nurse. The examination will include those elements which the Employer determines are appropriate in evaluating a Nurse's health relative to continued employment by the Employer.

SECTION 14 EDUCATIONAL ASSISTANCE AND PROFESSIONAL ENRICHMENT

14.1 Purpose.

- 14.1.1 To encourage and assist eligible Nurses in increasing their effectiveness in performing the duties of their present jobs and to increase their knowledge and effectiveness as Registered Nurses.
- 14.1.2 To recognize and support according to the terms set forth below the ongoing need for continuing education for many Nurses as part of maintaining licensure or certification required by the Employer.

14.2 <u>Eligibility</u>.

- 14.2.1 Regular full-time and part-time Nurses who have completed their trial period and one thousand (1000) hours actually worked and "C" and "D" Relief Status Nurses who have completed their six month trial period and the equivalent number of hours (1000), are eligible to participate in this program.
- 14.2.2 Nurses who have not completed the eligibility requirements set forth above are not eligible to participate in this program. Once eligible a Nurse must maintain an eligible status throughout the entire course(s) or reimbursement will not be granted. Nurses on leave of absence are not authorized to take any training or class and will

not receive pay for time they spend attending any such training or class and are not eligible for expense reimbursement under this section.

14.2.3 Nurses who have completed the eligibility requirements set forth above and who are granted an educational leave pursuant to Section 19 of this Agreement will continue to be eligible for reimbursement under the terms of this program so long as the Nurse maintains at least a "C" commitment.

14.3 Reimbursement.

- 14.3.1 The Employer will pay tuition for courses taken for credit (including Home Study), certification exams and certification renewals completed by eligible Nurses according to the Program, Course, and/or Exam qualifications in 14.7, and reimbursement procedures in the Employer's Reimbursement Policy. Eligible part-time Regular and Relief Nurses will be eligible for reimbursement on a pro-rata basis.
- 14.3.2 When required in connection with a qualified course, the reimbursement may cover lab fees, textbooks and software required by the course.
- 14.3.3 <u>Reimbursement Section</u>. The dollar maximum is Two Thousand Dollars (\$2000.00) per academic year (September 1 through August 31) for eligible full-time Regular Nurses and pro-rated amounts of that maximum for eligible part-time Regular and Relief Nurses.
 - a. Educational reimbursement may cover Continuing Education Course Tuition (includes cost of Continuing Education hours for ACLS, Chemo certification and Trauma Continuing Education hours, AR, PALS and NRP, tuition for PALS and NRP outside SHC, and Lane Library privileges); fees for professional certification and renewals, membership in a professional Nursing organization, Nursing Research Projects at SHC through SHC Nursing research; travel, lodging for out-of-area professional conferences; professional journals, books, software and audio-visual materials.
 - b. The Employer will respond to a Nurse's request for reimbursement or prior approval of expenses under this Section within thirty (30) days of receipt of the request; and, if the request is denied, the Employer's response will inform the Nurse of the position of the person who made the decision and the reason(s) why the request was denied.
- 14.3.4 <u>Continuing Education Hours</u>. All continuing education programs certified by the BRN and which are job related as defined in 14.7.1.b and accredited by the State of California for re-licensure shall qualify for educational assistance under the terms for reimbursement set forth in 14.7 and 14.8. In addition, all the Employer sponsored Continuing Education hours shall qualify for educational assistance under the terms for reimbursement.

14.4 Paid Educational Hours.

14.4.1 Effective with the 2022-2023 Academic Year, a full-time Regular Nurse shall be entitled to forty (40) hours with pay each academic year (the end of the pay period immediately before September 1 through the end of the pay period immediately before August 31, as set forth below) to attend programs and courses, in

accordance with the procedures and guidelines provided herein. Eligible Regular part-time Nurses and Relief Nurses will be eligible for reimbursement on a pro-rata basis.

- a. For the contract period, the dates inclusive of the academic year, for which a Regular Nurse will be able to use their education hours, are as follows:
 - (1) 2022/2023: August 21, 2022 through August 19, 2023
 - (2) 2023/2024: August 20, 2023 through August 17, 2024
 - (3) 2024/2025: August 18, 2024 through August 16, 2025
- b. For the current 2021 2022 academic year, Nurses still have the ability to use Education Hours through August 31, 2022.
- 14.4.2 <u>Procedure</u>.
 - a. If pre-approved educational hours have not been granted to a Nurse pursuant to the Employer's policy, the eligible Nurse must apply in writing at least thirty (30) days in advance, and no less than fifteen (15) days prior to the posting of the schedule during which the time is being requested, specifying the course, institute, workshop or class certification or exam the eligible Nurse wishes to attend, and the number of hours requested. Requests received less than thirty (30) days in advance and less than fifteen (15) days prior to the posting of the schedule during which the time is being requested may be granted at the sole discretion of the Manager.
 - b. A Nurse may request educational hours over the Nurse's commitment, for actual time spent attending a course, which may be granted at the sole discretion of the Manager.
 - c. The Employer shall schedule educational hours in accordance with scheduled work and staffing assignments and make a good faith effort to notify the Nurse within ten (10) days after receipt of a request whether or not the request for hours has been granted.
 - d. The Employer shall provide reimbursement within thirty (30) working days of receipt of the completed paperwork by the Department of Human Resources.
 - e. The decision to approve or deny a request for educational time off is within the sole discretion of the Employer, but such time will not be denied unreasonably.
- 14.4.3 <u>Yearly Basis</u>. Each eligible Nurse shall be eligible for educational hours as provided in this Section on the basis of the academic year, which is the period from September 1 August 31.
- 14.4.4 <u>Accumulation</u>. A Nurse entitled to apply for educational hours who does not apply waives them for the year. Nurses will make every attempt to request educational hours in a timely manner throughout the year. If an eligible Nurse requests educational hours for a program within the guidelines provided herein and does not

receive them in the year for which qualified, the eligible Nurse may accumulate them for use in the following year.

- 14.4.5 Payment for Educational Hours.
 - a. If an eligible Nurse requests and receives approval for an educational program taking less than the Nurse's full shift, the Nurse may elect to take the remaining hours of the shift, up to a maximum of the Nurse's full shift (i.e., eight (8), ten (10), or twelve (12) hours), as educational hours, PTO or absent time.
 - b. Subject to (a), above, if an eligible night shift Nurse requests permission to attend an educational program which would otherwise qualify under the educational hours and pay provisions provided herein, but which falls entirely outside the night shift, the Nurse's regularly scheduled shift, the Employer shall make a good faith effort to schedule the educational hours in lieu of the Nurse's night shift immediately preceding the program.
- 14.4.6 <u>Home Study</u>. Education time may be granted for home study courses according to the following procedure:
 - a. The eligible Nurse has received pre-approval for the home study course according to the procedures outlined in this Agreement.
 - b. Upon providing proof of successful completion of the home study course to the Manager, the eligible Nurse will be eligible for paid education time according to the terms of Section 14.2.
 - c. Education time for home study courses will be provided at the ratio of one (1) hour per one (1) credit earned.
 - d. All procedures for approval and reimbursement for education time for home study courses will be in compliance with applicable procedures of this Agreement.

14.5 Educational Training and Classes Required by the Employer.

- 14.5.1 The Employer will pay all Nurses (including Nurses not eligible to participate in the Educational Assistance and Professional Enrichment program) for the time spent and expenses incurred by the Nurse on educational training and classes that are required by the Employer, in accordance with applicable law and subject to advance approval of the Nurse's manager.
 - a. Notwithstanding the foregoing, the Employer will not pay the costs of fees, materials, and other expenses for any Nurse if the training and classes and related materials are available from or through the Employer but the Nurse takes the training or class elsewhere, unless the Nurse schedules to take the training or class pursuant to subparagraph (b) below.
 - b. As provided above in Section 14.2.2, the Employer will not pay a Nurse for time spent attending Employer-required training and classes while on a leave of absence. If a Nurse has been released to return from an approved leave of absence but must complete required training or a class before being permitted

to return to work, the Nurse's return to work date shall be scheduled by the Manager to occur on the first day that the Nurse is able to attend a scheduled session of the missing required training or class.

- 14.5.2 Educational training and classes required and provided by the Employer (including Skills Fair) will be charged against an eligible Nurse's educational hours if Continuing Education hours are provided and the Nurse elects to receive such Continuing Education hours, but a Nurse's educational hours will not be charged for Employer required training and classes if the Employer requires the Nurse to accept Continuing Education hours.
- 14.5.3 All other training and classes offered but not required by the Employer and taken by a Nurse will be charged against the eligible Nurse's educational hours if the Nurse elects to receive pay.
- 14.6 <u>Shift Differential for Educational Classes</u>. Nurses who take educational training and classes required by the Employer will receive shift differentials based upon the time of day the required training or class (s)he attends takes place. Nurses who take training and classes not required by the Employer shall be paid the Nurse's base hourly rate, without overtime premium or differentials, if the Nurse elects to receive pay.
- 14.7 Program, Course, and/or Exam Qualifications.
 - 14.7.1 Only courses taken for credit, certification exams and renewals, according to the following criteria, will be accepted for tuition assistance. A grade of "C" or higher must be obtained in a course taken for credit to qualify for reimbursement.

Criteria.

- a. Courses required for undergraduate, graduate or professional degree in the field of Nursing.
- b. Vocational, correspondence, extension, college or occupational skills courses, certification exams and renewals, and workshops which are clearly and demonstrably related to the effective performance of the eligible Nurse's present job or which in the judgment of the Employer will increase the Nurse's knowledge and effectiveness as a Registered Nurse with the Employer.
- 14.7.2 The approval of apparent course compliance will be made by the eligible Nurse's Unit Supervisor and will be then forwarded to the Vice President of Patient Care or her/his designee who may in consultation with the Human Resources Department either approve or deny the request.
- 14.8 Application Procedure.
 - 14.8.1 Program, Course, Exam Approval (other than Continuing Education Hours).
 - a. Before enrolling for a program, course, and/or exam, the eligible Nurse must present her/his supervisor with a written request describing how the program, course(s), and/or exam(s) meets the criteria set forth in 14.7 above.

- b. Programs, courses, and/or exams must be approved for qualification compliance by the applicant's Unit Supervisor and approved by the Vice President of Patient Care or designee prior to enrollment.
- c. If appropriate approval has been granted to an eligible Nurse for a total continuing program, e.g., completion of an undergraduate, graduate or professional degree in the field of Nursing, it is not necessary to request approval before registering for each course in the program.

If an eligible Nurse is not enrolled in a total program, such as the completion of a Bachelor's Degree, but desires to enroll in single courses, it is required that the Nurse seek appropriate approval for each individual course.

- d. The approved and completed form is retained in the Department of Nursing.
- 14.8.2 <u>Reimbursement</u>.
 - a. In order to receive reimbursement, the following must be submitted to the Nursing Administration within ninety (90) calendar days of the final date of the course(s):
 - (1) A completed current Tuition Reimbursement Application Form.
 - (2) Evidence of payment of expenses, e.g., a copy of a canceled check, receipt for payment or customer copy of credit card statement.
 - (3) Grade record "C" or higher or certification of course completion in courses where no grade is given, e.g., a copy of the BRN Continuing Education Certificate.
 - (4) Copy of exam/passing score notification.
 - b. Failure to submit paperwork before the defined time limit will result in nonreimbursement. If extenuating circumstances occur which are determined to be beyond the Nurse's control, e.g., grade record or canceled check has not been returned, the Nursing Administration Department must be informed in writing before the end of the ninety (90) calendar day limit.

14.9 Specialty Continuing Education.

- 14.9.1 When the Employer provides continuing education for Nurses, at least every tenth (10th) such opportunity per course will be reserved for a Nurse not currently assigned to a unit requiring training in the specialty subject offered.
- 14.9.2 The Nurse must have sufficient educational hours or PTO to complete the program and must request the time off from her/his Unit Supervisor as provided in 14.4.2 above.
- 14.10 Sabbatical Assignment.
 - 14.10.1 <u>Purpose</u>. The purpose of sabbatical assignments is to grant eligible Nurses the opportunity to receive assignments to a special project or a research project for the

purpose of enhancing professional development of the Nurse and of enhancing the practice and profession of nursing with the Employer.

- 14.10.2 <u>Eligibility and Qualifications</u>. All 0.8 time or more Regular Nurses with seven (7) or more years of continuous service with the Employer are eligible to apply for posted sabbatical assignment or to submit a proposal for a sabbatical assignment.
- 14.10.3 <u>Terms of Sabbatical Assignments</u>. The Vice President of Patient Care may identify the subject matter, duration, time commitment and other terms of the special project or research project, which will then be posted by the Nursing Department for application by eligible Nurses. In addition, any Regular Nurse with seven (7) or more years of continuous service may submit a written proposal for special project or research project. The proposal shall be submitted to the Regular Nurse's unit supervisor for review. If the supervisor determines it to be worthy of further consideration, the supervisor will submit it to the Vice President of Patient Care for review. If the Vice President of Patient Care approves the proposal in whole or in part, s/he will determine the duration, time commitment, and other terms of the special project.
- 14.10.4 <u>Selection</u>. Selection among eligible applicants or among proposals for sabbatical assignments shall be based on the Vice President of Patient Care's judgment of the merit of the proposal and qualifications, skill and ability to perform the assignment, the operational needs of the applicants' units, past participation in Employer or nursing service committees and special projects. Whenever two (2) or more Regular Nurses are determined by the Vice President of Patient Care to be substantially equal, priority shall be given to Nurses who have not previously been offered a sabbatical assignment.
- 14.10.5 <u>Pay and Benefits</u>. Regular Nurses selected for sabbatical assignment shall be paid at their base wage rate and benefits for the duration of the sabbatical assignment.
- 14.10.6 <u>Number of Sabbatical Assignments</u>. The number and terms of sabbatical assignments shall be determined by the Vice President of Patient Care. If a sufficient number of research projects or special projects are identified for posting by the Vice President of Patient Care or if a sufficient number of Nurse proposals are approved by the Vice President of Patient Care, sabbatical assignments shall be offered to up to three (3) Regular Nurses per contract year.

SECTION 15 RELIEF NURSE PROGRAM

- 15.1 <u>Summary of Responsibilities</u>. The Relief Nurse provides a per diem staffing resource to supplement the regular and part time staffing on Nursing Units. The Relief Nurse may be designated to act as Resource Nurse and is responsible for the delivery of coordinated Nursing care for a patient or a group of patients within a specified patient care unit. S/he performs the clinical related activities according to the respective assigned unit's criteria. All terms and conditions for the Employer's employment of Relief Nurses are set forth in this Section unless as otherwise expressly provided in another Section of this Agreement.
- 15.2 <u>New Hires</u>. All newly hired Relief Nurses will serve a trial period. After completion of no more than three (3) months of employment, the Employer will give the Relief Nurse an evaluation in writing if s/he is having difficulty in meeting expectations. This provision does not limit in any way the right of the Employer to terminate a Relief Nurse without cause

during the trial period; nor does it give rise to any claim concerning such termination under the Grievance and Arbitration provision. For "A" and "B" Relief Nurses the trial period shall commence on the first day of employment and extend until the date the Relief Nurse has actually worked one thousand (1,000) hours or twelve (12) months of employment provided the trial period shall in no case be less than six (6) months of employment. The trial period for "C" and "D" Relief Nurses shall be six (6) months. During the trial period a Relief Nurse may be terminated at any time at the Employer's sole discretion and the Employer's discretion shall not be subject to review under any provisions of this Agreement.

15.3 Qualifications for Employment.

- 15.3.1 At the option of the Employer a newly hired Relief Nurse may be required to attend and satisfactorily complete the Registered Nurse orientation program to qualify to be placed on the Relief Nurse availability list. Orientation requested over that required by the Employer to enable a Relief Nurse to function in a specific area must have the approval of Unit Supervision and Nursing Administration.
- 15.3.2 Newly hired Relief Nurses with less than twelve (12) months experience will be hired at the Relief Nurse I level.
- 15.3.3 Newly hired Relief Nurses with the equivalent of twelve (12) full months or more of experience will be hired at the Relief Nurse II level. The Employer will determine appropriate step placement based on its assessment of the Relief Nurse's experience provided that:
 - a. Experience must be within the past two (2) years.
 - b. Experience must be applicable to the unit or area(s) in which the Relief Nurse is hired as determined by the Patient Care Manager.

15.4 Movement from Regular Nurse Employment.

- 15.4.1 A move from Regular full-time or part-time Nurse status to Relief Nurse status must be approved by the Patient Care Manager and Department of Nursing. Evaluation of performance must be completed by the Patient Care Manager before request is considered.
- 15.4.2 Regular Clinical Nurse IIs approved to move to Relief Nurse status without interruption of employment will retain their Clinical Nurse II status. In the event a Clinical Nurse III or IV requests and is approved to move to Relief Nurse status, the Clinical Nurse III or IV will retain his/her Clinical Nurse status if the Clinical Nurse continues to meet the eligibility criteria of the PNDP. Otherwise, the Nurse will be converted to Clinical Nurse II and placed on the same step on the Clinical Nurse II salary scale.
- 15.4.3 A Regular Nurse who requests and receives an educational Leave of Absence to be employed as a Relief Nurse will upon satisfactory maintenance of Relief Nurse commitment and a return to employment as a Nurse, have no adjustment in her/his hire date.
- 15.4.4 A Regular Nurse who changes her/his status to Relief Nurse and returns to Regular Nurse status, will, if the Relief commitment has been met according to the requirements set forth in this Section, have no adjustment in her/his hire date.

15.5 <u>Rehires – Relief Nurses</u>.

- 15.5.1 If rehired into the same or similar unit or area at the same commitment previously held and maintained within twelve (12) months of resignation, the Relief Nurse is placed at the Relief Clinical Nurse II step that had been attained at the time of resignation.
- 15.5.2 In the case of the rehiring of a Nurse who had been employed as a Clinical Nurse III or IV, the Nurse shall be rehired as a Clinical Nurse II and the Nurse may reapply for Clinical Nurse III or IV status at the next available application period. If hiring into other than the same or similar unit or area the Relief Nurse will be treated as a new hire. The Employer will determine whether a unit or area is similar based on its review of the clinical skills and experience required.
- 15.5.3 If rehired after twelve (12) months from resignation with no intervening employment, the Relief Nurse will be considered as a new hire. The Employer will evaluate any intervening employment to determine appropriate step placement.
- 15.5.4 If a Relief Nurse is rehired into the same unit or area at the same commitment previously held and maintained within two (2) years from resignation, and has had continuous clinically comparable employment as determined by the Patient Care Manager, the Relief Nurse is placed in the Relief Clinical Nurse level II and step that had been attained at the time of resignation. If hiring into a different unit or area the Relief Nurse will be treated as a new hire.
- 15.6 <u>Transfers to Different Units or Areas</u>. Relief Nurses are not eligible for transfer during the trial period of employment. If a Relief Clinical Nurse III or IV requests and receives a transfer, Section XV of the Professional Nurse Development Program will apply.
- 15.7 <u>Movement to Regular Nurse Employment During the Trial Period</u>. Trial period Relief Nurses who apply and are hired into Regular full-time or part-time Nurse status without interruption of employment retain their Relief Nurse level and step but will be required to serve the trial period specified in Section 6.1. of this Agreement.
- 15.8 <u>Movement to Regular Nurse Employment After the Trial Period</u>. Relief Nurses who have successfully completed the Relief Nurse trial period who apply and are hired into Regular full-time or part-time Clinical Nurse II status without interruption of employment will retain their Relief Nurse step. In the event a Relief Clinical Nurse III or IV requests and is approved to move to Regular Nurse status, the Relief Clinical Nurse III or IV will retain his/her Clinical Nurse level and step if the Clinical Nurse continues to meet the eligibility criteria of the PNDP. Otherwise, the Nurse will be converted to Clinical Nurse II and placed on the same step on the Clinical Nurse II salary scale.
- 15.9 In applying for posted vacant positions, Relief Nurses shall accrue and use seniority pursuant to Section 23.
- 15.10 <u>Reclassifications of Relief Nurses</u>. Voluntary and involuntary reclassification shall be the same as for Regular Nurses (Section 6.3).
- 15.11 <u>Performance Evaluation</u>. Relief Nurses will be evaluated every twelve (12) months. In so far as practicable the Relief Nurse's review date will coincide with the Relief Nurse's original date of hire. The performance evaluation process for Relief Nurses will be the same as for Regular Nurses.

15.12 Employment Commitment and Status.

15.12.1 Relief Nurses are required to make a written commitment of availability (as defined below) to her/his respective schedule planner. Shifts may be worked in four (4), six (6), eight (8), ten (10), or twelve (12) hour blocks in order to fulfill commitment. Relief Nurse commitment is successfully met by actually working all hours of commitment for which the Relief Nurse is scheduled by the Employer unless s/he is canceled by the Employer. Failure of a Relief Nurse to keep her/his commitment may result in a corrective action notice and compliance will be expected for the next schedule period. Continued noncompliance may result in termination.

Classifications	Holiday Commitment	Weekend Commitment
"Limited – Minimum 24 shifts	At least 1 of the following	Minimum of 3 weekends
per 12 month period	Holidays:	per 12 month period
commencing with Relief	December 24-evening shift	commencing with Relief
Nurse review date	December 24-night shift	Nurse review date.
	December 25-day shift	
	December 25-evening shift	
	December 31-evening shift	
	December 31-night shift	
	January 1-day shift	
"A" – 16 hours per 80 hour	At least 1 of the following	1 weekend in a 4 week
pay period. Only RNs with	Holidays:	period.
at least 1 year of acceptable	December 24-evening shift	-
experience will be accepted	December 24-night shift	
as "A" commitment.	December 25-day shift	
	December 25-evening shift	
	December 31-evening shift	
	December 31-night shift	
	January 1-day shift	
"B" – 24-32 hours per 80	At least 1 of the following	1 weekend in a 4 week
hour pay period	Holidays:	period.
	December 24-evening shift	
	December 24-night shift	
	December 25-day shift	
	December 25-evening shift	
	December 31-evening shift	
	December 31-night shift	
	January 1-day shift	
"C" – 40-56 hours per 80	At least 1 of the following	2 weekends in a 4 week
hour pay period.	Holidays:	period.
	December 24-evening shift	
	December 24-night shift	
	December 25-day shift	
	December 25-evening shift	
	December 31-evening shift	
	December 31-night shift	
	January 1-day shift	
"D" – 64-80 hours per 80	At least 1 of the following	2 weekends in a 4 week
hour pay period	Holidays:	period
	December 24-evening shift	
	December 24-night shift	

Classifications	Holiday Commitment	Weekend Commitment
	December 25-day shift	
	December 25-evening shift	
	December 31-evening shift	
	December 31-night shift	
	January 1-day shift	

Holidays are considered the same as a weekend day for the purpose of fulfilling the weekend obligation. Only Holidays enumerated in Section 10.1.5 shall be compensated at one and one-half (1-1/2) times the Relief Nurse's regular hourly rate of pay. Relief Nurses are not eligible for the weekend premium specified in Section 17 of this Agreement. For purposes of meeting weekend commitments all required weekends must be worked or made-up within the following 12-week period. A Relief Nurse who has provided her/his weekend availability and who is not utilized is not required to make up that weekend. The Employer will consider exceptions to this requirement as appropriate.

- 15.12.2 Relief Nurses who work only in a specific unit on a "limited" commitment basis must meet a commitment arranged with the Nurse Manager and approved by the Department of Nursing according to the needs for that unit.
- 15.12.3 Working full shift on Saturday or Sunday (day or evening shift) or Friday or Saturday (night shift) will count as half of a weekend toward meeting the weekend commitment.
- 15.12.4 Relief Nurses who work in units which are not open on weekends are not required to make up weekend commitments elsewhere.
- 15.13 Compensation and Benefits.
 - 15.13.1 Relief Nurses will be paid in accordance with the salary schedules set forth in Appendix A of this Agreement on an all hours worked basis. Relief Nurses shall be eligible for progression to the next salary step upon the accumulation of one thousand (1000) hours of work, provided (a) no Relief Nurse shall advance more than one (1) step during the twelve (12) month period commencing with the date of employment or the date of the Relief Nurse's most recent step advancement, and (b) the accumulation of one thousand (1000) hours is accomplished in no more than four (4) years from the date of assignment to the Relief Nurse's current step. If a Relief Nurse does not work at least one thousand (1000) hours in such four (4) year period, the Relief Nurse will remain in the same step and must commence a new accumulation of the one thousand (1000) hours toward step advancement. In addition to meeting the requirements for Step 6, advancement to Step 7 requires one thousand (1,000) hours of work within the four (4) year period since advancement to Step 6 and one (1) year of continuous service since advancement to Step 6.
 - 15.13.2 Effective on May 22, 2016, Relief Nurses in the C and D classifications will receive an additional Seven Dollars and Fifty Cents (\$7.50) per hour added to the Relief Nurse's hourly base wage rate. Effective on May 22, 2016, Relief Nurses in the Limited, A, and B classifications will receive an additional Ten Dollars (\$10.00) per hour added to the Relief Nurse's hourly base wage rate.
 - 15.13.3 Shift differential is paid to Relief Nurses pursuant to Section 7.5. of this Agreement.

15.13.4 Relief Nurses are eligible for the Employer Group Insurance Programs in accordance with their commitment.

"A"	No option toward benefit.
"B"	Option to purchase any or all at group discount rate.
"C" & "D"	Group Insurance is paid by the Employer, however, Nurses on "off-call" for a period exceeding one month must pay the premium for any coverage desired until on-call status is resumed.

15.13.5 A Relief Nurse shall accrue California Sick Leave as provided by California law.

15.14 <u>Scheduling</u>.

- 15.14.1 It is understood that the Employer has no obligation to call Relief Nurses according to their availability commitments or areas of preference. However, Relief Nurses will be called and assigned to areas according to stated availability and individual preference whenever possible. Relief Nurse assignments will be made based on the staffing needs of the Employer and Nursing Supervision's assessment of the individual Nurse's competence and skills including but not limited to such factors as experience, education and specialty. In making this assessment the supervisor will seek input from the Relief Nurse and respect the individual's assessment of those skills. Relief Nurses who have been prescheduled to work are required to call in prior to the beginning of the scheduled shift to confirm that they are needed. Relief Nurses who fail to contact their unit pursuant to this subsection will not be eligible for report time pay.
- 15.14.2 Relief Nurses are expected to provide their availability including weekend and holiday commitment when applicable to the respective Schedule Planner according to the established cut-off dates for schedule planning. Availability must be provided up to the Relief Nurse's full commitment. A Relief Nurse who is scheduled to work is expected to work unless canceled by the Employer. Failure of a prescheduled Relief Nurse to work will be considered a failure to keep commitment under 15.12.1.
- 15.14.3 Every effort will be made to call the Relief Nurse for duty as far in advance as possible; however, since staffing needs change unexpectedly, a Relief Nurse may be requested to work on short notice. Cancellation of service by the Employer will be made at the earliest possible time. Relief Nurses not called prior to the actual start time of shift will not be obligated to remain available. To the extent practicable each unit shall attempt to cancel Relief Nurses in inverse order of relief commitment.
- 15.14.4 Relief Nurses called for duty after the beginning of a shift have the option of working a full eight (8) hour period, or may leave at the end of the shift and be paid only for hours worked.
- 15.14.5 Availability does not constitute confirmation of work. Definite confirmation must be made by the authorized Unit Services' personnel before the Nurse reports to work. If the staffing needs change after the Relief Nurse reports to work for a specific area, and no alternate assignment can be made, she or he is reimbursed for a minimum of four hours. Relief Nurses sent home because of health problems will be reimbursed for the time worked only.

- 15.14.6 A newly hired Relief Nurse must work three hundred fifty (350) hours to be eligible for "off-call" as defined in this subsection. An eligible Relief Nurse may take up to a total of eight (8) weeks off ("off-call") with a limit of four (4) weeks during the period from June 1 September 30. "Off-call" will be considered only applicable to requested periods comprising at least seven (7) consecutive days and when the Nurse's commitment is not met in that pay period. The three hundred fifty (350) hours worked must include meeting the appropriate holiday and weekend commitment as specified in 15.12.1. The "off-call" period is defined yearly from October 1 to September 30. Failure of Relief Nurses to meet the commitments of "off-call" requirements may result in termination of their services.
- 15.14.7 Relief Nurses are not eligible for Leaves of Absence except as provided by the Federal and California Family/Medical Leave Acts and other applicable laws. However, Relief Nurses incapable of performing their jobs due to a verified pregnancy related disability will be considered to be off-call for the period of verified disability up to a maximum of six (6) months. At the end of the period of verified disability the Relief Nurse is eligible to return to the Relief pool at the same commitment and Relief Nurse level held prior to the onset of pregnancy related disability.
- 15.14.8 A "C" or "D" Relief Nurse called for Jury Duty or a legal appearance on behalf of the Employer as defined in Section 21 of this Agreement will be compensated at her/his regular hourly rate of pay for time spent on Jury Duty or legal appearances on behalf of the Employer for all hours the Relief Nurse is actually scheduled to work (as opposed to providing availability). The Relief Nurse must notify her/his supervisor immediately upon receipt of the summons so that arrangements may be made for the absence. Upon request by the supervisor, proof of completion of time spent on Jury Duty will be provided. Jury payments received are kept by the Relief Nurse.
- 15.14.9 <u>Educational Assistance</u>. Nurses in relief category "C" or "D" shall be entitled to any educational assistance specified in this Agreement prorated according to commitment actually worked.

SECTION 16 TEMPORARY REASSIGNMENTS/FLOATING

- 16.1 CRONA recognizes the Employer's right to temporarily reassign/float Nurses. The Employer and CRONA also recognize their shared interest that such reassignments be consistent with the competence and skills of the Nurse.
- 16.2 <u>Voluntary Commitment Floating Outside Region</u>. Any qualified Nurses who wish to volunteer for floating outside their designated region shall notify the staffing office. Preference among Nurses for a float/reassignment outside the designated region will be made to Nurses who have volunteered a commitment, provided that the assignment is consistent with the competence and skills of the Nurse, including but not limited to such factors as experience, education, and specialty and cross-training opportunities in the assessment of nursing supervision. In making this assessment, the supervisor will seek input from the Nurse and give due weight to the Nurse's own skills checklist when available. To increase the availability of Nurses qualified to float outside the designated region, the Employer will provide orientation programs to those Nurses who have volunteered a specific commitment. Maintaining and fulfilling a specific commitment to float outside the Nurse's

designated region shall count as a committee/project toward the Nurse's Clinical Nurse criteria.

16.3 Except in the case of unforeseeable circumstances resulting in critical patient care needs, Nurses will not be floated outside her/his designated region in the absence of her/his commitment to do so. Such assignment shall be consistent with nursing supervision's assessment of the individual Nurse's competence and skills and the Nurse will not be required to perform any patient care assignments which exceed her/his qualifications and skills. If the Nurse asserts in good faith that s/he is not qualified to handle the assignment, s/he may request that the assignment be continued only until a qualified Nurse is available and assigned. The Employer will make reasonable efforts to identify a qualified Nurse for the assignment. If a qualified Nurse is available and assigned the Nurse may be excused from the remainder of the assignment and given absent time.

16.4 Floating to Different Work Locations/Multi-Location Assignment.

- 16.4.1 When the Employer floats a Nurse who is regularly assigned to a location or locations to another location to which the Nurse is not regularly assigned to work on the day in question, the Employer must seek volunteers and will make every effort to provide a minimum of two (2) hours notice.
- 16.4.2 For Nurses who are regularly assigned to work at the main medical center campus and are floated to work at a different location, the Employer will reimburse bridge tolls and mileage incurred in traveling to that location as provided by applicable law. For Nurses who are regularly assigned to work at a location or locations other than the main medical center campus, and who work at more than one location during a shift, the Employer will reimburse bridge tolls and mileage incurred in traveling between the locations during the shift as provided by applicable law.
- 16.4.3 Except in cases of critical staffing shortages, as determined by the Employer, a Nurse shall not be required to float to a different work location more than one time per shift or to a location more than thirty-five (35) miles from the Nurse's assigned work location for that shift. In addition, a Nurse who does float to a different work location shall not be required to return to the original location within that shift.
- 16.4.4 Nurses assigned to locations other than the main medical center campus may be assigned to multiple locations. Such Nurses who travel to or between regularly assigned locations are not deemed to be "floating" for purposes of this section, unless a Nurse is assigned to travel to such a location on a day on which, or at a time at which, the Nurse was not otherwise scheduled to work at that location. Nurses may be entitled to paid travel time and reimbursement for mileage, pursuant to applicable law. Effective May 8, 2019, the posting for a new position that requires assignment to multiple locations shall state the assigned locations known as of the time of the posting and whether additional locations may be assigned that are farther than thirty-five (35) miles from other assigned locations. Effective May 8, 2019, except in cases of critical staffing shortages, a Nurse in a position that has multiple location assignments may not be regularly assigned to work in a location farther than thirty-five (35) miles from the Nurse's primary work location to which she or he is assigned, absent the Nurse's agreement to such assignments. A Nurse's primary work location is that location at which a Nurse works more of her or his time than at any other location.
- 16.4.5 For the purposes of this section:

- a. The main medical center campus shall be considered to be one work location; and
- b. Nurses assigned to the float pool shall be considered to be assigned to the main medical center campus, except for those float pool Nurses who have been assigned by the Employer in writing to another work location on a regular and ongoing basis.

SECTION 17 WEEKEND STAFFING

17.1 <u>Weekend Defined</u>. A weekend is defined as Saturday and Sunday for the day and evening tours of duty. Shifts beginning Friday and Saturday, or Saturday and Sunday, as designated in writing by the individual Nurse will be considered a weekend for the night tour of duty to the extent effective scheduling permits. All such designations will remain in effect for a period of six (6) months. The Employer will use its best efforts to grant full-time and part-time Nurses working the night tour of duty their preferred weekend designation.

17.2 Guarantee of Weekends Off.

- 17.2.1 The Employer will use its best efforts to grant each full-time and part- time Regular Nurse at least every other weekend off. The Employer will guarantee that each Regular Nurse will not be required to perform any work on more than two (2) consecutive weekends without premium pay. If the Employer requires a Regular Nurse to perform any work on more than two (2) consecutive weekends, the Nurse will receive premium pay for work performed on the third (3rd) consecutive weekend and each successive weekend worked until granted a full weekend off. Premium pay will be paid in the pay period during which the additional weekend is worked. The premium pay referred to herein is one-and-one-half (1-1/2) times the Regular Nurse's hourly base wage plus any applicable shift differential. When a Regular Nurse is working overtime hours on a weekend, the Nurse shall be compensated at overtime rates in lieu of the premium rate provided herein. For purposes of this paragraph, "work" means engaged in patient care at Stanford Health Care. Schedules shall be posted for four (4) week periods at a time.
- 17.2.2 Absence from work on a weekend will not be required to be "made up" unless at the Employer's discretion it is considered necessary for the purpose of applying corrective action for excessive absenteeism.

17.3 Exceptions.

- 17.3.1 In the event of a major catastrophe, the Employer may temporarily suspend the weekend off and premium pay policies provided herein.
- 17.3.2 A Nurse may voluntarily trade weekends with another Nurse with the written consent of the Nurse Manager or CNS and the schedule planner. Time worked pursuant to an approved trade shall not be considered in determining schedule or computing premium pay under provision 17.2.
- 17.3.3 Regular Nurses who request and receive approval to be scheduled in a manner that would result in premium pay for working weekends under 17.2 or who request and receive approval to be scheduled weekdays off in lieu of a weekend off during a work week will not be eligible for premium pay under this Section.

SECTION 18 SHIFTS AND SHIFT ROTATION – REGULAR NURSES

The final determination of appropriate shifts is one of the rights reserved to the Employer as defined in Section 40 of this Agreement. CRONA recognizes that the Employer has the right to determine a variety of appropriate shifts according to the operational and budgetary needs of the Employer.

- 18.1 <u>Shift Rotation</u>. Any Regular Nurse assigned to rotating shifts who is scheduled, unless by request, to work within eight (8) hours after the end of a scheduled shift shall be paid at overtime rates for the entire second shift.
- 18.2 <u>Regular Shifts</u>. Regular shifts may consist of four (4), six (6), eight (8), ten (10), or twelve (12) consecutive hour periods worked between specific starting and ending times established by the Employer. Special requirements for units where only twelve (12) hour staffing patterns are in effect are set forth in 18.6.
- 18.3 <u>Discussions with CRONA</u>. The Employer will advise and update CRONA of the status of current and planned unit shift patterns in Joint Conferences pursuant to Section 32. The Employer will give thirty (30) days notice to CRONA of anticipated changes in shift patterns where change involves shifts of less than eight (8) hours. Such shifts shall be posted and filled on a voluntary basis only.
- 18.4 <u>Schedules will be posted for four (4) week periods</u>. These schedules will be posted at least two (2) weeks (which includes the day of the posting) prior to the beginning of the next schedule. The Employer will make a good faith effort to post the Thanksgiving and Christmas/New Year's schedules at least four (4) weeks in advance of the schedule period. No later than July 10, 2022, the posted schedule shall reflect when a Nurse is scheduled to work over the Nurse's minimum regular commitment.
- 18.5 <u>Breaks</u>. The Employer provides paid breaks and unpaid meal breaks as follows:

Shift	Breaks
Four (4) – Six (6) Hours	One (1) paid fifteen (15) minute break
Eight (8) – Ten (10) Hours	Two (2) paid fifteen (15) minute breaks One (1) unpaid thirty (30) minute meal break
Twelve (12) Hours	Three (3) paid fifteen (15) minute breaks One (1)
	unpaid thirty (30) minute meal break

- 18.6 <u>New Twelve (12) and Twelve/Eight (12/8) Hour Shift Patterns</u>. Before implementing new twelve (12) or twelve/eight (12/8) hour shift staffing patterns on units where prior to the execution date of this Agreement other patterns were in effect, the Employer will follow the following procedures:
 - 18.6.1 Department of Nursing and CRONA will agree upon a date for a meeting of all Regular full and part time Nurses on the affected unit to discuss the implementation of a twelve (12) or twelve/eight (12/8) hour staffing pattern for that unit. A representative of CRONA will be invited to attend and participate in the discussion.
 - 18.6.2 Within three (3) days of the discussion meeting Department of Nursing and CRONA will agree upon a time for a secret ballot vote by all full and part time Regular Nurses on the affected unit. In no event will the vote be scheduled more than ten (10) days after the meeting referenced in 18.6.1. The wording of the ballot will be subject to mutual agreement between CRONA and the Employer and a

representative of CRONA will be present to assist in the vote tally. A sixty-six and two-thirds percent (66-2/3%) majority of the eligible staff must vote in favor of the twelve (12) or twelve/eight (12/8) hour staffing pattern to constitute approval of the staffing plan for the unit.

- 18.6.3 It is understood that provision for some Regular Nurses to work less than a twelve (12) hour shift on a unit voting in favor of the twelve (12) hour plan will be made only to the extent deemed operationally feasible by Department of Nursing. Should Department of Nursing determine that no provision for less than twelve (12) hours can be made this fact will be announced and the reasons for the decision discussed in the meeting described in 18.6.1.
- 18.6.4 The Employer will make best efforts to assist any Nurse who is unwilling or unable to participate in a twelve (12) hour staffing pattern in arranging placement on another nursing unit. If the former staffing pattern is resumed within six (6) months any Regular Nurse so displaced will be given the option to return to the unit s/he left provided a position is available.
- 18.6.5 For the purpose of a twelve (12) hour day, shifts will be determined to be from 6:45 a.m. until 7:15 p.m. and 6:45 p.m. until 7:15 a.m. unless a different starting time is established based on the staffing needs of a particular unit. The shift commencing at 6:45 a.m. will be known as the day shift and no differential will be paid. The shift beginning at 6:45 p.m. will be known as the night shift and Nurses working that shift will receive the applicable night shift differential for all hours worked.
- 18.6.6 All PTO hours will be taken in twelve (12) hour segments, except for Nurses regularly scheduled for partial shifts, who will take and be scheduled for PTO in the same manner they are normally scheduled to work.
- 18.6.7 Except as provided in this section, all other accruals will continue as defined in this Agreement. Pursuant to Section 7, a Regular Nurse who is scheduled for and works a twelve (12) hour shift who is requested to work prior to the normal starting time of her/his assigned shift will be paid at double time for all early hours worked. A Nurse working overtime contiguous to a regular shift will receive the higher shift differential of the two (2) shifts worked for the overtime hours.
- 18.6.8 Educational hours will continue to be granted on an eight (8) hour basis, and as otherwise defined in this Agreement. Nurses regularly scheduled to a 10 or 12 hour shift may use PTO or "A" time for the remainder of the shift.
- 18.6.9 The Employer will make best efforts to ensure that Nurses working a twelve (12) hour schedule are not required to work more than three (3) consecutive days.
- 18.7 <u>Eight (8)/Twelve (12) Hour Shifts</u>. The Employer acknowledges the importance of and CRONA's interest in maintaining the eight (8)/twelve (12) hour shift mix and will endeavor to maintain them as long as it remains operationally feasible.
- 18.8 <u>Twelve (12) Hour Fill-ins</u>. Nurses may fill in and be paid as twelve (12) hour Nurses for periodic scheduling vacancies in twelve (12) hour shift positions.
- 18.9 <u>End of Established Staffing Patterns</u>. Subject to 18.7, in the event the Employer plans to end an established ten (10), twelve (12) or twelve/eight (12/8) hour staffing pattern on any unit where such patterns are in effect the Employer will notify CRONA of its decision in writing

and will meet with CRONA upon request to discuss any effects upon Nurses represented by CRONA.

SECTION 19 LEAVES OF ABSENCE – REGULAR NURSES

- 19.1 <u>Eligibility</u>. Regular Nurses covered by this Agreement who have successfully completed the trial period and who expect to be absent for family, personal, medical (including pregnancy related disability), military, or educational reasons are eligible to apply for a leave of absence. Requests for review of leaves of absence by trial period Regular Nurses will be considered by the Employer on a case by case basis. Granting or denying such requests shall be at the sole discretion of the Employer unless otherwise required by law and not otherwise reviewable under the terms of the Agreement. A granting of a leave to a Regular Nurse in the trial period shall be considered as an automatic extension of the trial period for the period of leave granted. If an approved leave of absence is granted for a period that is less than seven (7) days and the absence exceeds seven (7) days, a leave of absence must be requested for the time exceeding seven (7) days and, if approved, all of the time off whether paid or unpaid will be considered part of a leave of absence.
- 19.2 <u>Bereavement Leave</u>. Regular Nurses will be eligible for up to three (3) working days of pay in lieu of scheduled hours in the event of a death in the Nurses' immediate family, including parents, legal guardians, spouse, children, stepchildren, grandparents, grandchildren, siblings, step-siblings, step parents, mother-in-law, father-in-law, son-in-law, daughter-inlaw, and eligible domestic partner as defined in the Employer's Health Benefits Summary Plan Descriptions. Additional time off may be granted at the discretion of the Employer. Time off for deaths outside the "immediate family" may also be granted at the discretion of the Employer. If available, accumulated PTO will be used for time off.

A Relief Nurse will be entitled to seven (7) calendar days off without pay following the death of an "immediate family" member. Additional time off may be granted to the Relief Nurse at the discretion of the Employer.

Bereavement leave will be granted immediately following the death, unless funeral or memorial arrangements require other dates, which must be approved by the Nurses' manager. The Employer may require reasonable documentation of the death, and/or of the funeral or memorial arrangements requiring dates other than those immediately following the death.

A Regular Nurse will be eligible to take prescheduled vacation, even if the Nurse has insufficient PTO, if the insufficient PTO is a result of the use of PTO for bereavement, as per section 19.2. This will apply if the insufficient PTO is a result of PTO for bereavement which occurred following the scheduling of the vacation.

19.3 Leave Categories.

- 19.3.1 Educational Leave.
 - a. The purpose of educational leave is to grant Regular Nurses time off for the pursuit of education pertinent to her/his employment at Stanford Health Care. Job relatedness of any particular education program is to be determined by the definitions set forth in Section 14 of this Agreement.

- b. Regular Nurses who have satisfactorily completed one (1) year of regular fulltime employment, or eighteen (18) months of part- time employment of at least 0.5 time, or eighteen (18) months of combined full-time and part-time employment of at least 0.5 time, are eligible to apply for educational leave.
- c. Educational leave must be approved by the Nurse's supervisor and the Vice President of Patient Care.
- d. If an educational leave of absence exceeds one-hundred eighty (180) calendar days, the Regular Nurse's hire and review dates will be adjusted forward for the entire period of unpaid time off which exceeds one-hundred eighty (180) calendar days.
- 19.3.2 <u>State and Federal Family and Medical Leave</u>. Nurses who have been employed by Stanford Health Care for at least twelve (12) months and have worked for the Employer at least twelve hundred fifty (1250) hours during the previous twelve (12) months are eligible for Family/Medical Leave in accordance with State and Federal laws:
 - birth of the employee's child;
 - the placement of a child with the employee as a result of adoption or foster care;
 - the care of a spouse, domestic partner, child or parent with a serious health condition;
 - a serious health condition that makes the employee unable to perform her/his job.
- 19.3.3 <u>Medical Leaves of Absence (excluding pregnancy related disability)</u>. Regular Nurses who have passed the trial period who are not eligible for State and Federal Family and Medical Leave are eligible to apply for medical leave.
 - a. Regular Nurses who become ill, injured, or otherwise medically incapacitated, and who expect to be absent for more than seven (7) consecutive calendar days, must request a medical leave of absence using the intake process established by the Employer and submitting required Certification of Healthcare Provider for Employee's Serious Health Condition (Family and Medical Leave Act) Form. A medical leave of absence must be requested as soon as the Nurse is aware of the need for the leave.
 - b. The form must be completed in its entirety. If these dates change or if any different limitations are imposed by the physician, the Regular Nurse must furnish an updated Certification to the Employer or its designee. It is the Nurse's responsibility to inform her or his personal physician of the normal job requirements in order to assist in determining the cease-work date, work limitations, and the return to work date. A final return to work date will be determined after review of the Return to Work Authorization on the Physician's Letter signed by the Regular Nurse's personal physician. A clearance must be obtained by the employee from Occupational Health Services, or from another entity the Employer has informed the Nurse is responsible for providing clearance, for work related injury and illness, or infectious disease

prior to return to work. At the discretion of the employee's supervisor an authorization for return to work may be required for other injury or illness prior to the employee's return to work. The Nurse cannot return to work until cleared in accordance with the above.

- 19.3.4 <u>Pregnancy Related Disability Leave</u>. All female Regular Nurses covered by this Agreement are eligible for pregnancy related disability leave. The leave is granted at the request of the Regular Nurse, for the period of time during which she is incapable of performing her job because of medical disability resulting from pregnancy, delivery, or post-childbirth recovery. The Regular Nurse may take the leave without pay or use accrued PTO during any waiting period before ESL or state Disability becomes available, and may use PTO for any period after ESL is exhausted, or integrated with any State Disability income received, not to exceed the Regular Nurse's normal pay. The leave may be as long as six (6) months when a verified disability exists. If the Nurse remains disabled following the end date of the leave the situation will be treated as any other non-work-related medical disability as defined in this Section.
- 19.3.5 <u>Parental Leave of Absence (Non-Disability</u>). All Regular Nurses covered by this Agreement are eligible to request parental leave related to the birth or adoption of her/his child. If the Regular Nurse is not eligible for State and Federal Family Leave (see 19.3.2), a personal leave without pay may be granted for up to one-hundred eighty (180) days following childbirth or adoption or upon the conclusion of the Nurse's pregnancy related disability, if any. A personal leave for parental leave purposes will not be unreasonably denied. The Nurse may take the leave without pay or use accrued PTO.
- 19.3.6 Personal Leave of Absence.
 - a. Personal leave of absence may be granted at the discretion of the Regular Nurse's Manager. A personal leave of absence may not be granted for other compensated employment. Unless approval is granted pursuant to 19.8.3, below, unpaid Personal Leave will be granted only after PTO, including Cashout PTO, if any, has been exhausted.
 - b. Applications for personal leave of absence must be made using the intake process established by the Employer and must state the specific reason for the time off requested.
- 19.3.7 <u>Military Training Leaves</u>. When a Regular Nurse is required to perform annual military training duty or is called to active duty, a leave will be granted in accordance with Federal Law. A Regular Nurse will upon request provide the Employer copies of her or his induction papers, active duty orders and orders to report for active annual duty training.
- 19.4 <u>Duration of Leave</u>. Leaves of absence may be granted up to the following maximum amounts of time:
 - Personal One-hundred eighty (180) days with a possible one-hundred eighty (180) day extension. Educational One (1) calendar year.
 - Military A leave will be granted in accordance with federal law.
 - State and Federal Family and Medical Leave As required by Federal and State laws.

- Medical (including pregnancy related disability) Up to one-hundred eighty (180) days. A personal leave of absence may be requested up to an additional one-hundred eighty (180) days.
- Parental leave One-hundred eighty (180) days.
- 19.5 <u>Combinations of Leaves of Absences</u>. Excluding military and educational leaves of absence, no combination of leaves shall be granted within any three hundred sixty-five (365) day period, measured from the first day of the first such leave, which considered together exceed one hundred eighty (180) days unless a personal leave of absence extension of up to six (6) months is granted at the sole discretion of the Regular Nurse's supervisor with the approval of the Vice President of Patient Care. Regular Nurses with fifteen (15) years or more of service shall be granted an additional thirty (30) days of medical leave if the Nurse's evaluations have been satisfactory and there are no corrective <u>actions</u> in the Nurse's personnel file.

19.6 Reinstatement Rights.

- 19.6.1 In the case of State and Federal Family and Medical leaves of absence, Medical, pregnancy related disability, personal and parental leaves of absence (up to one-hundred eighty (180) days), Regular Nurses will be offered the same or similar positions. Similar includes the same classification, pay, and benefits, but not necessarily the same shift and/or unit to which the Nurse was assigned prior to the leave of absence.
- 19.6.2 Reinstatement is not guaranteed for personal leaves over one-hundred eighty (180) days or any combination of leaves over one-hundred eighty (180) days and educational leaves, but a Regular Nurse will be offered the same or any similar existing vacancy upon returning from the leave.
- 19.6.3 Return from military leave shall be in accordance with federal law provided that the Regular Nurse requests reinstatement within the time provided by federal law.
- 19.7 <u>Working Leave Status</u>. A Regular Nurse on any leave of absence may work on a Relief basis when and if her/his condition allows it and if the department has need for such Relief employment.

19.8 Procedures.

- 19.8.1 A Regular Nurse shall request a leave of absence using the established intake process thirty (30) days in advance of the desired starting date, except in the case of an extreme emergency. The request shall state the specific type of leave and provide supporting information as established in this Section and the Employer's policies, and the dates of the Leave.
- 19.8.2 A Regular Nurse who is granted a leave of absence will be informed on the Leave of Absence Request and Authorization the dates of the Leave, benefit limitations, insurance procedures, and the requirements for and the conditions under which the Nurse may return to work.
- 19.8.3 At the discretion of the Nurse's Manager, it is not necessary for a Regular Nurse to completely exhaust PTO before a personal leave is granted.

- 19.9 A Regular Nurse shall not forfeit any accrued benefits during an authorized leave of absence, or accrue any benefits during such leave. A Regular Nurse's Adjusted Hire Date will be changed for a leave of absence of more than six (6) months, except as provided in 19.10, below.
- 19.10 Subject to the terms and conditions established by the Employer's insurance plans, Regular Nurses on authorized Medical leave of absence or on State and Federal Family and Medical leaves of absence and who have Employer-paid premiums under the Employer's Group Medical. Dental, and Basic Life Insurance, described in Section 9, will continue to have such group coverage premiums paid by the Employer if they continue to pay employee contributions. Regular Nurses on any other authorized leave do not receive Employer contributions but may pay the entire required premium for continuous group coverage for themselves, their spouse/domestic partner and dependents. The Regular Nurse may arrange for continued coverage by making monthly premium payments through the Benefits Office, subject to the terms and conditions established by the Employer's insurance plans. Participation will continue in Retirement Plans and time lost due to a Worker's Compensation injury or occupational illness will count toward vesting requirements for Regular Nurses covered by the Employer's Pension Plan. Further, a Regular Nurse on an authorized Medical leave of absence due to a Worker's Compensation injury or occupational illness status shall not suffer any loss of seniority. The original date of hire or the adjusted hire date (if previously adjusted) and the next review date shall be maintained. If a Regular Nurse is still absent when eligible for review, the review will be postponed until the Nurse returns to work.
- 19.11 Any Regular Nurse covered by this Agreement will have the option to purchase Employer offered medical and dental insurance at group rates for up to one (1) year while on an approved Educational Leave of Absence.
- 19.12 Return from Leave.
 - 19.12.1 <u>Advance Notice</u>. Regular Nurses on approved leave of absence are expected to return to work on the first scheduled work day following the expiration date of the leave. In the event a Regular Nurse is unable to return for any reason, or is unable to perform any or all of the essential functions of the job, s/he must contact the Leave Coordinator at least two (2) weeks in advance of the expiration of the leave.
 - 19.12.2 <u>Failure to Return</u>. If a Regular Nurse fails to return to work at the expiration of a leave of absence, s/he will be deemed to have resigned her/his employment.

SECTION 20 CHRISTMAS AND NEW YEAR'S HOLIDAY

- 20.1 All Regular Nurses shall be given either the Christmas or New Year's holiday off, unless the Nurse requests to work both holidays. Christmas Eve and/or New Year's Eve may be considered as the holiday upon request, and/or may be given in conjunction with the holiday according to staffing needs.
 - 20.1.1 No more often than once every calendar year before September 1st, each unit that is open on the winter holidays has the option to vote on whether to "bundle" the winter holidays with the holiday eves for scheduling purposes for that year. Such a vote may be requested by any Nurse on the unit by making a written request to the unit manager, using a request form provided by CRONA. Such a vote shall be held within two (2) weeks of the request being made. The unit manager shall conduct

the vote among the Nurses assigned to the unit. If two-thirds (2/3) of the Nurses assigned to the unit vote to approve the "bundling", then for that year the Employer shall grant off each Nurse assigned to the unit either (a) the Holiday Shifts on Christmas and Christmas Eve, or (b) the Holiday Shifts on New Year's and New Year's Eve, unless the Nurse requests to work both holidays. A Nurse granted off the Holiday Shifts on Christmas and Christmas and Christmas Eve pursuant to this provision shall be available to be assigned to work on New Year's Eve and New Year's Day, and vice versa. For purposes of this provision, the Holiday Shifts are the shifts occurring during the 48-hour period commencing with the start of the Nurse's shift on the day before each holiday (i.e., Christmas Day or New Year's Day).

- 20.2 It shall be the responsibility of the Regular Nurse to request, according to established departmental policy, her/his preference with regard to this holiday time, including choice of holiday and whether it should be scheduled as a regular day off or PTO time. If it is not feasible to grant the holiday as a day off, PTO or "A" time shall be used.
- 20.3 The Employer shall, on the basis of the Employer's needs, decide which of the holidays shall be granted.
- 20.4 In those instances where the Employer does not grant the holiday requested, the other holiday shall be granted automatically.
- 20.5 Thanksgiving shall not be considered in determining Christmas/New Year's holidays.
- 20.6 Both holidays may be granted, subject to individual request and Employer approval.
 - 20.6.1 Individual Regular Nurses may request additional days off in conjunction with the Christmas or New Year's holidays. Except in unusual circumstances, a request for more than seven (7) additional calendar days off will not be considered. The Employer shall make the final determination whether any request shall be granted.

SECTION 21 JURY DUTY AND LEGAL APPEARANCES

- 21.1 Jury duty as used in this Agreement is defined as that time a Regular or eligible Relief Nurse is required to spend sitting on a jury or physically waiting at the courthouse in anticipation of being called to sit on a jury. Jury duty does not include time spent away from the courthouse on "telephone alert" or other forms of standby service not requiring a Nurse's physical presence at the courthouse.
- 21.2 There will be no loss in regular wages for time spent away from work when an eligible Nurse is called for jury duty or for legal appearances on behalf of the Employer. PTO accrual will continue for all such hours that the Nurse would normally have been scheduled to work.
- 21.3 Wage payments will not be made for jury duty on any day on which an eligible Nurse had been previously scheduled for PTO, or a Leave of Absence. Regular and eligible Relief Nurses will be paid at their hourly base wage rate for time spent for legal appearances on behalf of the Employer. If a Regular Nurse had previously approved PTO or time off scheduled prior to being notified of jury duty service, upon request of the Nurse, the PTO or time off shall be cancelled if the request for cancellation is provided within a reasonable time of the Nurse's receipt of the notice for jury duty service. Cancellation of previously approved PTO or time off is not intended to occur on a day-to-day basis. Proof of completion of the time spent as a juror must be submitted to the supervisor at the time of the request.

- 21.4 Payments received from the Federal government, State or County, are kept by the eligible Nurse to help defray expenses.
- 21.5 The eligible Nurse should notify the supervisor immediately upon receipt of the summons so that arrangements can be made for the absence. Upon request by the supervisor proof of completion of jury duty will be provided.
- 21.6 Eligible Nurses required to participate in jury duty, grand jury duty, or legal appearances as defined in this Section shall not be scheduled to work more than ten (10) days in every fourteen (14) day pay period with each day spent in jury duty or legal appearances considered a day worked for the purposes of this Section. Regardless of shift, the Nurse shall be relieved of her/his duties on the date s/he is required to be on jury duty or make legal appearances. Night shift eligible Nurses will be provided paid jury duty release time either the night before serving on jury duty or the night of the jury duty service, if scheduled, at the Nurse's discretion. If the eligible night shift Nurse opts for paid jury duty release time the night before serving on jury duty, the Nurse may use PTO or "A" time the night of jury duty service. This provision does not apply for on-call jury duty status.
- 21.7 Jury Duty and Legal Appearance procedures for Relief Nurses is also covered in Section 15.14.8.

SECTION 22 TIME OFF FOR CRONA OFFICERS

- 22.1 The CRONA president and two designated officers shall be allowed reasonable time with pay each pay period not to exceed a total of thirty-two (32) hours for the three (3) for the purpose of conducting CRONA business with the Employer. Different officers may be designated by CRONA to use this time off on a rotating basis. Paid time under this subsection shall not be considered work time for the purposes of calculating overtime. CRONA will notify Nursing Administration after the election of officers as to the distribution of the CRONA hours. In the event that the distribution of the CRONA hours changes due to officer absence, CRONA will notify Nursing Administration prior to the pay period that the change occurs.
- 22.2 If requested by the officer, the paid time provided for in this subsection will be scheduled and taken within the officer's regularly scheduled hours (FTE). The time shall be scheduled in advance by mutual agreement between the officer and his or her supervisor in increments of at least one (1) hour. All benefits shall accrue on these hours, including shift differential and education benefits, and these hours shall be considered as pension eligible earnings under Section 12.1.3.c.
- 22.3 Alternatively, if requested by the officer, and upon advance notice to his or her supervisor, the paid time provided for in this subsection may be paid in addition to the officer's regularly scheduled hours (FTE). For such paid time, benefits shall not accrue.
- 22.4 CRONA shall provide the Employer with a list of its officers and apprise the Employer of any changes during the term of this Agreement.
- 22.5 CRONA's Nurse designees shall not be unreasonably denied Absent days and/or the use of PTO to attend Nurses' Association Conventions. CRONA recognizes that the decision to approve or deny either Absent days and/or PTO to attend such conventions because of scheduled work and staffing assignments is within the sole discretion of the Employer. CRONA shall notify the Employer of the identity of their designees and the date of the conventions, at least thirty (30) days in advance of such leave.

SECTION 23 VACANCIES AND INTERNAL TRANSFERS

CRONA recognizes that the Employer announces job vacancies, recruits and hires in accordance with general procedures established by the Employer. To assist Nurses applying for transfer to Nurse vacancies established by the Employer, the following procedures shall be applicable to both Regular and Relief Nurses:

- 23.1 The Employer shall post all current lists of Nurse vacancies in the following order and manner:
 - 23.1.1 The Hospital shall send an email (via Outlook or a similar hospital-wide email system) identifying the vacancy to the Nurses on the unit in which the vacancy exists.
 - 23.1.2 At least five (5) calendar days after the foregoing notice has been sent and if the vacancy has not been filled with a Nurse on the unit, the Hospital shall post the vacancy on the online application system. The posting on the online application system shall contain at least the following information about the vacancy:
 - a. Commitment or full-time equivalent (FTE);
 - b. Shift and shift length;
 - c. Whether on call is currently required;
 - d. Name and description of the type of unit;
 - e. Location of the position, including all information about assignments to multilocations required under Section 16.4.4; and
 - f. That the vacancy is part of the CRONA-represented bargaining unit.
- 23.2 For purposes of this section, a "vacancy" shall be defined as an available Regular Nurse position, the addition of available hours which would result in a change to a Regular Nurse's commitment (including available hours arising from relocating hours from one shift to another), and an available Relief position. A position or hours shall not be deemed to be "available" until approved by the Employer. Hours worked over a Nurse's commitment would not be deemed a "vacancy" under this section.
- 23.3 Eligible Regular and Relief Nurses of the Employer who make a transfer request are given preference if qualified, as determined by the Employer, for vacancies in the following order and manner:
 - 23.3.1 Nurses who both (i) are assigned to the unit in which the vacancy occurs as of the date of the notice of the vacancy, and (ii) notify their manager (or designee) of their interest in the vacancy within the five (5) day period after notice is provided, shall be given preference over other Nurses. As between qualified applicants assigned to the unit in which the vacancy exists, if qualifications are determined to be substantially equal, seniority shall be the determining factor.
 - 23.3.2 Current Regular and Relief Nurses of the Employer, including Nurses on the unit in which the vacancy exists who did not notify their manager of their interest in the

vacancy within the five (5) day period, who apply for the vacancy shall be given preference over external applicants if the qualifications of the current Nurse and external applicants are determined to be substantially equal.

- a. As between qualified current Nurse applicants, if qualifications are determined to be substantially equal, seniority shall be the determining factor.
- b. As between qualified external applicants, if qualifications are determined to be substantially equal, Regular and Relief Nurses who are currently employed by Lucile Salter Packard Children's Hospital at Stanford shall be given preference over other external applicants.
- c. For the purposes of this section, "external applicants" shall be applicants who are not currently employed at the Hospital in a position in the bargaining unit.
- 23.4 Transfers pursuant to this Section 23 shall include Relief Nurse applications for vacancies. When applying for vacancies, "B", "C", and "D" Relief Nurses will be credited with seniority as follows, provided there has been no break in employment as set forth in Section 37:
 - a. the full period the Relief Nurse has been employed by the Employer as a Nurse, and
 - b. at a ratio of two (2) years of service for one (1) year of seniority credit for all periods of employment as a Relief Nurse, provided the Relief Nurse maintained her/his commitment during such periods.

As an example, a Nurse who has worked 4 years as a Nurse and then, without a break in employment, has continued to work an additional 10 years as a "B" Relief Nurse will be credited with a total of 9 years seniority for the purposes of Section 23.

23.5 Transfers pursuant to this Section 23 shall occur on the first day of the pay period that is agreed upon by the releasing manager and the new manager. No Nurse shall be denied a transfer based on a failure of the releasing and new managers to agree on the transfer date.

SECTION 24 REDUCED COMMITMENTS

- 24.1 <u>Voluntary Requested Reduction in Scheduled Commitment</u>: Any Regular Nurse may submit a request in writing to her or his Nurse Manager requesting a voluntary reduction in schedule commitment. The Nurse Manager will consider the request and will respond in writing to the Nurse within thirty (30) calendar days of the submission of the request. When the Nurse Manager determines in her or his discretion that a reduction in schedule commitment may be appropriate, the reduced commitment position shall be posted pursuant to the Internal Transfers procedures contained in Section 23 of the Agreement. The parties recognize that, even when the Nurse Manager agrees to post a reduced commitment position, the Nurse who obtains the position may not be the Nurse whose request prompted the posting of the reduced commitment position.
- 24.2 Vacant part time positions that the Employer decides in its discretion to re-post shall be posted internally as Regular part time positions with the same or lower commitment level pursuant to the provisions of Section 23 (Vacancies and Internal Transfers). The Employer may re-post such a position on the same or a different shift. If the position is not filled by an internal applicant within twenty-one (21) days of posting on the online application system

pursuant to Section 23.1.2, the Employer may (but is not required to) re-post the position as a full-time position, pursuant to the provisions of Section 23 (Vacancies and Internal Transfers). This provision (Section 24.2) shall not apply to units with either fewer than five (5) Regular Nurses or in which more than forty percent (40%) of the headcount of Regular Nurses consists of regular part time positions. This provision (Section 24.2) shall not apply to a unit with five (5) to nine (9) Regular Nurses that has two (2) or more Regular Nurses in part time positions. For example, a unit with six (6) Regular Nurses, two of whom are part time Nurses, shall not be required by this provision to re-post a part time position if only one of the part time Nurses leaves her or his position.

- 24.3 The Hospital has an interest in staffing its units (including clinics) with a mix of full time and part time regular positions.
 - 24.3.1 To that end, the Hospital will staff its units using a standard that no more than eighty percent (80%) of the headcount of Regular Nurses in each unit at the Hospital consists of regular full time positions. This standard shall not apply to any unit in which there are fewer than ten (10) Regular Nurses.
 - 24.3.2 To the extent a unit at the Hospital has a staffing mix of Regular Nurses above this standard, the Hospital, upon the written request of a regular Nurse in that unit, shall re-post existing full time position(s) and, pursuant to the provisions of Section 23, shall, within forty-five (45) days of the request, post as part time position(s) in a sufficient number to bring the number of full time positions in the unit to eighty percent (80%) or lower.
 - a. To the extent operationally feasible, the Hospital shall offer a mix of part time positions.
 - b. A Nurse who has accepted a part time position posted pursuant to this paragraph shall be placed on a schedule reflecting her or his reduced commitment as soon as practicable, but in no event later than sixty (60) days after the Nurse's acceptance of the position, absent extraordinary circumstances or mutual agreement of the Nurse and the Nurse's manager to extend this sixty (60) day period.
- 24.4 For purposes of this Section 24, a "part time" position is defined as a regular position with a FTE equal to or lower than .80, and a "full time" position is defined as a regular position with a FTE of greater than .80.

SECTION 25 PAYROLL PROCEDURES

- 25.1 <u>Direct Deposit</u>: The Employer shall, upon written authorization by the individual Nurse, deposit the Nurse's pay in a bank of the Nurse's choice that is located in the State of California, any bank in the United States that is a member of the Federal Reserve Banking System, or the Stanford Federal Credit Union. For nurses who do not elect direct deposit, paychecks will be mailed to the Nurse's address on file on the Wednesday following the end of the two (2) week pay period ending the previous Saturday. A documentary record of direct deposits will be available online, or if requested, will be mailed to the last mailing address provided by the Nurse.
- 25.2 <u>Late and Missing Time Cards</u>: Wages owed due to late or missing timecards or time entries for an entire pay period will be paid on Monday following the normal Friday pay day.

25.3 Correction of Payroll Errors:

- 25.3.1 An underpayment in a paycheck of pay for eight (8) or more hours due to an error by the Employer shall be remedied no later than five (5) business days after the error has been reported, unless Payroll has not been able to verify the error during that time period. Such payments shall be made by check or direct deposit, in accordance with the Nurse's election pursuant to Section 25.1.
- 25.3.2 Other underpayments and overpayments will normally be rectified on the paycheck following the detection of the error. Repayment plans may be developed by the Payroll Office if an amount to be repaid would cause a hardship to the Nurse. All repayment plans must ensure the return of the full amount prior to completion of any employment contract or termination of employment.
- 25.3.3 Payroll shall make diligent efforts to verify and correct any payroll errors that are reported to or otherwise become known by Payroll. If Payroll is not able to verify an error covered by Section 25.3.1 within the designated time period, the Nurse(s) who reported the error and CRONA shall be informed within the designated time period of the reasons for the delay in resolving the error.
- 25.4 <u>Paycheck Questions</u>: Each Unit manager shall notify the Nurses in the Unit of a designated person to whom questions from Nurses regarding paychecks should be directed. Nurses must advise the designated person of any error in their paychecks promptly.

SECTION 26 NURSE INFORMATION CENTERS

- 26.1 The Employer will provide a conveniently located bulletin board in the Main Center to be used for posting Employer and CRONA communications with Nurses covered by this Agreement. Designated CRONA representatives shall be responsible for posting CRONA materials. It is agreed that these boards will be the sole location for all such posted communications. In addition, each Nursing Unit will have designated space available on its bulletin boards for CRONA postings.
- 26.2 The Employer shall maintain on-line information for the following programs:
 - 1. Retirement Plan;
 - 2. Medical, dental and vision insurance;
 - 3. Flexible spending accounts;
 - 4. Educational expense reimbursement; and
 - 5. Other information as may from time to time be agreed upon by CRONA and the Employer.
- 26.3 At the beginning of each calendar year the Employer shall furnish to CRONA a list of all ongoing Department of Nursing standing committees. The Employer will inform CRONA in Joint Conferences of Nurse vacancies on these committees as they occur.

SECTION 27 PERFORMANCE EVALUATION

The Employer provides Nurses performance appraisals periodically and/or in conjunction with any change in a Nurse's classification in accordance with procedures established by the Employer. Regarding written performance evaluations, CRONA and the Employer agree as follows:

- 27.1 The Nurse's supervisor will advise the Nurse thirty (30) days prior to the annual review date that the evaluation is due on the review date. The Nurse will provide the required documentation/paperwork to his/her supervisor within fourteen (14) days of receiving such advice. If the Nurse fails to provide the required documentation/paperwork, the supervisor can evaluate the Nurse without benefit of the documentation/paperwork. If the Nurse's Supervisor is not a registered Nurse, the assessment of the Nurse's clinical skills shall be provided by a Registered Nurse not covered by this Agreement. Performance problems will be identified with the Nurse prior to documentation in the performance appraisal.
- 27.2 Areas needing improvement during the period covered by the evaluation will be noted on the evaluation form. For example, excessive absences during the period will be noted on the evaluation form. However, specific reference to a disciplinary action (e.g., written warning for excessive absences) will not be documented on the form. The Nurse's supervisor may seek to work out a written plan with the Nurse, if needed, intended to eliminate the performance problems and specifying the improvement that is expected and the timeframe in which it is expected to occur. If mutual agreement is not reached, the supervisor will specify in writing the elements of the performance improvement plan. If the Nurse's performance in any areas that needed improvement has not been satisfactory, such performance shall be addressed in the next performance evaluation. Overtime issues which are not performance related will not be addressed in the evaluation, except in summary.
- 27.3 After the fourteen (14) day period for the Nurse to provide the required documentation/paperwork has expired, and prior to the review date, the supervisor will contact the Nurse to schedule an appointment to discuss the Nurse's performance.
- 27.4 Peer evaluations may be requested by the Nurse or Manager. On units where peer evaluations are requested, the supervisor will assume the responsibility for gathering such evaluations unless the individual Nurse and supervisor mutually agree otherwise. It is understood that any comments solicited from non-Nurse coworkers will be limited to the coworker's assessment of the Nurse's interpersonal skills and/or leadership ability in the working environment. New peer review committees for the purposes of annual evaluation of Nurses will not be created except by mutual agreement. The Nurse will be allowed to review any evaluations from peers in the form provided by the peer to the supervisor.
- 27.5 The supervisor will make all reasonable efforts to provide the Nurse with the written performance evaluations within thirty (30) days of the annual review date. Any necessary pay actions for Clinical Nurse IIIs and IVs will be implemented according to the terms of the Professional Nurse Development Program.
- 27.6 The supervisor will upon completion of the evaluation process prepare a brief summary of the evaluations and peer evaluations where appropriate to become part of the Nurse's permanent personnel file. A copy will be provided to the Nurse upon request.

SECTION 28 WRITTEN WARNINGS AND DISCIPLINE

28.1 <u>Written Warning (excludes other Disciplinary Notices)</u>.

28.1.1 A written warning is provided primarily where previous verbal communications have been ineffective and the supervisor wishes to impress upon the Nurse the seriousness of a problem and/or more serious corrective action or discharge is not deemed warranted. If the Nurse's supervisor is not a Registered Nurse, the

assessment of the Nurse's clinical skills shall be provided by a non-CRONA Nurse. The supervisor will advise the Nurse that s/he may request a CRONA representative at the meeting in which the written warning will be delivered. A CRONA representative must be available for a meeting within three (3) business days of the Nurse's request to have a CRONA representative at the meeting, or within such additional time as is agreed to by the Employer. A recurrence of similar related misconduct may result in further corrective action or discharge. Regular and Relief Nurses who have completed the trial periods specified in Section 6 and 15 of the Agreement may grieve the Written Warning pursuant to the provisions of Section 29 and the Nurse may provide written objections to the warning within fifteen (15) days of receipt. The objections will be placed in the Nurse's permanent personnel file.

- 28.1.2 Written warnings and attachments will be removed from the Nurse's permanent personnel file and will not be considered in any corrective action after twelve (12) months have lapsed since the date the warning was issued, unless the Nurse engages in similar or related misconduct within that period. Similar or related misconduct during that period may result in the warnings being extended an additional twelve (12) months and may result in whatever corrective action may be deemed appropriate. After a Nurse's termination of employment, written warnings shall not be disclosed to other prospective employers.
- 28.2 <u>Temporary Relief of Duty Pending Investigation</u>. If the Hospital determines that the circumstances are such that they warrant removing a Nurse from the workplace pending investigation for a possible disciplinary action or termination, the Nurse will be advised that (s)he is being temporarily relieved of duty, pending investigation. At the same time, the Hospital will also inform the Nurse of the date(s) and nature of the incident(s), as known at the time, unless the Hospital reasonably determines that because of the particular circumstances to do so would jeopardize the integrity of the investigation. The Nurse temporarily relieved of duty pending investigation shall remain in paid status for his or her previously scheduled shifts, except as provided in Section 28.2.
 - a. The Hospital will conduct its investigation diligently, consistent with the circumstances.
 - b. Both the Nurse and CRONA shall cooperate with the Hospital during the scheduling of the investigatory interviews. The Nurse and a CRONA representative shall be available to attend the investigatory interviews within three (3) calendar days after the Hospital requests the investigatory interview. If the investigatory interview does not occur within this three (3) day period because either the Nurse or the CRONA representative is not available during this period, then the Nurse shall be on unpaid status until the investigatory interview is actually held.
 - c. If upon investigation the Hospital imposes a termination or a suspension longer than the period during which the Nurse was relieved of duty without pay, the Hospital shall not attempt to recoup any amount paid to the Nurse while (s)he was relieved of duty, and all unpaid time when the Nurse was relieved of duty shall count against the suspension imposed.
 - d. For the effect of written warnings on Clinical Nurse IIIs and IVs, see Section XVI of the Professional Nurse Development Program.
28.3 Discipline.

- 28.3.1 During the Trial Period of Employment.
 - a. All newly hired Regular Nurses will serve a trial period of six (6) months, with no more than one (1) extension of three (3) months. The extension must be by mutual agreement between CRONA and the Employer unless the trial period has been automatically extended pursuant to Section 6.1. During the trial period of employment a Nurse may be disciplined for conduct or performance which is deemed unacceptable by the Employer. A trial period Regular Nurse is not eligible to use the Grievance Procedure in Section 29.
 - b. All newly hired Relief Nurses will serve a trial period. For "A" and "B" Relief Nurses the trial period shall commence on the first day of employment and extend until the date the Relief Nurse has actually worked one thousand (1,000) hours or twelve (12) months of employment provided that the trial period shall in no case be less than six (6) months of employment. The trial period for "C" and "D" Relief Nurses shall be six (6) months. During the trial period a Relief Nurse may be terminated at any time at the Employer's sole discretion and the Employer's discretion shall not be subject to review under any provisions of this Agreement.
 - c. Before a Regular or Relief Nurse in her/his trial period can be terminated primarily for unsatisfactory performance not involving serious performance deficiencies warranting immediate discharge, s/he must be interviewed by supervision. During the interview the Nurse must be thoroughly informed of those areas of job performance considered unsatisfactory. Supervision should then seek to work out a written plan with the Nurse, aimed at eliminating performance problems, or, if this is not practicable, the Nurse should be told what sort of improvement will be expected of the Nurse before the end of the trial period.
 - d. Except in serious cases warranting immediate discharge, a Regular Nurse discharged during the trial period of employment who has completed the first ninety (90) calendar days of the trial period will receive a one (1) week notice of separation. The Regular Nurse may be required to work as usual during the notice period, or may be given pay in lieu of notice, or may be required to work part of the notice period and be paid in lieu of working the remainder of the period.
 - e. Except in serious cases warranting immediate discharge or when the discharge involves a failure to maintain commitment, a Relief Nurse discharged during the trial period of employment who has completed the first five hundred (500) hours of employment will receive a one (1) week notice of separation. The Relief Nurse will not be required to work during the notice period and is not eligible for pay in lieu of notice.
 - f. A Relief Nurse discharged during the trial period for failure to maintain commitment pursuant to Section 15.12.1 will be advised of the termination by letter to her/his last known address.
 - g. CRONA will be notified of any Regular or Relief Nurse being considered for discharge in the trial period.

28.3.2 After the Trial Period of Employment.

- a. <u>Discipline for Cause Regular Nurses</u>. Regular Nurses who have successfully completed the trial period of employment will not be disciplined except for just cause. Any discipline may be appealed by the Nurse under the provisions of Section 29. Except in serious cases warranting immediate discharge, Regular Nurses discharged after the trial period of employment will receive two (2) weeks notice of discharge. Regular Nurses may be required to work as usual during the two (2) weeks or may be given pay in lieu of notice, or may be required to work part of the notice period and be paid in lieu of working the remainder of the period.
- b. <u>Discipline for Cause Relief Nurses</u>. Relief Nurses who have successfully completed the trial period of employment will not be disciplined except for just cause. Any discipline may be appealed by the Relief Nurse under the provisions of Section 29. Except in serious cases warranting immediate discharge or when the discharge involves a failure to maintain commitment, Relief Nurses discharged after the trial period of employment will receive two (2) weeks notice of discharge. The Relief Nurse will not be required to work during the notice period and is not eligible for pay in lieu of notice.
- c. A Relief Nurse discharged for failure to maintain commitment pursuant to Section 15.12.1. will be advised of the termination by letter to her/his last known address.

28.3.3 Termination Primarily for Unsatisfactory Performance - Regular and Relief.

- Probation for Non-Trial Period Nurses. Before a Nurse who has successfully a. passed the trial period can be terminated primarily for unsatisfactory performance (not including failure to maintain commitment) under the just cause provision herein, s/he must be interviewed by supervision. During the interview the Nurse must be thoroughly informed of those areas of job performance where s/he is considered unsatisfactory. Supervision shall then seek to work out a written plan with the Nurse, aimed at eliminating the performance problems and specifying what sort of improvement will be expected of her/him over the term of a defined probationary period not to exceed sixty (60) days in length, if supervision determines that a probationary period would be useful. The CRONA Nurse Advocate or a CRONA officer or officers designated by CRONA as the representative for the purposes of this paragraph will be notified when supervision determines that a Nurse may be placed on probation and at the Nurse's request may be present at the meeting to discuss the probation. CRONA will inform the Employer in writing of the CRONA officers so designated within thirty (30) working days after execution of this agreement and thereafter within ten (10) working days of any change in officer designation. The Nurse will be notified in writing concerning her/his status within ten (10) working days of the end date of the probationary period. A copy of the notification will be provided to CRONA.
- b. <u>Termination for Cause Regular and Relief</u>. If the Nurse does not demonstrate her/his ability to satisfactorily perform her/his job after a defined probationary period or if supervision determines that a probationary period is not feasible, the Nurse may be terminated under the just cause provision set

forth above. A CRONA representative will be notified of the decision to terminate.

- c. <u>Opportunity to Transfer After Trial Period Regular Nurses</u>. A Regular Nurse who is subject to probation or termination under this Section may request a transfer to another Nursing Unit provided that a position exists and that supervision of both affected units agree to the transfer. If a transfer is approved the Regular Nurse will be evaluated pursuant to Section 6.2 of this Agreement. The decision of unit supervision as to the feasibility of transfer shall not be reviewable under the terms of this Agreement.
- d. <u>Opportunity to Transfer After Trial Period Relief Nurses</u>. A Relief Nurse who is subject to probation or termination under this Section may request a transfer to another Nursing Unit provided that a position exists and that supervision of both affected units agree to the transfer. If a transfer is approved the Relief Nurse will be evaluated pursuant to Section 15.6 of this Agreement. The decision of unit supervision as to the feasibility of transfer shall not be reviewable under the terms of this Agreement.

SECTION 29 GRIEVANCE AND ARBITRATION

- 29.1 <u>Purpose</u>: The purpose of the procedures set forth herein is to provide the parties with an orderly means of resolving differences which may arise between them. The parties intend that the procedures set forth shall be in lieu of any other formal procedure established by the Employer for resolution of employee grievances and shall be the exclusive means for resolution of CRONA's grievances against the Employer.
- 29.2 <u>Grievances</u>.
 - 29.2.1 Only Nurses who have successfully completed the trial period of employment are eligible to use this grievance procedure, except that trial period Nurses may use this procedure for disputes over whether their paychecks are in accordance with the pay provisions of this Agreement, or whether they have been subjected to any discrimination of the type described in Section 2.
 - 29.2.2 A formal grievance is a written claim by a Nurse or CRONA concerning a Nurse's wages, hours, or working conditions and involving the interpretation or application of this Agreement. A grievance may be filed by any individual Nurse covered by this Agreement or by any officer designated by CRONA. A grievance filed by CRONA when not on behalf of an individual Nurse shall be started at Step Two of this procedure.
- 29.3 Grievance Procedure.
 - 29.3.1 <u>Step One Informal Discussions</u>. A grievance may be started when a Nurse (called "grievant") tells her/his supervisor the facts of the grievance and asks for resolution within five (5) working days. A CRONA representative may be present at the option of the individual Nurse. If the grievant does not accept resolution of any part of the grievance, s/he may proceed to Step Two of this procedure or the grievant may at her/his option skip Step One and begin the grievance at Step Two.

- 29.3.2 <u>Step Two Submission of Formal Grievance</u>. The grievance shall be submitted in writing to the Vice President of Patient Care within the time frames set forth in paragraph b. Timeliness below. The Vice President of Patient Care will submit a copy of the grievance to CRONA and to the Director of Employee and Labor Relations within two (2) working days from receipt of such grievance.
 - a. <u>Content of Formal Grievance</u>. The formal grievance shall be signed and dated by the grievant or designated CRONA officer and shall contain a brief description of the action or inaction complained of, the Employer management representative, if known, who is believed to have taken the action or failed to act, the date the action occurred or should have occurred, the resolution desired, and the Section or Sections of the Agreement alleged to be involved.
 - b. <u>Timeliness</u>. In cases of discipline, separation or layoff, the formal grievance shall be submitted no later than twenty-one (21) calendar days, and in all other cases no later than thirty (30) calendar days, after the date the action occurred or should have occurred. Notwithstanding the preceding sentence, if in the case of a grievance, the Nurse aggrieved did not know of the action or failure to act when it occurred, then the grievance shall be submitted no later than thirty (30) calendar days after the day when the Nurse could reasonably have been expected to have known. If, in the case of a grievance which is not a claim on behalf of any Nurse, neither CRONA nor any of its agents, knew of the action or failure to act when it occurred, then the grievance shall be submitted within thirty (30) calendar days after the day after the day CRONA or any of its agents could reasonably have been expected to have known. Disputes as to timeliness shall be resolved by the arbitrator should the grievance proceed to that stage.
- 29.3.3 <u>Step Three Resolution or Referral for Review</u>. The Vice President of Patient Care Services shall respond to the grievance or refer it to a reviewing Employer representative within seven (7) working days of receipt. A copy of the Vice President of Patient Care Services' resolution or notice of referral will be provided to the Nurse grievant, and CRONA, and the Director of Employee Labor Relations, or designee.
 - a. <u>Review Meeting</u>. The reviewing representative shall call a meeting with the grievant and/or CRONA officers, and any other person or persons as can, in the view of the reviewing representative, because of their knowledge of the facts, contribute to a discussion of the grievance. The review meeting will be conducted no more than fourteen (14) calendar days after the date of referral by the Vice President of Patient Care Services, provided CRONA representatives are available, and provided further that the parties may agree to extend the time period to a mutually convenient date.
 - b. <u>Resolution</u>. The reviewing Employer representative shall present a written determination on the grievance with copies to the grievant and CRONA within fifteen (15) working days after the review meeting unless mutually extended. The determination shall state the specific reasons for the decision either to grant or deny the grievance.
 - c. If CRONA does not accept the Step Three determination then within fifteen (15) working days of receipt of the determination, CRONA may refer the issue to mediation or arbitration, as described below, by written notice to the

Director of Employee and Labor Relations with a copy to the Vice President of Patient Care Services. CRONA or its representatives shall contact the Employer's representative to initiate selection of a mediator or arbitrator within fourteen (14) calendar days of its referral of the grievance to mediation or arbitration.

29.3.4 <u>Step Four (Optional) – Mediation</u>.

- a. As specified above, CRONA may request that the grievance be submitted to mediation, which shall be subject to agreement by the Employer. Within seven (7) days of CRONA's request to submit to mediation, the Employer shall inform the Union in writing whether the Employer is agreeing to mediation. If the Employer does not agree to mediation or fails to respond on a timely basis, the grievance shall instead be submitted to arbitration, as provided in Section 29.2.5, below.
- b. If CRONA and the Employer have agreed to mediation, as provided above, the parties shall request the appointment of a mediator to hear the grievance from the Federal Mediation and Conciliation Service. The mediation session shall occur within thirty (30) days of the FMCS's designation of the mediator to mediate the grievance.
- c. If the grievance is not satisfactorily resolved at the mediation session, CRONA may appeal the grievance to arbitration within three (3) calendar days of the date of the mediation session.

29.3.5 <u>Step Five – Arbitration</u>.

a. When CRONA has requested arbitration in accordance with this Section, CRONA and a representative designated by the Employer shall attempt to reach Agreement on an arbitrator by informal discussion. If agreement has not been reached within five (5) working days of the request for arbitration, the arbitrator shall be selected from the following seven (7) persons by the alternative striking of names, with the Employer striking first, until one remains, who shall be the arbitrator:

Alexander Cohn, Matthew Goldberg, Ronald Hoh, John Kagel, Anita Christine Knowlton, Geraldine Randall, and Ilona Turner.

The first strike for arbitrators will be rotated between CRONA and the Employer.

b. <u>Hearing</u>. The arbitration shall begin as soon as possible giving due consideration to the schedules of the representatives and witnesses of the parties provided that failure of CRONA to request the setting of a hearing date within sixty (60) calendar days of the referral to arbitration shall result in a waiver of the claim. The hearing shall be closed unless the arbitrator rules otherwise. Prior to the hearing the parties shall attempt to reach agreement on a joint submission of the case to the arbitrator. If the parties fail to agree on a joint submission, each shall present a separate submission, and the arbitrator shall determine the issue or issues to be heard provided that the issue is arbitrable in accordance with this Section. The joint or separate

submissions shall state the issue or issues and the specific Section or Sections of this Agreement which the arbitrator is to interpret or apply.

- c. <u>Resolution</u>.
 - (1) After such hearing the arbitrator shall render as soon as possible a decision which shall be final and binding on all parties.
 - (2) The arbitrator shall have no power to add to, subtract from, alter, modify or amend any of the terms or provisions of this Agreement.
 - (3) The arbitrator's authority to award monetary damages shall be limited to compensatory damages.
- d. <u>Expense</u>. The cost of compensation and expenses of the arbitrator, including the cost of a transcript unless a transcript is waived by mutual agreement of the parties and the arbitrator, shall be divided equally between the parties. However, each party shall bear its own expenses of representation and witnesses.
- e. <u>Expedited Arbitration</u>. The parties may agree in writing that an individual grievance or grievances be submitted to expedited arbitration as set forth in this subsection e.
 - (1) The arbitrator shall conduct a hearing at the earliest date;
 - (2) Unless ordered by the Arbitrator, there shall be no transcript of the hearing and post-hearing briefs shall be waived;
 - (3) The Arbitrator shall use best efforts to render a decision within ten (10) working days following the Arbitrator's closing the hearing record.

Except to the extent modified in this subsection e, the remaining provisions of Section 29 shall be applicable to expedited arbitration.

29.3.6 <u>Right to Representation</u>. The Nurse may upon request have a CRONA representative present.

- a. <u>Representation at Grievance Meetings</u>. A grievant may be assisted or represented by up to two (2) representatives of CRONA at any Step Three meeting provided that no meeting will be delayed in order to obtain the presence of a second representative. In the event the Hospital intends to have more than two (2) representatives (not including a management witness, the management person alleged to have violated the Agreement, or the reviewing representative), it will notify CRONA in advance and CRONA may have an equal number of representatives, provided that the meeting will not be delayed to obtain additional CRONA representatives.
- b. <u>Representation at Investigatory Interview</u>. When in the Employer's judgment an investigatory meeting is called for prior to any decision to discipline a Nurse, the supervisor shall inform the Nurse of the purpose and subject of the meeting prior to the meeting. The Nurse may upon request have up to two (2) CRONA representatives present, provided that the interview will not be

delayed to obtain the presence of a second CRONA representative. If the Nurse requests the presence of a CRONA representative, the meeting shall occur within three (3) calendar days of the Employer's request for the investigatory meeting. If the Nurse requests representation and if the Hospital intends to have more than two (2) representatives (not including a management witness), the Hospital will notify CRONA in advance, and CRONA may have an equal number of representatives provided that the meeting will not be delayed to obtain additional CRONA representatives. The Hospital will conduct its investigation diligently, consistent with the circumstances. The Nurse, and at the Nurse's request, CRONA will be informed of the status of the investigation within thirty (30) days of the date of an investigatory interview.

- c. <u>Representation at Disciplinary Meeting</u>. When the Employer has determined that a Nurse is to be disciplined, the supervisor will advise the Nurse that s/he may request a CRONA representative at the meeting in which the written warning will be delivered. If a CRONA representative is requested to attend the meeting, the meeting must be held within three (3) calendar days of the date the supervisor notifies the Nurse of the disciplinary meeting, or within such additional time as is agreed to by the Employer. If the CRONA representative is not available to attend the meeting during this time period, then the disciplinary meeting may proceed without the presence of the CRONA representative. Prior to the meeting supervision will inform the Nurse of the purpose for the meeting.
- d. <u>Definition of Discipline</u>. For the purposes of sub-sections b. and c. above only, an oral warning or reprimand shall not be considered discipline. Nothing in Section 29 shall require the Employer to schedule an investigatory meeting prior to making a decision to discipline a Nurse.

29.3.7 Adherence to Time Limits.

- a. <u>Processing of Grievances</u>. The Employer and CRONA agree that grievances should be raised, and settlement attempted, promptly. Failure of CRONA or a Nurse to proceed within any time limit set forth in this Section shall constitute a waiver of the claim. Failure of the Employer to act within any time limit set forth herein shall entitle the grievant or CRONA officer to proceed to the next step. If the Employer has not responded within the required time limit, the Employer shall be deemed to have rejected the grievance on the last day of the period for response and the matter may be appealed to the next level. However, any of the time limits set forth in this Section may be extended by mutual written agreement of the Employer and CRONA.
- b. <u>Limit of Liability</u>. The Employer shall not be liable for, nor shall any review or arbitration hearing concern, a claim for back wages or other financial reimbursement for any period prior to one hundred twenty (120) calendar days before the filing of the formal grievance which is the subject of the claim, review or arbitration hearing.
- 29.3.8 <u>Grievance Pending</u>. Grievances filed prior to the effective date of the current Agreement shall be heard under the terms and procedures set forth in the prior Agreement.

29.3.9 <u>Informal Settlement Discussions</u>. CRONA or the Employer may attempt to resolve a grievance at any time or at any level through informal settlement discussions. Such discussions shall in no way interfere with the grievance procedure nor require the participation of the grieving Nurse. All such discussions shall be treated as confidential and shall not be used as evidence for or against any position in any subsequent arbitration. No adjustment of a grievance through such discussions shall conflict with or supersede the terms of this Agreement or serve as precedent for the settlement of any other grievances filed under this Agreement.

SECTION 30 NO STRIKES OR LOCKOUT

There shall be no strikes, slowdowns, sympathy strikes, work stoppages, picketing, or concerted interference with the business of the Employer, on the part of CRONA, CRONA officers or its agents. There shall be no lockout on the part of the Employer during the term of this Agreement.

SECTION 31 NOTICE OF PERSONNEL GUIDELINES AND PROCEDURE CHANGES

CRONA recognizes that, except as provided in this Agreement, the Employer establishes and maintains personnel guidelines and procedures of general application to all the employees including Nurses, and, subject to this Agreement, the Employer retains the sole discretion to add to, delete from, or otherwise change the provisions of these guidelines and procedures. The Employer will notify CRONA in writing at least forty-five (45) calendar days in advance of implementation of any changes in those personnel guidelines and procedures which apply to Nurses covered by this Agreement and upon request meet with CRONA to discuss CRONA's recommendations and suggestions concerning the proposed changes. Upon request by CRONA, the Employer will meet with CRONA to discuss CRONA's recommendations and suggestions regarding the current policies which apply to Nurses covered by this Agreement to discuss CRONA's recommendations and suggestions regarding the current policies with the goal of achieving consistency. The Employer's discretion with regard to personnel guidelines and procedures be subject to review under this Agreement.

SECTION 32 JOINT CONFERENCES

- 32.1 The Employer and CRONA recognize their mutual interest and concerns regarding numerous matters and their effects on Nurses, e.g., health and safety, work environment, human resources matters, parking and security and personnel policies and procedures of specific application to Nurses and of general application to the Employer's employees, including Nurses. The Employer desires to have CRONA share with the Employer their suggestions and recommendations regarding such matters. Therefore the Employer and CRONA agree that at the request of either the Employer or CRONA, joint conferences with CRONA, Stanford Health Care and Lucile Salter Packard Children's Hospital at Stanford shall be held monthly between appropriate Employer representatives and CRONA officers for the purpose of discussing such matters. A representative from Nursing Administration from each Hospital shall attend the meetings. It is understood that no matters discussed or action taken as a result of such conference shall, in any way, change or alter any of the provisions of this Agreement or the rights or obligations of either party under the terms of this Agreement. The parties also may mutually agree to schedule special conferences.
- 32.2 The Employer respects the legitimate concerns of Nurses who may prefer not to participate in procedures in keeping with an individual Nurse's moral, ethical, and/or religious beliefs except in an emergency when a patient's life is endangered or when the Nurse's actions are

not consistent with good nursing practice. The Employer will raise and discuss as appropriate individual cases involving these issues in joint conferences. CRONA recognizes that the Employer retains the sole discretion to make a final determination in such matters and agrees that to the extent disputes are processed in the grievance procedure the Employer's determination will not be overturned except upon a showing that such determination is arbitrary or capricious.

32.3 The parties recognize that issues may arise during the term of this Agreement regarding a possible conflict between the terms of this Agreement and a reasonable accommodation for qualified individuals with a disability. Where such issues arise, the Employer will consult with CRONA to determine on a case-by- case basis whether a mutually acceptable solution can be found.

SECTION 33 NURSE STAFFING

33.1 Patient Classification System and Committee.

- 33.1.1 As required by law, the Hospital shall maintain a documented and legally compliant patient classification system for determining the nursing care needs of individual patients that reflects the assessment, made by a registered nurse, of patient requirements and provides for shift-by-shift staffing based on those requirements. As provided by law, a patient classification system means a method for establishing staffing requirements by unit or area of practice, patient, and shift that includes (a) a method to predict nursing care requirements of individual patients, (b) an established method by which the amount of nursing care needed for each category of patient is validated for each unit and for each shift, (c) an established method to discern trends and patterns of nursing care delivery by each unit, each shift, and each level of licensed and unlicensed staff. (d) a mechanism by which the accuracy of the nursing care validation method can be tested, (e) a method to determine staff resource allocations based on nursing care requirements for each shift and each unit, and (f) a method by which the hospital validates the reliability of the patient classification system for each unit and for each shift.
- 33.1.2 The Hospital shall maintain a committee, referred to here as the workload acuity validation work group ("WAVWG"), which, consistent with applicable law, shall review at least annually the reliability of the patient classification system for validating staffing requirements to determine whether or not the system accurately measures patient care needs. Consistent with applicable law, the Hospital shall maintain and document a process by which all interested Nurses may provide input about the patient classification system, the system's required revisions, and the overall staffing plan.
- 33.1.3 The WAVWG shall consist of at least fifty percent (50%) Registered Nurses who provide direct patient care and are represented by CRONA. CRONA shall have the right to recommend to the Vice President of Patient Care Services the members of the WAVWG who provide direct patient care. The Vice President of Patient Care Services shall appoint the members of the WAVWG, not unreasonably rejecting recommendations by CRONA, with a goal to have balanced representation from each area and shift.

33.2 Staffing Ratios.

The Hospital shall meet or exceed state-required patient staffing ratios in every applicable unit or area of practice at all times, including during meal periods and rest breaks, as provided by applicable law. As provided by law, additional staff in excess of these prescribed ratios shall be assigned in accordance with the patient classification system for determining nursing care requirements, considering factors that include the severity of the illness, the need for specialized equipment and technology, the complexity of clinical judgment needed to design, implement and evaluate the patient care plan, the ability for self-care, and the licensure of the personnel required for care. If, considering the factors identified in the preceding sentence, staffing is assigned in excess of the state-prescribed ratios, such assigned staffing shall be maintained during meal periods and rest breaks.

33.3 Staffing Plans.

- 33.3.1 Each unit shall maintain a staffing plan that is compliant with applicable law and based on patient care needs as determined by the patient classification system provided in Section 33.1. The unit or area's staffing plan shall be available to the Nurses on that unit. The staffing plan shall be developed and implemented for each patient care unit or area of practice and shall specify patient care requirements and the staffing levels for registered nurses. The plan shall include the following: (1) staffing requirements as determined by the patient classification system for each unit or area of practice, documented on a day-to-day, shift-by-shift basis; (2) the actual staff and staff mix provided, documented on a day-to-day, shift-by-shift basis; (3) the variance between required and actual staffing patterns, documented on a day-to-day, shift-by-shift basis.
- 33.3.2 Staffing is determined by a combination of professional judgment, acuity and staffing plans/matrices. In meeting the staffing needs of the unit or area of practice, the Resource Nurse is permitted to use judgment in adjusting staffing levels that may not be fully addressed by the acuity and staffing plan/matrix. The patient classification system shall provide guidance to the Resource Nurses regarding when, based on projected patient care needs, including patient census and acuity, a shift needs more or fewer Nurses than are scheduled to work; provided, however, that nothing in such guidance shall be inconsistent with applicable law.
- 33.4 <u>Dispute Resolution</u>. Disputes regarding this Section shall not be subject to the grievance and arbitration procedure under Section 29, but instead the procedures in Section 34.4 shall apply to such disputes. If the Vice President of Patient Care and the President of CRONA cannot resolve a pattern of disputes regarding the provisions of this Section pursuant to the procedures in Section 34.4, the Hospital and CRONA will engage a mediator through the Federal Mediation and Conciliation Service to facilitate consensus building. The parties will request the assignment of a mediator with experience in the health care field. If consensus is not reached and with the mutual agreement of the parties, the mediator is authorized to issue to the parties the mediator's proposed resolution regarding the dispute.

SECTION 34 NURSE PRACTICE COMMITTEE

34.1 <u>Composition</u>. The Nurse Practice Committee is a joint Employer/CRONA committee consisting of four (4) representatives selected by the Employer and four (4) Nurses selected by CRONA, and with advance mutual agreement, additional ad hoc Nurse or Patient Care Manager participants as the Committee determines to be of assistance on particular issues

or problems. In the event that either the Employer or CRONA wishes to have additional representatives appointed to the Nurse Practice Committee from departments outside the Department of Nursing who have CRONA R.N. staff, the appointment would be subject to the Committee's approval. This committee will have a co-chair from the Employer and from CRONA chosen by the respective parties from within the eight (8) appointed members.

- 34.2 <u>Purpose</u>. To promote, develop, and continually enhance professional nursing practice, effective utilization of nursing resources, and an organizational climate conducive to professional practice.
- 34.3 Responsibilities.
 - 34.3.1 To discuss items mutually agreed to be of concern as issues affecting Nursing Practice by both the Employer and CRONA.
 - 34.3.2 To discuss and develop guidelines for the involvement of Nurses and physicians in a collaborative partnership in the provision of excellent patient care.
 - 34.3.3 To consult, explore problems, and make recommendations to the Vice President of Patient Care regarding issues arising from staffing and patient care assignments or from Nurse objections to assignments.
 - a. The Nurse Practice Committee may review each unit's current acuity and matrix tool, and monitor the appropriateness of staffing levels by reviewing the Nursing Department's benchmarks and quality indicators. Upon request the Committee will review changes in the composition of personnel of the affected units which result in a material change in the duties or work load of Nurses working on the affected unit. Recommendations on staffing levels will be made to the Vice President of Patient Care. If the Nurse Practice Committee does not reach consensus on its recommendations, the issues will be referred for final discussion between the Vice President of Patient Care and the President of CRONA.
 - b. Assignment Despite Objection Form (ADO) as prepared by CRONA, will be an open agenda item for review by the Nurse Practice Committee with recommendations, if any, made in accordance with 33.3.3. ADO forms will be made available on the units. The manager or designee shall respond in writing to this Assignment Despite Objection Form (ADO within two (2) weeks after notification by CRONA. A copy shall be submitted to CRONA and the respective Patient Care Director.
 - 34.3.4 To consider and make recommendations regarding changes that are desired from time to time by either party regarding "float regions" and "closed staffing units" before decision by the Vice President of Patient Care, the Committee will discuss ideas for other operational arrangements, if feasible, to minimize the need for Nurses to float outside their designated float region.
 - 34.3.5 To develop guidelines for delegation of nursing care to non-R.N. staff.
- 34.4 Procedures.
 - 34.4.1 Upon request of either the Employer or CRONA representatives, the issue or problem and the unit/region shall be identified and the Committee shall determine

the appropriate procedure to discuss and review that issue or problem, including the appropriateness of inviting additional participants (Nurse(s) or Patient Care Manager(s)) from the unit or region to provide useful facts or particular expertise. The Committee shall defer placement of the issue on its agenda until the issue has been fully discussed and reviewed at the appropriate unit or region level.

- 34.4.2 The Nurse Practice Committee shall discharge its responsibilities in accordance with mutually agreed upon procedures. The Committee shall meet once a month, if necessary, to work on outstanding agenda items.
- 34.4.3 The Committee's discussions will focus on the facts of the particular issue and identification of potential ways, if any, to resolve the issues. The Committee shall attempt to prepare a joint recommendation to the Vice President of Patient Care if appropriate to the particular issue or problem. If a joint recommendation is not made, either the Employer or CRONA representatives may request that the Committee refer the issue or problem for final discussion and review between the Vice President of Patient Care and the President of CRONA.
- 34.4.4 It is understood that in order to promote full exploration of issues and efforts at problem solving, no matters discussed or action taken as a result of the work of the Committee shall change or alter any of the provisions of the Agreement or the obligations or rights under the Agreement of either CRONA or the Employer.

SECTION 35 SPECIAL CONFERENCES

CRONA and the Employer recognize that they have shared interests on many matters including the importance of maintaining an appropriate level of staffing for patients, resolving Nurse objections to patient care or unit assignments, and operational changes. At the written request of either CRONA or the Employer identifying the issue to be discussed, a special conference shall be held between appropriate representatives of the Employer and CRONA. When requested, a special conference will be held within fourteen (14) days of the request. At such special conferences appropriate information will be exchanged and consideration given to suggested ways to resolve issues and alternative approaches to the problems being discussed. It is understood that no matters discussed or action taken as a result of a special conference shall change or alter any of the provisions of the Agreement or the obligations or rights under the Agreement of either CRONA or the Employer.

The parties recognize that issues may arise during the term of this Agreement regarding a possible conflict between the terms of this Agreement and a reasonable accommodation for qualified individuals with a disability. Where such issues arise, the Employer will consult with CRONA to determine on a case-by-case basis whether a mutually acceptable solution can be found.

SECTION 36 HEALTH, WELFARE AND SAFETY

36.1 The Employer shall be responsible for and shall maintain reasonable provisions for the health, welfare and safety of Nurses. CRONA recognizes the duty of employees, and shall encourage employees, to cooperate with the Employer by complying with the Employer's health and safety rules and regulations and utilizing personal efforts for the prevention of accidents or illnesses to employees. CRONA may make recommendations and suggestions concerning ways to improve safety conditions (including at parking locations used by Nurses) pursuant to Section 32. For this purpose the Employer will notify CRONA of work-incurred

injuries within the bargaining unit and provide copies of accident reports of such injuries on request.

- 36.2 The Employer recognizes its obligation to seek to accommodate when operationally feasible and consistent with the provision of safe patient care the needs of Nurses seeking to return to work following a leave for a verified work related disability. The Employer will provide CRONA every sixty (60) days the names of Nurses who have been released to return to work with restrictions and who have not been returned to work within sixty (60) days of that release. Upon request from CRONA, the Employer will meet with a representative of CRONA to discuss the feasibility of finding employment for the Nurse.
- 36.3 A Nurse who has been assaulted or threatened by a patient or a member of the patient's family should immediately report the assault or threat to her or his manager and may request a patient reassignment. Such a request shall not be unreasonably denied, taking into account whether other nursing staff are available to care for the patient. A Nurse whose request for reassignment has been denied may request that Joint Conference review the denial at the following month's meeting.
- 36.4 The Employer shall maintain a Workplace Violence Prevention Program compliant with applicable law. The Employer, in consultation with the Workplace Violence Committee ("WVC"), shall maintain a behavioral response team that is tailored to the relevant work setting and is available to each unit and area of practice in which Nurses work during that unit's or area's hours of operation to respond immediately to situations that a Nurse identifies as having the potential to lead to workplace violence, including situations involving patients or family members who demonstrate disruptive behaviors. In addition, the recommended team for in-person responses at the main medical center campus shall include the treating physician or delegate, a social worker, a representative of security and a representative of nursing leadership. The WVC shall discuss the appropriate composition of the behavioral response team for units not located on the main medical center campus. The Employer shall announce no later than start of the fiscal year following ratification, but not earlier than 120 days after ratification the composition of the behavioral response team for units not located on the main medical center campus, which shall include a representative of site leadership or a delegate who is not a CRONA-represented Nurse.
- 36.5 The Employer's Workplace Violence Prevention Program shall be discussed at Nurse Practice Committee at least annually. Nurse Practice Committee shall review the program as it affects Nurses and shall provide input on appropriate modifications to the program, unitbased workplace violence prevention plans, the effectiveness of the behavioral response team, and training provided to Nurses. As part of these discussions at the Nurse Practice Committee, the Employer and CRONA shall review the recommendations of the WVC regarding the risk level assessment of each unit and area of practice, the type and modality of training regarding workplace violence prevention, de-escalation techniques to be provided to each unit and area of practice, and the Workplace Violence Incident Log for incidents related to Nurses.
- 36.6 On an annual basis, each unit shall provide to Nurses in-person training, which may include training during staff meetings, on the current protocols for responding in real time to incidents of workplace violence. The training shall include an opportunity for Nurses to ask questions regarding the protocols. Nurses will have access to the written behavioral response protocols.

- 36.7 If an issue related to the Workplace Violence Prevention Program is placed on the agenda of a Nurse Practice Committee meeting, the Employer shall appoint a member of its Employee Safety Council to attend the meeting as a representative of the Employer.
- 36.8 CRONA shall appoint two Nurses from different patient care regions who shall serve on the Employer's WVC that has responsibility for the Workplace Violence Prevention Program.
- 36.9 No Nurse shall be subject to reprisals or retaliation for reporting a concern or incident relating to workplace violence.
- 36.10 Consistent with its obligation to provide reasonable provisions for the health, welfare, and safety of Nurses, the Employer shall:
 - 36.10.1 Maintain a process for providing security escorts (such as individual escorts or shuttle services) for Nurses to and from parking locations at the main medical center campus;
 - 36.10.2 Review annually the safety conditions of parking locations controlled by the Employer, including the presence of video cameras and security guards; and
 - 36.10.3 Discuss with the WVC safety concerns related to parking locations at which Nurses regularly park but which are not controlled by the Employer ("Non-Controlled Parking Locations"). In addition, the Employer agrees to raise and propose a resolution of concern(s) identified by the WVC related to Non-Controlled Parking Locations with the entity that controls those locations.

SECTION 37 SENIORITY ACCRUAL AND APPLICATION

37.1 <u>Seniority – Definition</u>.

Subject to the conditions contained in this Section 37 "seniority" is defined as follows:

- 37.1.1 Seniority applies to Regular full-time and part-time Nurses.
- 37.1.2 Seniority means the period of continuous, unbroken service within Stanford Health Care.
- 37.1.3 If seniority is determinative among two or more Regular Nurses with equal seniority, the order of seniority will be determined by the date of the Nurse's signature on her/his application for employment, subject to seniority adjustments. If the dates of the signatures are the same, or one of the Nurses does not have a copy of the application on file, seniority among such Nurses will be determined by the employee number, with the lower number being considered the most senior.
- 37.2 Regular Nurses who were employed by UCSF Stanford Health Care on March 31, 2000 and were employed by Stanford Health Care (previously known as Stanford Hospital and Clinics) on April 1, 2000 will be credited with their accrued seniority as of March 31, 2000.
- 37.3 Nurses who were employed by Lucile Salter Packard Children's Hospital at Stanford ("LPCH") immediately prior to being hired by the Hospital with no Break in Service will be credited with their accrued seniority from LPCH. Benefits shall be provided under the terms of this Agreement, except that such Nurses may carry over benefit accruals from LPCH as

provided by Hospital policy. For the purposes of this Section, a "Break in Service" shall be defined as having a termination date from LPCH that is not within the pay period immediately preceding the pay period in which the Nurse commences employment with the Hospital.

- 37.4 Continuity of service for seniority purposes shall be broken by any of the following:
 - a. Resignation.
 - b. Failure to return to employment at the end of an approved leave of absence.
 - c. Termination.
 - d. Layoff for twelve (12) months or more.
- 37.5 Adjusted Hire Date for Nurses is the date of hire in any position with Stanford Health Care, except hire as an "A" Relief or Limited Relief Nurse. The Adjusted Hire Date on a Nurse's personnel records will not be subject to review under this Agreement. For a Nurse moving from Relief to Regular status this section will be effective as of September 14, 2001.
- 37.6 Adjustment of seniority and the reinstatement of former employees will be in accordance with the Employer's "Reinstatement of Former Employees" policy. This section will be effective September 14, 2001 for a former employee who is reinstated to a position covered under this Agreement.
- 37.7 Seniority shall be considered with existing qualifications and ability, patient care needs, and staffing requirements for the purpose of selecting Regular Nurses for shift rotations and on call preference, participation in Department of Nursing or Employer committees, special projects, and recruitment activities as designated by Department of Nursing. In addition to the factors listed above in this section, unit equity shall also be a factor for choice of vacation and holiday time. Whenever two or more Regular Nurses are being considered for selection in the areas designated in this subsection, seniority shall be the tie-breaker in making the selection decision.
- 37.8 For temporary staff reductions, Regular Nurses as defined in Section 38, seniority shall be the determining factor for making any necessary adjustments in staffing, provided that, in the judgment of the Employer, the skill and ability of the Nurses affected is substantially equal.
- 37.9 For permanent staff reductions as defined in Section 38, seniority shall be the determining factor in the order of layoff in the affected area or areas provided that, in the judgment of the Employer, the skill and ability of the Nurses affected is substantially equal.
- 37.10 Nothing in this Section shall restrict the right of the Employer to evaluate a Nurse's qualifications and ability before considering the appropriateness of seniority as a factor.

SECTION 38 STAFF REDUCTIONS AND LAYOFF

- 38.1 <u>Staffing</u>. The final determination of appropriate staffing levels is one of the rights reserved to the Employer as defined in Section 40 of this Agreement. CRONA recognizes that the Employer's daily staffing needs may change from unit to unit, shift to shift and that the Employer has the right to schedule Regular and Relief Nurses according to those needs.
- 38.2 <u>Temporary Staff Reductions</u>.

- 38.2.1 If, in the judgment of the Employer, budgetary or operational considerations require a temporary reduction of Regular Nurses covered by this Agreement the Employer will notify CRONA and at CRONA's request meet and discuss ways in which to develop equitable staffing for the Regular full-time and part-time Nurses affected. These discussions shall include the feasibility of the following:
 - a. Voluntary temporary reduction in work commitment.
 - b. Credit for voluntary use of PTO.
 - c. Credit for mandatory "A" days.
 - d. Implementation of rotation of mandatory absent days based on inverse seniority.
 - e. Freeze on hiring.
 - f. Proportional reduction of hours among all Regular Nurses.
 - g. Length of credit given, if any, for voluntary PTO and/or mandatory absent days.
 - h. Cross-Training.
 - i. Reduction in direct patient care assignments to Registry Nurses and other non-Regular Nurses in the affected unit(s).

In periods of staff reduction and layoffs, Relief Nurses shall be scheduled according to need and pursuant to the terms of Section 15 of this Agreement.

- 38.2.2 When it is necessary to cancel Nurses on a unit, cancellations will occur in the following order if the remaining staff have the necessary skills and abilities to meet operational needs: voluntary, over commitment, Registry, Relief and Regular. A Nurse Supervisor/Manager will be taken out of the count before a Regular Nurse is canceled; however a Nurse may be required to float within the Nurse's region prior to a Nurse Supervisor/Manager being taken out of the count.
- 38.2.3 <u>"A" Time Credit</u>. Nurses eligible to accrue PTO shall accrue "A" Time Credit on the same productive hours as PTO is accrued, at the following accrual rate. The purpose of the "A" Time Credit is to compensate Regular Nurses at their base hourly wage rate, including shift differential, when they are absent from work as a result of "A" Time resulting from excess staffing on a unit. The Regular Nurse may choose to use PTO, "A" Time Credit or unpaid hours for Absent time. Accrued "A" Time may be used during any waiting period for State Disability insurance or Workers' Compensation and to supplement any such disability payments during a period of disability.

"A" Time Credit	Time Accrued Per Hour Worked
First Year of Employment	.0243
2 nd Through 3 rd year	.0253
4 th through 9 th year	.0265
10 th and subsequent years	.0273

Effective in August of each year a Regular Nurse's accumulated "A" Time will automatically be retained in the Nurse's "A" Time bank, unless the Nurse requests that the "A" Time be converted to PTO, or cashed out based on the same calculation as cashing out PTO.

- 38.3 <u>Permanent or Indefinite Layoff</u>. If, in the judgment of the Employer, budgetary or operational considerations require a permanent curtailment of operations or permanent layoff of Regular full-time or part-time Regular Nurses covered by this Agreement, the layoffs will be accomplished in accordance with this subsection.
 - 38.3.1 <u>Notice</u>. When the Employer determines that a permanent or indefinite layoff of Regular Nurses is imminent, it shall give CRONA such advance notice as is reasonable under the circumstances. When individual Nurses are selected for layoff, each shall be given at least thirty (30) calendar days notice or pay in lieu of such notice. A copy of each individual notice shall be sent to CRONA at the same time. The notice may be rescinded if circumstances develop that the Nurse is to be retained, reassigned, or offered another position. If a Nurse resigns after being given notice of permanent layoff, the balance of the notice period will not be converted to pay. However, the Nurse will, if otherwise eligible, continue to be entitled to severance pay.
 - 38.3.2 <u>Selection of Those to be Laid-Off</u>. The Employer shall designate the area or areas including the unit as appropriate where layoffs are to occur and the number of Regular Nurse positions. Within the designated areas, the Employer will select Nurses for layoff by seniority provided that, in the Employer's judgment, the skills and abilities of the affected Nurses are substantially equal, and in accordance with the Employer's assessment of operational needs. Where travelers are employed in a unit where layoffs are occurring, the contracts for the travelers in the unit will be terminated as soon as the terms of the contract permit, until all laid off Nurses from that unit have the opportunity to return to a regular position, but not to exceed a period of twelve (12) months from the effective date of layoff.
 - 38.3.3 <u>Placement Alternative to Layoff</u>. The Employer will, to the extent practicable, attempt to place Nurses selected for layoff in units with unfilled vacant positions, provided that in the Employer's judgment, the Nurse possesses the requisite skill and ability to perform the actually expected work without additional training (as distinguished from orientation).

The Employer will meet with CRONA in Joint Conference to discuss the placement of Nurses under this subsection. A Nurse who is offered and refuses placement in a position at substantially equivalent pay shall not be entitled to severance allowance as defined in subsection 4 below.

In the event there are no unfilled vacant positions for which the Nurse possesses the requisite skill and ability without additional training, the Nurse may elect to displace the Relief Nurse in the unit of layoff, if any, with the most recent adjusted date of hire, provided that adjusted date of hire is more recent than that of the Nurse being laid off.

38.3.4 <u>Severance Allowance</u>. Regular full-time and part-time Nurses with one (1) year or more continuous employment as a Regular full-time or part-time Nurse, who have not refused a position of substantially equivalent pay offered by Stanford Health Care, Lucile Salter Packard Children's Hospital at Stanford, or Stanford University, will be entitled to receive a severance allowance from the Employer according to the

calculation schedule set forth below. Repayment of severance is in accordance with 37.3.5.

If a Nurse subsequently accepts employment in another division or department with the Employer at any location other than those listed above, the Nurse is obligated to repay the severance in accordance with 37.3.5.

Calculation: Severance pay will be calculated by using the base monthly pay at time of layoff. The base monthly pay is obtained by multiplying the base hourly wage times one hundred seventy-three and thirty-three hundredths (173.33) hours times the Nurse's percent of full- time commitment. It does not include shift differential, overtime or other premium pay.

Schedule: Severance is payable on the Nurse's last day of work according to the following table:

Years of Continuous Employment	Severance Pay Eligibility In Months Of Base Pay
1 year but less than 2	.5
2 years but less than 4	1
4 years but less than 7	2
7 years but less than 10	3
10 years but less than 12	4
12 years but less than 14	5
14 years but less than 16	6
16 years but less than 18	7
18 years but less than 20	8
20 years but less than 22	9
22 years but less than 24	10
24 years but less than 26	11
26 years or more	12

- 38.3.5 <u>Repayment of Severance</u>. Prior to receiving severance pay, a Regular Nurse must sign an agreement (Severance Repayment Agreement) to repay severance to the Employer if re-employed (temporary or regular) by the Employer within one year of the effective date of layoff. In that event, the Nurse may retain that portion of the severance pay equal to the base pay s/he would have earned if not laid off. The balance is to be repaid either in full at the time of re-employment or by payroll deduction. A Nurse may request other arrangements and, if approved, the schedule of repayment will be established by written agreement between the Nurse and the Director of Employee and Labor Relations, or designee.
- 38.3.6 <u>Continuation of Benefits</u>. Nurses on layoff status have the option of purchasing health, dental and life insurance at the group rate available to the Employer pursuant to COBRA. These premiums must be paid monthly, in advance by the Nurse.
- 38.3.7 <u>Re-Employment</u>. Nurses who have been laid off have preference for reemployment (temporary or regular) as specified below, for twelve (12) months from date of layoff. Repayment of a portion of the severance allowance may be required in accordance with 37.3.5. The Employer and CRONA will discuss Joint Conference ways in which laid off Nurses can be made aware of vacant positions as they arise.

38.3.8 <u>Re-Employment Preference</u>. A Nurse who has been permanently or indefinitely laid off shall be given preference for employment in any posted Nurse vacancy for which s/he applies, if in the judgment of supervision s/he possesses sufficient skill and ability to perform without additional training the actually expected work, unless another applicant for the position is better qualified. The Employer shall have sole discretion as to the determination of appropriate Nurse level for Nurses exercising this preference.

SECTION 39 HOME HEALTH CARE

- 39.1 Nurses employed in the Home Health Care service shall have the following special arrangements.
 - 39.1.1 Mileage reimbursement for the use of personal automobiles shall be provided in accordance with applicable policy as defined in Section 31 of this Agreement. In addition, Nurses will receive a quarterly payment of \$50.00 toward excess auto expenses associated with Home Care visits.
 - 39.1.2 All special arrangements outlined in this Section shall be subject to the grievance procedures of this Agreement.
- 39.2 The Employer does not require a Home Health Care Nurse to enter or remain in any situation that might present a threat to his/her personal safety. The Nurse will make all reasonable efforts to ensure appropriate care for the patient either by arranging for care in the Emergency Department or arranging for a police escort for the Nurse, as appropriate. The Nurse will notify the supervisor as soon as practicable after taking the action deemed necessary.

SECTION 40 RESERVE RIGHTS OF STANFORD HEALTH CARE

Except as modified or restricted by the express terms of this Agreement, the Employer reserves to itself all rights and functions of management including, but not limited to, the ability to determine the nature and scope of Employer functions, the ability to establish and alter methods of operation, including the determination of appropriate staffing levels, the determination of shift and duty assignments and the right to hire, promote, demote, suspend, discipline or discharge. It is agreed that this Agreement sets forth expressly all restrictions on the functions and rights of the Employer and no implied restrictions or obligations exist or may be relied upon in interpreting or applying this Agreement.

SECTION 41 PROFESSIONAL NURSE DEVELOPMENT PROGRAM

During the course of their 2010-2011 negotiations, the parties negotiated a Professional Nurse Development Program ("PNDP"), which is incorporated by reference herein. The intent and purpose of the PNDP is to improve nursing practice at the Hospitals and not to limit or cap the number of nurses who can achieve or retain Clinical Nurse III or IV status if they satisfy the criteria of the PNDP. The parties agree to apply the criteria and procedures set forth in the PNDP in good faith. The Hospital shall not use the criteria or procedures set forth in the PNDP to impose a cap on the number of Nurses who may achieve or retain Clinical Nurse III or IV status. A Nurse denied a promotion or renewal may exercise the appeal rights set forth in the PNDP, but may not grieve the

decision. Nothing in this Agreement shall prevent the parties from negotiating changes to the PNDP in future negotiations, including the introduction of limits on the number of Clinical Nurse III or IVs.

- 1. The Hospital agrees to provide training on how to write exemplars.
- 2. Applicants will be provided information regarding programs that exist to support RNs in improving their clinical expertise and professional development such as training and scholarship programs that are available.
- Panel members appointed by CRONA will be paid by the hospital pursuant to the "Staff and Committee Meetings" provisions of the Agreements (Section 7.9). A Nurse appointee may take the panel time within his or her commitment with the advance agreement of the supervisor.
- 4. The wage scales for the Clinical Nurse III and Clinical Nurse IV positions shall be as set forth in Appendix "A" – Wage Schedule. The change to a Clinical Nurse rate of pay will be effective beginning the pay period following written notice of acceptance for promotion or final appeal of a demotion.
- 5. Transition Period
- a. If a Clinical Nurse II applies for Clinical Nurse IV and is denied, but the Panel determines that the Nurse meets the Clinical Nurse III requirements, the Panel shall offer a Clinical Nurse III position to the Nurse.
- b. At the time of renewal in a non-panel year, if a Clinical Nurse IV is unable to maintain his/her Clinical Nurse IV status but the Nurse Manager determines that the Nurse meets the requirements for Clinical Nurse III, the Nurse Manager shall offer the Nurse a Clinical Nurse III position.
- c. All Clinical Nurses who terminate and seek reemployment must come back as Clinical Nurse II or a Relief Clinical Nurse II.
- d. BSN Waiver A nurse with five (5) or more years of service as a registered nurse, two (2) or more continuous years of service with Stanford Health Care and/or Lucile Salter Packard Children's Hospital at Stanford, and who was a Staff Nurse III or Staff Nurse IV as of October 1, 2015, but does not possess a BSN or MSN degree or a Doctorate of Nursing and does not desire to enroll in classes to obtain such a degree, will be considered to meet the educational requirement for becoming a Clinical Nurse III or Clinical Nurse IV if the nurse has or obtains a nationally recognized certification, as specified in subsection (iii) below. Notwithstanding the foregoing but only until March 31, 2022, a nurse who was a Staff Nurse III or Staff Nurse IV as of October 1, 2015 with at least twenty (20) years of service with Stanford Health Care and/or Lucile Salter Packard Children's Hospital at Stanford, shall be eligible for this waiver of a BSN or higher degree, regardless whether that nurse has or obtains a nationally recognized certification.
 - i. To be eligible for a waiver of a BSN or higher degree, the nurse must have obtained a Clinical Nurse III or Clinical Nurse IV position no later than March 31, 2022 and thereafter may use the BSN waiver to maintain such a position. The nurse may use the BSN waiver to apply multiple times for Clinical Nurse III and/or Clinical Nurse IV level.

- ii. A nurse who obtains a Clinical Nurse III or Clinical Nurse IV position pursuant to this BSN waiver provision and thereafter maintains a nationally recognized certification specified in subsection (iii) below (if required under this provision) will be deemed during future review periods to meet the requirement for purposes of retaining the Clinical Nurse III or Clinical Nurse IV status that the nurse obtained prior to March 31, 2022.
- iii. A certification that qualifies for the BSN waiver is either the most applicable certification in the nurse's area of specialty or a certification that supports the basic clinical practice in the nurse's area of work, e.g., pediatric certification for pediatric units and clinics at LPCH; O.B. certification for O.B. clinics and Labor & Delivery and Postpartum units at LPCH; medical/surgical certification for acute care pediatric units and clinics at LPCH, acute care units and clinics at SHC, and the cancer centers at LPCH and SHC; critical care certification for critical care units at SHC or LPCH; or ambulatory care certification for outpatient clinics at LPCH and SHC (other than the cancer centers).
- iv. The seniority and experience requirements provided for in this section shall be determined as of April 7, 2011.
- e. Nurses who apply for and are accepted for Clinical Nurse III or Clinical Nurse IV status under the Professional Nurse Development Program will be placed on the step on the appropriate Clinical Nurse III or Clinical Nurse IV scale commensurate with their service or prior experience credit (e.g., if they are on Step 5 of the Staff II wage scale and are accepted as a Clinical Nurse III or IV, they will be placed at the Step 5 of the applicable wage scale).
- f. The Hospital agrees that, during the life of the 2019 2022 Agreement, it will maintain records of all applicants for Clinical Nurse III or Clinical Nurse IV who go before the Clinical Nurse Selection Panel, whether they are accepted or rejected by the Panel, the reason given for rejecting each nurse, who among those rejected utilized the appeal process, and the result of the appeal process in each case. At the completion of each application period, the information concerning nurses who are rejected (identified by Employee ID Number) will be compiled and shared with CRONA in Joint Conference.
- g. Role of Panel during term of 2019 2022 Agreement: Notwithstanding Section XI (7) of the PNDP, no changes shall be made to the negotiated terms of the PNDP during the term of the 2019 – 2022 Agreement. As provided by Section XI(6) of the PNDP, the PNDP panel may continue to exercise its authority to interpret the terms of the PNDP. In addition, the PNDP Panel may make recommendations as provided by Section XI(7) on procedural and other issues that are not addressed by the terms of the PNDP.

SECTION 42 SEPARABILITY AND SAVINGS CLAUSE

If any provision of this Agreement should be held invalid by operation of law or by any tribunal of competent jurisdiction, or if compliance with or enforcement of any paragraph or subparagraph should be restrained by such tribunal pending a final determination as to its validity, the remainder of this Agreement, or the application of such paragraph or subparagraph to persons or circumstances

other than those as to which it has been held invalid or as to which compliance with or enforcement of has been restrained, shall not be affected thereby.

The undersigned, as authorized representatives of CRONA and Stanford Health Care, attest the ratification and approval of this Agreement, its Glossary and attached Side Letters.

SECTION 43 SUPERVISORY STATUS OF COVERED NURSES

- 43.1 Notwithstanding any other provision in this Agreement, the Employer agrees not to, and hereby expressly waives any right it may have to, withdraw recognition concerning, or to petition for unit clarification concerning, or take any other action direct or indirect for the purpose of challenging the inclusion in the bargaining unit of, any Nurses, or job classifications or titles, who or which are currently in the bargaining unit (including, but not limited to, Resource Nurses), on the ground that they are, or may be supervisors, or supervisory, within the meaning of the National Labor Relations Act, as amended ("NLRA").
- 43.2 The Employer further agrees that it will not challenge in any manner CRONA's right to represent any Nurse employed in any job classification covered by the Agreement (including but not limited to Resource Nurses) based on a claim, in whole or in part, that such Nurse is or may be a supervisor within the meaning of the NLRA.
- 43.3 If, at the expiration of the Agreement, the Employer and CRONA agree to extend the terms of the Agreement while they negotiate a successor agreement, subsections 1 and 2 shall remain binding on the Employer during the period of the extension.
- 43.4 This Section 43 is intended to preserve the existing covered classifications and is not intended to expand the coverage of Section 1, Recognition and Coverage.

SIGNATURE PAGE

Dated:, 2022	11/28/2022 9:43 AM PST Dated:, 2022					
Committee For Recognition Of Nursing Achievement	Stanford Health Care					
Colleen Borges	Mary Gaines					
Colleen Borges, R.N. President, CRONA fatan Stomferg	Mary Gaines Director, Employee & Labor Relations Julie Tisnado					
Kathy Stormberg, R.N. Vice President, CRONA <i>Jackie Campbell, RN</i>	Julie Tisnado Associate Chief Nursing Officer, Ambulatory Care Patrice Callagy					
Jackie Campbell, R.N. Savah Magalin	Patrice Callagy Exec. Dir, Emergency Services					
Sarah Moraga, R.N.	Kevin Tsui Director Strategic Services PCS					
Mark O'Neill, R.N. Kimberley. Reed	Grissel Hernandez Exec. Dir CEPD Mark Kosuwr					
Kimberley Reed, R.N. InnaMaria Varo	Mark Roesner Director Clinical Services, Operating Rooms Mulissa I. Waltur					
AnnaMarie Varo, R.N. Heine Yılmı	Melissa (Aimee) Walter, Exec. Director, Ambulatory Care Lonne ฟ ปาดคุ					
Helina Yilma, R.N. BJ (Lusholm	Donna Healy, Admin. Director Cancer Network NSG Dubbic Bishop					
B.J. Chisholm Labor Counsel Amanda Lynch	_ Debbie Bishop, Payroll Manager Aaron L. Agenbroad					
Amanda Lynch Labor Counsel	_ Aaron Agenbroad Labor Counsel <i>Dayme Sanchey</i>					
	Dayme Sanchez Labor Counsel					

APPENDIX "A" WAGE SCHEDULE

<u>Basic Salary</u>. Effective April 1, 2022 and extending through March 31, 2025 the following will be the minimum salaries, subject to the requirements for step advancement, for Nurses in the following classifications.

Effective at the start of the pay period closest to April 1, 2022 (5% wage increase for Clinical Nurse I, II, III, IV) (Relief differential for C and D classifications: \$7.50; Relief differential for Limited, A, and B classifications: \$10.00):

Step	Clinical Nur	Clinical Nurse I		Clinical Nurse II		se III Clinic	al Nurse IV	
1	76.2	:7	-		-		-	
2			79.22		82.87		86.56	
3			82.47		86.16		90.01	
4					89.61		93.69	
5			89.27		93.27		97.37	
6			92.76		96.94		101.27	
7*			96.59		100.87		105.43	
8**			97.56		101.86		106.48	
9***			99.51		103.91		108.62	
10^			101.47		105.98		110.79	
11^^			102.50		107.04		111.90	
12^^^			103.54		108.10		113.02	
Step	Relief Clinical Nurse I (Limited, A and B)	Relief Clinical Nurse II (Limited, A and B)	Relief Clinical Nurse III (Limited, A and B)	Relief Clinical Nurse IV (Limited, A and B)	Relief Clinical Nurse I (C and D)	Relief Clinical Nurse II (C and D)	Relief Clinical Nurse III (C and D)	Relief Clinical Nurse IV (and D)
1	86.27	-	-	-	83.77	-	-	-
2	-	89.22	92.87	96.56	-	86.72	90.37	94.06
3	-	92.47	96.16	100.01	-	89.97	93.66	97.51
4	-	95.77	99.61	103.69	-	93.27	97.11	101.19
5	-	99.27	103.27	107.37	-	96.77	100.77	104.87
6	-	102.76	106.94	111.27	-	100.26	104.44	108.77
7*	-	106.59	110.87	115.43	-	104.09	108.37	112.93
8**	-	107.56	111.86	116.48	-	105.06	109.36	113.98
9***	-	109.51	113.91	118.62	-	107.01	111.41	116.12
10^	-	111.47	115.98	120.79	-	108.97	113.48	118.29
11^^	-	112.50	117.04	121.90	-	110.00	114.54	119.40
12^^^	-	113.54	118.10	123.02	-	111.04	115.60	120.52

Longevity Step 7 after completion of 2 years at Step 6.

** Longevity Step 8 at 10.5 years of continuous service.

- *** Longevity Step 9 at 15 years of continuous service.
- ^ Longevity Step 10 at 20 years of continuous service.

^{^^} Longevity Step 11 at 25 years of continuous service.

AMA Longevity Step 12 at 30 years of continuous service.

Effective at the start of the pay period closest to December 1, 2022, which the parties agree for purposes of this Agreement will be November 27, 2022 (2% wage increase for Clinical Nurse I, II, III, IV) (Relief differential for C and D classifications: \$7.50; Relief differential for Limited, A, and B classifications: \$10.00):

Step	Clinical Nurse I	Clinical Nurse II	Clinical Nurse III	Clinical Nurse IV
1	77.80	-	-	-
2		80.80	84.53	88.29
3		84.12	87.88	91.81
4		87.49	91.40	95.56
5		91.06	95.14	99.32
6		94.62	98.88	103.30
7*		98.52	102.89	107.54
8**		99.51	103.90	108.61
9***		101.50	105.99	110.79
10^		103.50	108.10	113.01
11^^		104.55	109.18	114.14
12^^^		105.61	110.26	115.28

Step	Relief Clinical Nurse I (Limited, A and B)	Relief Clinical Nurse II (Limited, A and B)	Relief Clinical Nurse III (Limited, A and B)	Relief Clinical Nurse IV (Limited, A and B)	Relief Clinical Nurse I (C and D)	Relief Clinical Nurse II (C and D)	Relief Clinical Nurse III (C and D)	Relief Clinical Nurse IV (C and D)
1	87.80	-	-	-	85.30	-	-	-
2	-	90.80	94.53	98.29	-	88.30	92.03	95.79
3	-	94.12	97.88	101.81	-	91.62	95.38	99.31
4	-	97.49	101.40	105.56	-	94.99	98.90	103.06
5	-	101.06	105.14	109.32	-	98.56	102.64	106.82
6	-	104.62	108.88	113.30	-	102.12	106.38	110.80
7*	-	108.52	112.89	117.54	-	106.02	110.39	115.04
8**	-	109.51	113.90	118.61	-	107.01	111.40	116.11
9***	-	111.50	115.99	120.79	-	109.00	113.49	118.29
10^	-	113.50	118.10	123.01	-	111.00	115.60	120.51
11^^	-	114.55	119.18	124.14	-	112.05	116.68	121.64
12^^^	-	115.61	120.26	125.28	-	113.11	117.76	122.78

* Longevity Step 7 after completion of 2 years at Step 6.

Longevity Step 10 at 20 years of continuous service. Longevity Step 11 at 25 years of continuous service. ~~

^^^

^{**} Longevity Step 8 at 10.5 years of continuous service. ***

Longevity Step 9 at 15 years of continuous service. ۸

Effective at the start of the pay period closest to April 1, 2023, which the parties agree for purposes of this Agreement will be April 2, 2023 (5% wage increase for Clinical Nurse I, II, III, IV) (Relief differential for C and D classifications: \$7.50; Relief differential for Limited, A, and B classifications: \$10.00):

Step	Clinical Nurse I	Clinical Nurse II	Clinical Nurse III	Clinical Nurse IV
1	81.69	-	-	-
2		84.84	88.76	92.70
3		88.33	92.27	96.40
4		91.86	95.97	100.34
5		95.61	99.90	104.29
6		99.35	103.82	108.47
7*		103.45	108.03	112.92
8**		104.49	109.10	114.04
9***		106.58	111.29	116.33
10^		108.68	113.51	118.66
11^^		109.78	114.64	119.85
12^^^		110.89	115.77	121.04

Step	Relief Clinical Nurse I (Limited, A and B)	Relief Clinical Nurse II (Limited, A and B)	Relief Clinical Nurse III (Limited, A and B)	Relief Clinical Nurse IV (Limited, A and B)	Relief Clinical Nurse I (C and D)	Relief Clinical Nurse II (C and D)	Relief Clinical Nurse III (C and D)	Relief Clinical Nurse IV (C and D)
1	91.69	-	-	-	89.19	-	-	-
2	-	94.84	98.76	102.70	-	92.34	96.26	100.20
3	-	98.33	102.27	106.40	-	95.83	99.77	103.90
4	-	101.86	105.97	110.34	-	99.36	103.47	107.84
5	-	105.61	109.90	114.29	-	103.11	107.40	111.79
6	-	109.35	113.82	118.47	-	106.85	111.32	115.97
7*	-	113.45	118.03	122.92	-	110.95	115.53	120.42
8**	-	114.49	119.10	124.04	-	111.99	116.60	121.54
9***	-	116.58	121.29	126.33	-	114.08	118.79	123.83
10^	-	118.68	123.51	128.66	-	116.18	121.01	126.16
11^^	-	119.78	124.64	129.85	-	117.28	122.14	127.35
12^^^	-	120.89	125.77	131.04	-	118.39	123.27	128.54

* Longevity Step 7 after completion of 2 years at Step 6.

** Longevity Step 8 at 10.5 years of continuous service.

۸ Longevity Step 10 at 20 years of continuous service.

~~ Longevity Step 11 at 25 years of continuous service. ^^^

Longevity Step 12 at 30 years of continuous service.

^{***} Longevity Step 9 at 15 years of continuous service.

Effective at the start of the pay period closest to April 1, 2024, which the parties agree for purposes of this Agreement will be March 31, 2024 (5% wage increase for Clinical Nurse I, II, III, IV) (Relief differential for C and D classifications: \$7.50; Relief differential for Limited, A, and B classifications: \$10.00):

Step	Clinical Nurse I	Clinical Nurse II	Clinical Nurse III	Clinical Nurse IV
1	85.77	-	-	-
2		89.08	93.20	97.34
3		92.75	96.88	101.22
4		96.45	100.77	105.36
5		100.39	104.90	109.50
6		104.32	109.01	113.89
7*		108.62	113.43	118.57
8**		109.71	114.56	119.74
9***		111.91	116.85	122.15
10^		114.11	119.19	124.59
11^^		115.27	120.37	125.84
12^^^		116.43	121.56	127.09

Step	Relief Clinical Nurse I (Limited, A and B)	Relief Clinical Nurse II (Limited, A and B)	Relief Clinical Nurse III (Limited, A and B)	Relief Clinical Nurse IV (Limited, A and B)	Relief Clinical Nurse I (C and D)	Relief Clinical Nurse II (C and D)	Relief Clinical Nurse III (C and D)	Relief Clinical Nurse IV (C and D)
1	95.77	-	-	-	93.27	-	-	-
2	-	99.08	103.20	107.34	-	96.58	100.70	104.84
3	-	102.75	106.88	111.22	-	100.25	104.38	108.72
4	-	106.45	110.77	115.36	-	103.95	108.27	112.86
5	-	110.39	114.90	119.50	-	107.89	112.40	117.00
6	-	114.32	119.01	123.89	-	111.82	116.51	121.39
7*	-	118.62	123.43	128.57	-	116.12	120.93	126.07
8**	-	119.71	124.56	129.74	-	117.21	122.06	127.24
9***	-	121.91	126.85	132.15	-	119.41	124.35	129.65
10^	-	124.11	129.19	134.59	-	121.61	126.69	132.09
11^^	-	125.27	130.37	135.84	-	122.77	127.87	133.34
12^^^	-	126.43	131.56	137.09	-	123.93	129.06	134.59

* Longevity Step 7 after completion of 2 years at Step 6.

** Longevity Step 8 at 10.5 years of continuous service.

*** Longevity Step 9 at 15 years of continuous service.

Longevity Step 10 at 20 years of continuous service.
Longevity Step 11 at 25 years of continuous service.

Longevity Step 11 at 25 years of continuous service.

GLOSSARY

The following definitions are intended to clarify the use of certain phrases and terms used throughout this Agreement. It is not necessarily all-inclusive.

<u>Hourly Base Wage Rate</u>: The actual hourly base pay, excluding any applicable shift differential, overtime, Employer paid benefits, or other allowances or premiums.

<u>Regular Hourly Rate of Pay</u>: An hourly rate which is equal to the Nurse's hourly base wage rate inclusive of any applicable shift differential.

<u>Overtime and Holiday Pay</u>: An hourly rate which is equal to the Nurse's hourly base wage rate plus applicable shift differential plus one-half the regular hourly rate of pay as defined by the Fair Labor Standards Act.

<u>Premium Pay</u>: A rate of pay which is equal to one and one-half times the Nurse's hourly base wage plus any applicable shift differential.

<u>Work Day</u>: The Nurse's individual work day which is the consecutive twenty-four hour period beginning with the time the Nurse is actually scheduled to begin work. Example: A Nurse usually scheduled to work the day shift beginning at 6:45 a.m. has an individual work day from 6:45 a.m. one calendar day until 6:44 a.m. the next calendar day.

<u>Salary Review Date</u>: The date of eligibility for increase in rate of pay as established in Section 7.1 and modified for Relief Nurses in Section 15.13.1. If the Nurse's hire date is adjusted forward, the salary review date is also adjusted forward.

Hire Date: Most recent date of employment (first day of work) with the Employer.

<u>Adjusted Hire Date</u>: Most recent date of hire with the Employer, unless hire was as an "A" or Limited Relief Nurse, see section 37.5, and as otherwise may be modified by the terms of this Agreement.

<u>Absent "A" Time</u>: Time off given to staff who are not needed to work for the shift or part of the shift on their unit.

<u>Absent "A" Time Credit</u>: Refers to hours accrued on time worked which can be used if the Nurse so desires to cover "A" time taken, whether voluntary or involuntarily.

SIDE LETTER Between Stanford Health Care and Committee for Recognition of Nursing Achievement (CRONA) 2010

During negotiations between Stanford Health Care and CRONA the following items were agreed to be included in a side letter for the term of the current agreement.

A. <u>Parking</u>.

The Employer recognizes the concerns raised by CRONA over security and parking and will use its best efforts to work with Stanford University to provide adequate shuttle service and security for Nurses on all shifts. During the term of the agreement, the parties will continue to explore problem-solving issues regarding parking and lighting at the Quarry Road parking lot.

B. <u>Staffing</u>.

For the term of this Agreement, the parties agree that if there are issues under Section 33.3.3.a which have been referred to the Vice President of Patient Care Services and the President of CRONA and have not been resolved, they will engage a mediator through the Federal Mediation and Conciliation Service to facilitate consensus building. The parties further agree that notwithstanding this side letter and Section 33.3.3.a the Employer retains the ultimate discretion regarding issues that arise under Section 33.

C. <u>Relief Nurses</u>.

The employer shall not unreasonably restrict a Relief Nurse "A" from moving to Relief Nurse "B".

SIDE LETTER Between Stanford Health Care and Committee for Recognition of Nursing Achievement (CRONA) 2010

The following side letters were previously agreed to between the Stanford Health Care and CRONA and will be continued for the term of the current agreement.

A. <u>Weekend Differential</u>. (8/20/96)

This letter will record and confirm our agreement pertaining to the interpretation of the application of weekend differential contained in the 1996-1998 collective bargaining agreement.

Weekend differential will be paid to Nurses for productive hours worked on a weekend. Refer to Section 7.5.4 of the Agreement for details. The weekend for the purpose of weekend differential pay only is defined as follows:

Saturday and/or Sunday for all or part of the day and evening shifts; Night shift beginning Friday and/or Saturday night unless otherwise designated by agreement between Nursing Administration and the Nurse (i.e., Saturday and/or Sunday; Friday and/or Sunday).

In order to clarify in more detail how the definition of the weekend and the application of the weekend differential would apply, the following examples will serve as guidelines:

A Day or Evening Nurse:

Who is called into work the Friday night shift is eligible for weekend differential. However, the Nurse who works 3:00 PM to 11:00 PM on a Friday is not eligible.

Who works a Sunday night shift is not eligible. However, the Nurse who works 3:00 PM to 11:00 PM on Sunday is eligible.

A Night Nurse:

Who works 7:00 PM Friday to 7:00 AM Saturday is eligible. However, the Nurse who works 7:00 PM Sunday to 7:00 AM Monday is not eligible, unless this is the designated defined weekend agreement.

Who works Friday night, but who has an agreement with Nursing that the weekend designation is Saturday and Sunday nights, is not eligible.

All Nurses:

Who are on absent time or PTO are not eligible.

Who work overtime hours which are not part of the weekend shift as defined above, but which are contiguous to a weekend shift worked, will receive the weekend differential if the overtime is for a period of less than three (3) hours. If the overtime is for a period of three (3) hours or more, weekend differential is not applicable to the shift worked. Conversely, Nurses who work overtime hours which are part of the weekend shift, but which are contiguous to a

non-weekend shift worked, will not receive weekend differential unless the overtime is for a period of three (3) or more hours.

For example:

An evening Nurse who works 3:00 PM Friday to 12:01 AM Saturday does not receive weekend differential. If that Nurse continues to work until 4:00 AM Saturday, the Nurse is eligible for weekend differential for the entire overtime shift.

An evening Nurse who works 3:00 PM Sunday to 12:01 AM Monday receives the weekend differential. If that Nurse continues to work until 4:00 AM Monday, the Nurse does not receive weekend differential for the overtime shift worked.

B. <u>16-Hours</u>. <u>(8-20-96)</u>

CRONA and the Employer agree that it is in the interest of the Employer and the Nurses to relieve Nurses who have worked sixteen contiguous hours who request to be relieved of duty. If the Nurse does request to be relieved of duty, the Employer will use its best efforts to relieve the Nurse as soon as practicable after the request is made.

C. Vacation and Holiday Time. (6/11/98)

We will continue to grant vacation and holiday time based on seniority and unit equity. Disputes will be reviewed and resolved on a case by case basis.

SIDE LETTER Between Stanford Health Care and CRONA Re On Call and Overtime Issues 2016

This Side Letter applies to the Employees in Units that utilize an on call system.

A. On Call Shift Selection/Assignment Procedures

With the exception of the Cath Lab, the on call shift selection and assignment procedures in effect in the various departments will remain in effect unless and until the Unit/Department on call committee recommends a change which is adopted by the vote of 2/3 of the employees performing call in the Unit/Department, and reviewed and approved by the Employer.

In the Cath Lab and in any new unit where call is utilized during the life of this Side Agreement prior to the development of an on call selection and assignment procedure by an on call committee, and adoption by 2/3 of the employees to perform on call in the unit and review and approval by the Employer, the following selection process will apply:

Each Nurse will select one weekday on call shift and, if there is weekend call, one weekend on call shift in order of seniority until every Nurse eligible to take call has had an opportunity to select one shift. Thereafter, Nurses may select one or more additional shifts in order of seniority until all shifts are selected, or no one wishes to select additional shifts. The Employer may limit the number of shifts selected by any Nurse based upon considerations of patient and employee safety.

If all of the on call shifts are not selected through the preceding process, the Employer will assign on call shifts by inverse seniority, except that it will first assign a shift to any Nurse who declined to select any on call shifts in the voluntary selection process, allowing the Nurse to select which of the unassigned on call shifts he or she wishes from the shifts remaining at the time, until all of the shifts have been assigned. The Employer may decline to assign a shift to a junior Nurse based upon considerations of patient and employee safety.

With specific regard to the Cath Lab, in adopting this selection process, the on call schedule will cover a period of twelve (12) weeks, and weekend shifts remain at 48 hours in length.

The Hospital shall provide written notice to CRONA of a vote of Nurses under this provision at least seven (7) days prior to the vote. The notice shall include the subject of the vote, the Nurses eligible to vote, and the process that will be used to conduct the vote and tally the results. The Hospital shall notify CRONA of the results of the vote within seven (7) days of the conclusion of the vote.

B. On Call Schedules

On call schedules will be posted at least two (2) weeks prior to the effective date of the on call schedule.

C. Call Back Contiguous With a Scheduled Shift

Contiguous Call Back Work shall be defined as call back work that meets all of the following requirements:

- 1. The Nurse's on call shift (regardless whether assigned or voluntary) begins immediately upon the end of the Nurse's regular work shift; and
- 2. The Nurse either is advised that his or her services on a call back basis will be required at the beginning of the on call shift or is called and reports back to work within ninety (90) minutes of the end of her/his regular work shift; and
- 3. The call back work is not time spent beyond the shift to finish reporting or charting, or to finish the Nurse's own assigned duties where the time spent is sixty (60) minutes or less.

If all of these requirements for Contiguous Call Back Work are satisfied, the Nurse shall be paid an additional half-time premium for all hours worked during the eligible on call shift (including the first hour worked during that on call shift) on the second and subsequent incidents of Contiguous Call Back Work during each pay period.

The additional half time premium to be paid for eligible Contiguous Call Back Work shall be double time if the call back would otherwise be paid at time and one-half, and double time and a half if the call back pay would otherwise be paid at double time.

D. <u>Mandatory Overtime</u>

If, during a pay period, a Nurse has already performed mandated overtime and the Nurse is subsequently mandated by the Employer to perform additional overtime during the same pay period, the mandated overtime hours worked on the second occasion and all subsequent mandated hours of overtime worked during the pay period shall be compensated at two times (2X) the Nurse's regular rate of pay if the overtime would otherwise be at time and one-half (I-I/2X), or at double time and one-half (2-1/2X) of the Nurse's regular rate if the overtime would otherwise be at double time (2X). In order for the overtime to be "mandated" or "mandatory," the Nurse must have refused in writing to take the assignment, and thereafter been directed to do so by the Employer. Overtime at the end of the Nurse's shift of up to one hour to complete a procedure, to complete charting or report, or otherwise to finish the Nurse's own assigned duties will not be considered to be mandated overtime for purposes of this Side Letter.

E. <u>Unavoidable Conflicts and Unforeseen Emergencies</u>

If a Nurse has a specific unavoidable commitment or an emergency that affects the Nurse's ability to stay beyond the end of his or her shift beyond the time required to complete charting and report, or to finish his/her own assigned duties, the Nurse will inform the Patient Care Manager, Assistant Patient Care Manager, or in their absence the Resource Nurse, of the unavoidable conflict or emergency as far in advance as is possible. Should the need for overtime thereafter arise on the date and shift involved, the Employer will make all reasonable efforts to secure the needed coverage by other

Nurses who are present and working, or by floating in order to accommodate the conflict or emergency if possible. The Nurse so accommodated will be expected to be reasonably available to assist in providing coverage should another Nurse have an unavoidable conflict or emergency on a subsequent occasion.

If a Nurse has volunteered for or been assigned to an on call shift and is unable to perform it because of an unavoidable conflict or emergency, the Nurse will inform the Patient Care Manager, the Assistant Patient Care Manager, or in their absence the Resource Nurse, as far in advance as is possible, and if the conflict is known in advance will make all reasonable efforts to obtain a trade of on call shifts with another Nurse. If the Nurse cannot obtain a trade of on call shifts with another Nurse to take the on call shift with the result that the Employer is

required to provide the necessary coverage by involuntary assignment to another Nurse or Nurses, the Nurse who was unable to work the shift as scheduled will be expected to be reasonably available to assist in providing coverage by a trade of on call shifts, or by volunteering to take an additional on call shift, should another Nurse subsequently have an unavoidable conflict or emergency requiring similar efforts to accommodate his/her need.

While the Employer will continue generally to make efforts to obtain voluntary coverage for overtime and on call shifts, it is understood that the provisions described above are intended to address the occasional unavoidable conflict and emergency situations only.

The provisions of this Section do not constitute a guarantee that the Employer will be able to accommodate the occasional unavoidable conflict or unforeseen emergency in every instance.

Disputes concerning the application and interpretation of the provisions of this Side Letter will be subject to the grievance and arbitration provisions of the Master Agreement between the parties.

SIDE LETTER Between Stanford Hospital and CRONA Re Resource Nurse Positions 2004

CRONA recognizes the Hospital's concerns regarding the continuity of the application of the Resource Nurse role. CRONA and the Hospital also recognize that each and every Nurse who fulfills that role on any shift is deserving of acknowledgement of the role they play and should be compensated therefor. All Nurses who act in the role of Resource Nurse will receive appropriate education and training.

Any program to establish a regular designated Resource Nurse position for any unit will be discussed and details finalized in Nurse Practice Committee. Such details should include (but are not limited to): the minimum number of clinical hours required to maintain clinical skills, etc., the clinical and leadership requirements, the number of permanent Resource Nurses required, and an evaluation process for the program.

SIDE LETTER Between Stanford Health Care and CRONA Re Use of Term "Regular" in 2011 – 2013 Agreement

In conjunction with their 2010 – 2011 negotiations, Stanford Health Care and CRONA revised Section 1.2 to clarify the terms "Regular Nurse" and "Relief Nurse" and the classifications covered by the 2011 – 2013 collective bargaining agreement ("Agreement"). In addition, the parties also agreed to the following:

"[T]he various references in the Agreement to the different categories and levels of nurses covered by this Agreement shall be revised to be consistent with the amended Section 1.2. The parties agree that these revisions are intended to provide consistency and clarity in the terminology used in the Agreement and are not intended to either expand or contract the substantive rights, benefits, and obligations provided by the Agreement."

During the course of updating the Agreement to reflect the agreed-upon terms, various issues arose regarding the replacement of the term "Staff" with "Regular," including the applicability of various provisions of the Agreement to Relief Nurses. In order to resolve those issues without modifying either party's underlying position regarding the applicability of certain provisions to Regular and/or Relief Nurses, the parties have agreed to this Side Letter and the following:

- 1. Despite the use of the term "Staff" in prior collective bargaining agreements, CRONA is preserving its right to argue that the past practices of the parties modified the terms of those contracts to the extent that certain provisions of those agreements that, on their face, applied just to Regular nurses were modified by the parties to apply to both Regular and Relief nurses. In addition, the use of the term "Regular" in a provision of the 2011-2013 agreement shall not be deemed a waiver by CRONA of its ability to argue that the provision has, in fact, been applied in the past to both Regular and Relief nurses. The fact that CRONA has permitted the term 'Regular Nurse' to be used in the 2011-13 Agreement in a provision in dispute shall not be interpreted as an agreement by CRONA that the provision does not apply to Relief Nurses.
- 2. The parties agree that SHC can, in future disputes over the applicability of a provision to Relief Nurses, argue that the use of the term "Staff Nurse" in that provision in contracts prior to the 2011-2013 Agreement shows that it was not meant to apply to Relief Nurses, but SHC agrees not to assert an argument that the use of the term "Regular Nurse" in the 2011-2013 Agreement demonstrates that CRONA agreed that the provision does not apply to Relief Nurses.
- 3. SHC acknowledges and agrees that a past practice exists that Relief Nurses have been paid in accordance with Section 9.4.5 (Holidays) [now Section 10.1.5], even though this provision in [] prior agreements stated on its face that it was applicable to "Staff" nurses. CRONA has agreed to replace the word "Staff" with "Regular" in Section 9.4.5 of the 2011-2013 Agreement [now Section 10.1.5], but the parties agree that this should not be interpreted as an agreement by CRONA that the provision does not apply to Relief Nurses currently and in the future.
- 4. SHC acknowledges and agrees that a past practice exists that Section 31.2 [now Section 36.2] has been applied to both Regular and Relief Nurses who meet the criteria of this
section, even though this section in prior agreements stated on its face that it was applicable only to "Staff" nurses. CRONA has agreed to replace the word "Staff" with "Regular" in Section 31.2 of the 2011-2013 Agreement [now Section 36.2], but the parties agree that this should not be interpreted as an agreement by CRONA that the provision does not apply to Relief Nurses currently and in the future.

SIDE LETTER Between Stanford Health Care and CRONA Re Use of Travelers 2022

During negotiations conducted in 2022 as well as prior negotiations, Stanford Health Care and CRONA discussed the Hospital's use of Travel Nurses ("Travelers"), including CRONA's concerns regarding extent of the use of Travelers at the Hospital. In light of these discussions, the parties have agreed as follows:

- 1. The Hospital will include the commitment level and active status of Nurses represented by CRONA in the information provided on a monthly basis to CRONA, in Excel or a comparable electronic format, under Section 4.3 of the Agreement;
- 2. The vacancies posted at the Hospital shall remain available online to CRONA;
- 3. On a monthly basis, the Hospital shall provide a list of all posted vacancies for positions within the bargaining unit, by Cost Center, that have remained unfilled for more than eight (8) weeks;
- 4. Upon CRONA's written request, which may be made no more often than every month, the Hospital shall provide to CRONA a report showing the number and monthly total of FTEs of Travelers used by each unit or cost center, along with the commitment levels of the individual Travelers used in each unit or cost center, since the last such report was provided to CRONA, up to the prior twelve (12) months. The report shall be provided in Excel or a comparable electronic format; and
- 5. The Hospital's use of Travelers, including but not limited to the use reflected in and/or related to the information provided by the Hospital to CRONA pursuant to this Side Letter, may be discussed at the Joint Conference. If CRONA seeks to discuss this topic at a Joint Conference meeting, CRONA shall give at least seven (7) calendar days notice to the Hospital.

This Side Letter is without prejudice to either CRONA's position or the Hospital's position regarding any contractual or other limitations on the Hospital's use of Travelers. CRONA has reserved any and all rights it has under the collective bargaining agreement to challenge the Hospital's use of Travelers, and the Hospital has reserved any and all rights it has to use Travelers at the Hospital.

SIDE LETTER Between Stanford Health Care and CRONA Re Attendance and Pre-Approved Vacation and Education Days Policies 2022

Stanford Health Care and CRONA have agreed as follows:

- The Hospital shall not change the points definitions in Sections III(C) and (D) or Section V(B)("Standards – Including Adjustments for FTE Commitment and Length of Shift Differences") of the Hospital's Attendance Policy provided, however, that the parties have agreed to changes to the Attendance Policy provided to CRONA by the Hospital on March 22, 2019; and
- 2. In conjunction with their 2016 2019 negotiations, the Hospital proposed changes to its current Pre-Approved Vacation Policy and its Staffing and Scheduling Guidelines ("Guidelines"), and CRONA commented and proposed revisions. The Hospital has agreed to adopt the revised policies, pursuant to its standard protocols related to the adoption of policies. The revised Pre-Approved Vacation Policy has been re-titled as the Pre-Approved Vacation and Pre-Approved Education Days Policy ("Pre-Approval Policy").
- 3. The Hospital shall not change the following:
 - a. The order of scheduling in Section II(M)(1-5) of the Pre-Approval Policy and in Section II(A)(1)(1-5) of the Scheduling Guidelines;
 - b. The number of weeks that may be used for Pre-Approved Vacation and the number of hours that may be used for Pre-Approved Education set forth in Section II(F) of the Pre-Approval Policy, except that the Hospital will amend Section II(G) to increase by one (1) week for each seniority tier the number of weeks that may be used for pre-approved vacation. This change shall be effective with pre-scheduled vacations commencing on January 1, 2024, which may be pre-scheduled beginning January 1, 2023; and
 - c. The authorization to submit pre-approved education days and pre-approved vacation days up to one year in advance set forth in Section III(1)(a and b) in the Pre-Approval Policy.
- 4. Except as limited herein and by the Agreement, the Hospital may modify its Attendance Policy, Pre-Approval Policy, and Guidelines.

SIDE LETTER Between Stanford Health Care and CRONA Re 500 Pasteur Dr. 2019

In conjunction with their 2019-2022 negotiations, Stanford Health Care and CRONA discussed the Hospital's plan to open a new hospital at 500 Pasteur Drive.

The parties agree that they will discuss the Hospital's plans to staff the new hospital with Nurses. The parties agree that they will discuss the consolidation or re-aggregation of existing units at least thirty (30) days prior to the Hospital posting Nurse positions in a unit that is subject to such consolidation or re-aggregation.

The Hospital acknowledges that its intent is that Nurses who are affected by the move from the existing hospital to the new hospital will be provided the opportunity to maintain their current commitment levels and shifts, and agrees to use reasonable efforts to provide Nurses with such an opportunity. The Hospital also agrees to discuss implementation of this process with CRONA through Nurse Practice Committee.

CRONA acknowledges that this Side Letter does not affect the Hospital's management rights under the Agreement, except to the extent it obligates the Hospital to use reasonable efforts to provide Nurses the opportunity to maintain their current commitment levels and shifts, as provided above.

SIDE LETTER Between Stanford Health Care And CRONA Re Temporary Relief Position While Pursuing Education 2019

- A. <u>Purpose</u>. The purpose of this Side Letter is to provide eligible Nurses the opportunity to move to a Relief B position temporarily to allow them to pursue a BSN or MSN degree while continuing to work at the Employer.
- **B.** <u>Eligibility and Qualifications</u>. All 0.8 time or more Regular Nurses with five (5) or more years of continuous service with the Employer and who are enrolled in an accredited educational program culminating in a BSN or MSN degree are eligible to request a temporary Relief B position and to apply for a posted temporary Relief B position under this Side Letter.
- C. <u>Number of Temporary Relief Positions</u>. In each unit with at least ten (10) Regular Nurses, one temporary Relief Nurse position will be made available under this Side Letter, with one (1) additional temporary Relief Nurse position made available for every seventy-five (75) Regular Nurses who are regularly assigned to the unit. As an example, a unit with eighty (80) Regular Nurses assigned to the unit shall have two (2) temporary Relief Nurse positions available.
- D. <u>Assignment to Temporary Relief Position</u>. At least sixty (60) days in advance of the requested start of the temporary relief position, an eligible Nurse may submit a written request to the Nurse's supervisor for a temporary Relief B position under this Side Letter. Upon receipt of the request, if the unit does not currently have the full number of temporary Relief positions provided for in paragraph (C) filled or if the unit's temporary Relief positions are all filled but at least one of those positions is currently occupied by a returning Nurse as defined in paragraph (I), the Employer will post to all Nurses on the unit a temporary Relief Nurse position open only to Nurses who are eligible under paragraph (B). All such eligible Nurses may apply for the posting. If the unit's temporary Relief positions are all filled but at least one of those position by a returning Nurse, any temporary Relief position posted in the unit shall be filled when the returning Nurse vacates the temporary Relief position.
- E. <u>Selection</u>. Selection among eligible applicants shall be based on seniority.
- **F.** <u>**No Trial Period**</u>. A Nurse granted a temporary Relief position under this Side Letter will not serve a trial period as a Relief Nurse.
- **G.** <u>**Return to Open Position**</u>. A Nurse who receives a temporary Relief position shall have the right to apply at any time for any available posted position for which the Nurse is qualified, pursuant to the procedures of Section 23 (Vacancies and Internal Transfers). A Nurse granted a temporary Relief position who subsequently returns to a Regular Nurse position or accepts another position will have no adjustment to her/his hire date, except that the provisions of Section 23.4 shall apply to the Nurse.
- **H.** <u>**Documentation**</u>. A Nurse granted a temporary Relief position under this Side Letter will provide documentation of enrollment in the accredited educational program each academic term to the Employer.

I. <u>End of Temporary Relief Nurse Position</u>. The term of a temporary Relief position under this Side Letter shall last so long as the Nurse is enrolled in the degree program, for up to a maximum of two years. When the term of the temporary Relief position ends, the Nurse, referred to as a "returning Nurse" for purposes of this Side Letter, shall have up to three (3) additional months in the Relief position to obtain a vacant permanent position. A returning Nurse who has not obtained a permanent position by the end of the three (3) months shall be separated from employment.

SIDE LETTER Between Stanford Health Care and CRONA Re Student Loan Repayment Program 2022

The Hospital shall establish a student loan repayment program. Payments made pursuant to the student loan repayment program may be used for principal or interest payments on qualified educational loans incurred by an eligible Nurse for the purpose of obtaining a nursing degree (i.e., ADN, BSN, MSN, DNP, or Ph.D in Nursing) and shall not exceed the amount owed by the Nurse. The program shall allow repayments as provided in this Side Letter:

1) Tuition Reimbursement Funds: Effective on September 1, 2022, a Nurse eligible for educational assistance and professional enrichment under Section 14.2 may elect to use up to one hundred percent (100%) of the Nurse's annual educational reimbursement funds under Section 14.3.3 for eligible student loan payments. The amount elected by the Nurse may not exceed the maximum tax-free student loan payment amount permitted by the Coronavirus Aid, Relief, and Economic Security Act ("CARES Act"), as extended by the Consolidated Appropriations Act of 2021. Any amount of a Nurse's annual educational reimbursement that a Nurse does not elect to use for eligible student loan payments shall remain available for use pursuant to the terms of Section 14.

In advance of each academic year, a Nurse electing to use all or a portion of the Nurse's educational reimbursement funds for student loan payments shall make an election to use a designated portion of the Nurse's educational reimbursement funds for this purpose and shall agree to the terms of the program, including providing relevant information to process the loan payment and connecting the Nurse's student loan service provider to the Employer's Student Loan Benefit system to receive contributions.

The Employer shall prepare the application procedures for Nurses to make this election and shall establish a process for the submission of documentation regarding qualifying loans, which shall be discussed in advance with CRONA.

All payments under this Side Letter shall be made in compliance with the law and IRS regulations, including applicable IRS limits. The student loan repayment program shall remain in place while allowed by applicable law.

SIDE LETTER Between Stanford Health Care and CRONA Re Strategic Recruitment and Retention Program 2022

Throughout the course of their 2022 negotiations, Stanford Health Care and CRONA discussed the increasing acuity of patients being cared for at the hospital and the impacts that this has had on bedside Nurses. The Hospital and CRONA are concerned that Nurses across the nation are leaving the profession due to burnout and other factors. At the same time and due in part to the excellent patient care delivered by the Nurses represented by CRONA, SHC has experienced significant growth in its patient care operations, leading to a significant increase in need for registered nurses at SHC. To address this growth in operations, SHC has a need both to recruit additional nurses and to retain its valuable incumbent nurses. During the period from 2019 – 2022, SHC has significantly increased its nursing workforce. Particularly coming out of the pandemic, however, certain patient care units have had more difficulty recruiting and in some circumstances retaining Clinical Nurses. CRONA and SHC share a common interest in recruiting and retaining qualified Nurses to care for SHC's patients. In an effort to help this common goal, SHC and CRONA have entered into this temporary Side Letter establishing a Strategic Recruitment and Retention Program under the terms outlined below:

- 1. Assuming a ratification date of May 1, 2022, this Side Letter shall become effective on June 1, 2022. If the Agreement is not ratified on May 1, 2022, this Side Letter shall become effective on the beginning of the quarter that commences at least sixty (60) days after ratification.
- 2. The quarterly periods for purposes of this Side Letter begin on June 1st, September 1st, December 1st, and March 1st.
- 3. To identify eligible units under this Side Letter, the parties look to the following metrics:
 - a. The time to fill a position: Regular Nurse Positions not filled for more than ninety (90) days after a vacancy in a given unit has been posted are "Hard to Fill" positions. The ninety (90) day period shall be determined by the requisition final approval date to the offer accepted date.
 - b. Vacancy rates: A patient care unit or area of practice shall be deemed to have a vacancy rate greater than twenty percent (20%), where the vacancy rate is determined by the number of open Hard to Fill positions relative to the target number of Regular Nurse positions in the unit.
 - c. Additional staffing needed due to the expansion of beds, services, or new units shall not be considered "Hard to Fill" positions for six (6) months after the requisitions for the new positions are made.
- 4. During the first six quarterly periods of this Side Letter, the following units shall be deemed "Hard to Fill Units" and are eligible to receive the incentive payments provided for under this Side Letter: M4, J2, J4, B2, E-Ground, E2, K4, L4, ED – Adult, ED – Peds, LifeFlight, CC Float Pool, and CC Response Team.
- 5. After the first six quarterly periods of this Side Letter, the eligible units for the incentive payments under this Side Letter shall be determined based on those units that are either (a) mutually agreed upon by the parties to be included (including any areas of practice that the parties agree to include); or (b) other units that, as of a Designated Date, have a vacancy rate of more than twenty percent (20%), as described in paragraph 3(b). The Designated Dates for quarterly periods are as follows:
 - a. October 31, 2023 for the December 1, 2023 and March 1, 2024 quarters;
 - b. April 30, 2024 for the June 1, 2024 and September 1, 2024 quarters; and
 - c. October 31, 2024 for the December 1, 2024 quarter.

- 6. SHC and CRONA shall meet no later than sixty (60) days before the first Designated Date (October 31, 2023) to commence discussions regarding the units that should be eligible for incentive payments under this Side Letter and the information available and necessary to the parties to facilitate those discussions, including the use of traveler nurses in the units. Thereafter, the parties shall meet no later than thirty (30) days before the remaining two Designated Dates (April 30 and October 31, 2024) for this discussion.
- 7. During the term of this Side Letter, Nurses assigned to a Hard to Fill unit, as defined in paragraphs 4 and 5 above, shall be eligible for a Retention Incentive Payment according to the following terms:
 - a. The Retention Incentive Payment shall be paid on a quarterly basis to those Nurses who:
 - i. Are assigned to an eligible unit throughout the applicable period, including at the beginning and end of each applicable period; and
 - ii. Remain active throughout the applicable period.
 - b. Each quarterly payment for eligible Full-Time (1.0 or .90 FTE) Regular Nurses shall be One Thousand Two Hundred Fifty Dollars (\$1,250.00). Eligible Part-Time Nurses shall be prorated based on hours paid during the applicable period, relative to a .90 FTE Nurse. For Relief Nurses, the quarterly payment shall be based on the hours worked by that Relief Nurse in an eligible unit during that quarterly period, relative to a .90 FTE Nurse. Notwithstanding the foregoing, no Nurse shall receive a quarterly payment higher than One Thousand Two Hundred Fifty Dollars (\$1,250.00).
 - c. The quarterly payments for eligible Nurses shall be paid to Nurses in the second paycheck following the end of the applicable period.
- 8. The parties will work together jointly to make this new program a success. This Side Letter shall expire at the end of the last quarterly period that ends prior to the expiration date of the Agreement, i.e., February 28, 2025, unless the parties agree that the program has been successful and agree to extend it.

SIDE LETTER Between Stanford Health Care and CRONA Re: Transfer Center Nurses 2021

On June 24, 2021, the National Labor Relations Board certified that the following group of Stanford Health Care ("SHC") employees is now part of CRONA's collective bargaining unit described in Section 1 of this Agreement:

All full-time and regular part-time Registered Nurses employed by the Employer at the Transfer Center within Stanford Hospital located at 300 Pasteur Dr., Palo Alto, CA; excluding Registered Nurses employed as managerial employees, confidential employees, office clerical employees, guards, and supervisors, as defined in the Act. Also eligible to vote are all employees in the unit who have worked an average of four (4) hours or more per week during the 13 weeks immediately preceding the eligibility date for the election.

These newly added members of CRONA's collective bargaining unit are referred to in this Side Letter as the "Transfer Center Nurses."

Application of the Agreement

SHC and CRONA agree that, unless otherwise indicated below:

- A. Beginning on October 31, 2021 (the "First Implementation Date"), the sections of the current SHC-CRONA collective bargaining agreement (the "Agreement") listed in Attachment A will apply to the Transfer Center Nurses, with the adjustments set forth in the Adjustments paragraphs below.
- B. Beginning on November 14, 2021 (the "Second Implementation Date"), the sections of the Agreement listed in Attachment B will apply to the Transfer Center Nurses, with the adjustments set forth in the Adjustments paragraphs below.
- C. Unless provided otherwise in Attachment A, B, or below, the Transfer Center shall continue to operate according to its current policies, protocols, and procedures, until January 8, 2022.
- D. Beginning on January 9, 2022 (the "Third Implementation Date"), all terms of the Agreement will apply to the Transfer Center Nurses, except as modified by the terms of this Side Letter.

Adjustments.

- 1. Classification and Steps. As of the Second Implementation Date, current Transfer Center Nurses will have the classification and step shown on Attachment C. During the term of this Side Letter, the Regular and Relief RNs assigned to the Transfer Center shall have one of the following classifications: Clinical Nurse II (Transfer Center); Clinical Nurse III (Transfer Center); Clinical Nurse IV (Transfer Center); Relief Clinical Nurse II (Transfer Center); Relief Clinical Nurse III (Transfer Center); After the Second Implementation Date, Transfer Center Nurses' step increases will be governed by the Agreement. Except as provided in this Side Letter, the terms of the Agreement applicable to Clinical Nurses and Relief Clinical Nurses shall be applicable to the Transfer Center Nurses.
- 2. **Wages.** From the Second Implementation Date through the date that the Nurse retires or leaves the Transfer Center, each current Transfer Center Nurse listed on Attachment C will

be paid pursuant to the Agreement at a base wage rate that is the higher of either (a) the Nurse's current base wage rate as of the day this Side Letter is executed, or (b) the base wage rate corresponding to the Nurse's classification and step pursuant to the Agreement, as provided by Attachment C.

3. Retroactive Pay for Interim Pay Period. After the Second Implementation Date, Transfer Center Nurses who submit a payroll correction request by no later than November 30, 2021, will receive retroactive payment for time worked during the single pay period between the First Implementation Date and the Second Implementation Date (the "Interim Pay Period"). This retroactive payment will be equal to the following: (a) any additional pay the Transfer Center Nurse would have received had the base wage rate corresponding to the Nurse's classification and step pursuant to the Agreement, as provided by Attachment C, been in effect during the Interim Pay Period; and (b) any additional pay that the Transfer Center Nurse would have received during the Interim Pay Period for work covered by Section 7.8 of the Agreement (Short Notice Compensation), less any additional compensation that the Transfer Center Nurse received for this work due to the Transfer Center's current pay practices.

4. Commitment, Shift, and Position.

- a. Until the Third Implementation Date, current Transfer Center Nurses will have the same commitment/FTE, shift, and position as their current commitment/FTE, shift, and position.
- b. Shifts that occur prior to the Third Implementation Date, including the shifts occurring between November 14, 2021 and January 8, 2022, will be scheduled based on the current scheduling procedures in the Transfer Center. For shifts occurring on and after the Third Implementation Date, the Transfer Center shall follow the Agreement's posting requirements in Sections 17 and 18.4.
- c. As of the Third Implementation Date, the following shall apply:
 - i. Current Regular Transfer Center Nurses will have the same commitment/FTE, shift (i.e., day 12-hour, evening 12-hour, or night 12-hour), and position as their current commitment/FTE, shift, and position as reflected in Attachment C.
 - ii. Current Relief Transfer Center Nurses will be assigned by the Hospital to either a Relief A or Relief B classification after consultation between the Transfer Center management and the affected Relief Transfer Center Nurses. In addition, following consultation with the affected Relief Transfer Center Nurses, the Hospital may assign the Relief Transfer Center Nurses to a 12hour shift assignment.
 - iii. Subject to the foregoing, Regular and Relief Nurses shall be assigned to one of the following shifts:

Day: 6:45 a.m. to 7:15 p.m. Evening: 10:45 a.m. to 11:15 p.m. Night: 6:45 p.m. to 7:15 a.m.

- d. Following the Third Implementation Date, Transfer Center Nurses may have their shift assignments modified pursuant to the terms of the Agreement.
- 5. **PTO, ESL, ATO.** Current Transfer Center Nurses will continue to accrue and may use PTO and ESL under current policies, protocols, and procedures until the Second Implementation Date. As of the Second Implementation Date, current Transfer Center Nurses will maintain

all accrued PTO and ESL. Effective on the Second Implementation date, Transfer Center Nurses will begin to accrue and may use ATO, PTO and ESL in accordance with the Agreement.

- 6. **Benefits.** Current Transfer Center Nurses will maintain their current medical, dental, vision, and other annual benefits plans for the remainder of the 2021 benefits plan year. Beginning in the 2022 benefits plan year, Transfer Center Nurses will have the same annual benefits as other CRONA Nurses, pursuant to Sections 9 and 11 of the Agreement. Transfer Center Nurses will receive the same information and participate in open enrollment on the same terms as other CRONA-represented Nurses for 2022 benefits.
- 7. Educational Assistance. As of the First Implementation Date, Section 14, Educational Assistance and Professional Development will apply to all Transfer Center Nurses. Reimbursements for educational expenses incurred by any current Transfer Center Nurse between September 1, 2021 and the First Implementation Date will be deducted from the reimbursement funds available to her or him under Section 14.3 of the Agreement for the September 1, 2021 to August 30, 2022 academic year.
- 8. **Prior Absences.** Prior unexcused absences or tardies will continue to count for purposes of application of SHC's Attendance Policy.
- 9. **Float-In Only Unit.** As of the Third Implementation Date, the Transfer Center will be added to SHC's Floating Policy as a float-in only unit.
- 10. Pre-Approved Vacation and Education. Pursuant to SHC's Pre-Approved Vacation and Pre-Approved Education Days Policy and the Side Letter Re Attendance and Pre-Approved Vacation and Education Days Policies, Transfer Center management will develop and implement a vacation and education time tracker to determine the maximum number of vacation and education slots available each week. As of the First Implementation Date, that tracker will be used for requests for pre-approved vacation and education days. Existing preapproved vacation and education days for Transfer Center Nurses that were approved before the First Implementation Date, if any, will be honored.
- 11. **Information and Payroll Deduction.** SHC will provide each Transfer Center Nurse an informational packet provided by CRONA within seven (7) days of the Second Implementation Date. Section 5.2 of the Agreement will apply to Transfer Center Nurses, except the timelines that run from the beginning of a Nurse's employment in Section 5.2 will instead run from the Second Implementation Date for current Transfer Center Nurses.
- 12. Job Description and Clinical Performance Criteria. During the term of this Side Letter, the current Transfer Center RN job descriptions shall continue to be used for Clinical Nurses assigned to the Transfer Center, with the exception of the "FLSA Status" and "Bargaining Unit" provisions in the current Regular Transfer Center RN job description. The parties shall discuss modifications to the Transfer Center RN job descriptions during the term of this Side Letter.
- 13. **PNDP.** The PNDP shall be applicable to Transfer Center Nurses, except as follows:
 - a. The Assessment of Clinical Expertise for Transfer Center Nurses shall be tailored for the Transfer Center Nurses as shown in Attachment D; and
 - b. As of the First Implementation Date, the unit transfer provisions as stated in Section XV of the PNDP shall be applicable to Transfer Center Nurses who transferred to the

Transfer Center after September 30, 2021 and, immediately prior to transfer to the Transfer Center, were Clinical Nurse IIIs/IVs.

14. **SIDE LETTER (2019) Re Temporary Relief Position While Pursuing Education.** During the term of this Side Letter, the SIDE LETTER (2019) Re Temporary Relief Position While Pursuing Education shall not be applicable to the Transfer Center.

*Attachments A, B, C, and D to this Side Letter are not printed here but remain a part of this Side Letter. Attachment D is printed in the PNDP Handbook.

SIDE LETTER Between Stanford Health Care and CRONA Re Weekend Premium Pay

Stanford Health Care and CRONA have agreed as follows:

The parties agree that a Nurse who voluntarily agrees to work in a "weekends only" position is not eligible for premium pay under Section 17.2 (titled "Guarantee of Weekends Off") of the parties' collective bargaining agreement. Effective no later than May 1, 2021, the job description for a Clinical Nurse position and a posting for a new "weekends only" position shall state that a Nurse who voluntarily agrees to work in a "weekends only" position is not eligible for premium pay under Section 17.2.

INDEX SHC/CRONA Contract

#

500 Pasteur Dr100 <i>A</i>	1
"ADO" Assignment Despite Objection Form	
Education Days Policies	1

В

Back Up Care	21
Basic Life Insurance	20, 52
Basic Salary	
Benefits	74, 78
Annual Physical Examinations	
Benefit Plan Documents	19
Changes in Employer Provided Benefits	18
COBRA	
Dental Plan	20
Educational Assistance	
Eligibility	18
Legal Care Plan	
Optional Coverage	
Payment of Premiums	
Pre-placement and Annual Physical	
Examinations	31
Retiree Medical Insurance	21
Retirement	29
Vision Plan	20
Bereavement Leave	
Breaks	46

С

Call Back Contiguous With a	
Scheduled Shift	92
Certification Pay	16
Christmas and New Year's Holiday	52
Classification and Compensation Procedures for	
Regular Nurses	6
Classification and Compensation Procedures for	
Training Programs	17
COBRA	
Combination of Shifts (Differentials)	11
Day/Evening or Night	11
Evening/Night	11
Combinations of Leaves of Absences	
Committee Meetings	15
Compensation	8

Compensation and Benefits41
Contiguous Overtime12
Continuing Education Hours
CRONA
CRONA Officers
CRONA Representation
CRONA Security and Dues Deductions
Current Employees
New Employees
Payroll Deductions of CRONA Dues and Service
Fees4

D

Day/Evening or Night Shifts Dental Plan	
Differentials	
Combination of Shifts	11
Evening Shift	11
Night Shift	
Part-time	11
Payment	11
PTO	
Resource Nurse	10
Weekend	11, 90
Discipline	61, 62, 67
After the Trial Period of Employment	62
During Trial Period of Employment	61
Discipline for Cause - Regular Nurses	62
Discipline for Cause - Relief Nurses	62
Discussions with CRONA	
Double Back Premium	13
Duration of Leave	50

Ε

Educational Assistance	
Accumulation	.33
Educational Training and Classes	
Required by the Employer	.34
Eligibility	
Home Study	.34
Paid Educational Hours	.32
Payment for Educational Hours	.34
Procedure	
Purpose	.31
Reimbursement32	, 36
Shift Differential for Required	
Educational Classes	.35
Yearly Basis	.33
Educational Assistance - Relief Nurse	.43
Educational Assistance and Professional	
Enrichment	.31
Educational Training and Classes Required by the	
Employer	.34
Educational Leave	.48
Temporary Relief Position While Pursuing	
Education	101
Eight (8)/Twelve (12) Hour Shifts	.47
Employee Assistance Program	.22

INDEX SHC/CRONA Contract (Continued)

Employer Retirement Plan Employment Commitment and Status End of Established Staffing Patterns Evaluation, Performance Evaluations	40 47
Annual	58
Extension of Trial Period	7
Interim	6
New Hires	6
Peer	
Performance Problems	59
Relief Nurses	37
Evening/Night Shift	
Exclusive Agreement, Amendments, Term of	
Agreement	
Extended Orientation Training Programs	17
Extended Sick Leave (ESL)	28
Accumulation of ESL	28
Eligibility	28
Integration of ESL	
Purpose and Rate of Pay	
Use of ESL	

F

Family Leave	49
Flexible Spending and Health Savings Account	
Programs	22
Floating	13, 44
Not Qualified	44
Outside Region	43
To Different Work Locations	

G

Glossary	88
Grievance and Arbitration	63
Grievances	63
Procedure	63
Step 1 – Informal Discussions	63
Step 2 – Submission of Formal Grievance	64
Content	64
Timeliness	64
Step 3 – Resolution or Referral for Review	64
Step 4 – (Optional) - Mediation	65
Step 5 – Arbitration	65
Government Programs	28
Guarantee of Weekends Off	

Н

Health, Welfare and Safety	72
Hire Date	
Holidays	27
Christmas and New Year's	52
Bundling	53
Unit Equity	75, 91
Home Health Care	
Home Study	34
Hourly Base Wage Rate	

Ι

Insurance

Basic Life	20
Long Term Disability	20
Malpractice and General Liability	22
Retiree Medical	21
State Disability	28
State Unemployment	
Supplemental Accidental Death and	
Dismemberment Insurance	22
Supplemental Group Life	21
Supplemental Long Term Disability	
Supplemental Short Term Disability	
Workers' Compensation	29
Internal Transfers	

J

Joint Conferences	68
Jury Duty and Legal Appearances	53

L

Layoff	75, 77
Leave Categories	48
Bereavement	48
Combinations	51
Educational	48
Medical	49
Military	50
Parental	
Personal	50
Pregnancy Related Disability	50
State/Federal Family and Medical	49
Leaves of Absence - Regular Nurses	48
Legal Care Plan	22
Long Term Disability Insurance	

М

22
93
19
49
49
23
50
38
al
39
Frial
39

Ν

New Hires	6, 38
New Hospital	100
New Twelve (12) and Twelve/Eight (12/8) Hour \$	Shift
Patterns	46
Night Shift	11
No Discrimination	2
No Strikes or Lockout	68
Non-contiguous Overtime	12
Notice of Personnel Guidelines and Procedure	
Changes	68

INDEX SHC/CRONA Contract (Continued)

Nurse Information Centers	58
Nurse Feedback on Benefit Programs	
Nurse Practice Committee	
Nurse Staffing	69

0

Off-call Relief Nurse Official Communications, Requests for	43
Information and Meetings	2
On Call Shift Selection/Assignment Procedures .	
On Call Schedules	
On-Call Systems	
Opportunity to Transfer After Trial Period –	10
Regular Nurses	63
Opportunity to Transfer After Trial Period –	00
Relief Nurses	63
Optional Coverage	
Overtime	
Contiguous	
Double Back Premium	
Mandatory	
Non-Contiguous	
Short Notice	
Overtime and Holiday Pay	
Overtime Compensation	
•	

Ρ

Paid Educational Hours	32
Paid Time Off (PTO)	23
Accrual	23
Cash Out	26
Differential	12
Eligibility	
Purpose and Rate of Pay	23
Use	24
Voluntary PTO Cash Out	27
Parental Leave of Absence (Non-Disability)	50
Parking	
Part-time Nurses	10
Part-time Shifts	
Patient Classification System and Committee	
Payment for Educational Hours	
Payroll Deductions of CRONA Dues and Service	
Fees	
Payroll Procedures	
Performance Evaluation	
Permanent or Indefinite Layoff	77
Personal Leave of Absence	
Placement Alternative to Layoff	
Posting of Schedules	
Preamble	
Pregnancy Related Disability Leave	
Premium Pay	
Pre-placement and Annual Physical Examinatio	ns31
Preceptor Program Differential	
Probation for Non-Trial Period Nurses	
Professional Nurse Development Program	
Program, Course, and/or Exam Qualifications	35
Program, Course, Exam Approval (other than	
Continuing Education Hours)	35

Q

Qualifications for Employment	38
R	
Reclassifications	7
Involuntary - Non-Transfer	
Voluntary	7
Reclassifications of Relief Nurses	39
Recognition and Coverage	1
Reduced Commitments	
Re-Employment	78
Regular Hourly Rate of Pay	
Regular Shifts	46
Rehires - Regular Nurses	7
Rehires - Relief Nurses	
Reinstatement Rights	
Relief Nurse Program	
Relief Nurses	
Repayment of Severance	78
Reserve Rights of Stanford Health Care	
Resource Nurse	
Differentials	
Supervisory Status	
Restricted On-Call	13
Retiree Health Reimbursement Account	
Retiree Medical Insurance	
Retirement Plan	
Return from Leave	
Right to Representation	
Disciplinary Meeting	
Grievance Meetings Investigatory Interview	
Investigatory interview	
S	
Sabbatical Assignment	36
Eligibility and Qualifications	
Purpose	
Selection	
Terms	
Salary Placement	
Part-time Nurses	
Resource Nurse	
Steps	
Salary Review Date	
Scheduling	42
12 hour shifts	
16-Hours	
Cancellations	

 Eight (8)/Twelve (12) Hour
 47

 Twelve (12) Hour Fill-ins
 47

 Seniority
 74

 Seniority Accrual and Application
 74

 Separability and Savings Clause
 81

 Severance Allowance
 77

 Shift Definitions
 46

 Shift Definitions (for Shift Differential Purposes)
 11

 Shifts and Shift Rotation
 46

 Short Notice Compensation
 14

INDEX SHC/CRONA Contract (Continued)

Sleep Room Special Conferences Specialty Continuing Education Staff Meetings	72 36
Staff Reductions and Layoff	75
Staffing	
Staffing Plans	70
Staffing Ratios	70
State and Federal Family and Medical Leave	49
State Disability Insurance (SDI)	
State Unemployment Insurance	29
Supervisory Status of Covered Nurses	82
Supplemental Accidental Death and	
Dismemberment Insurance	22
Supplemental Group Life Insurance	21
Supplemental Long Term Disability Insurance.	
Supplemental Short Term Disability Insurance	

Т

Tax-Deferred Annuity Program	31
Temporary Reassignments/Floating	43
Temporary Relief of Duty Pending Investigation	60
Temporary Staff Reductions	75
Termination for Cause – Regular and Relief	62
Termination Primarily for Unsatisfactory Performance	e
- Regular and Relief	62
Thanksgiving Day	27
Time Off for CRONA Officers	54
Transfers After Completion of Trial Period	7
Transfers to Different Units or Areas	39
Traveler Orientation	16
Tuition Based Specialty Training Programs	18

Twelve (12) Hour Fill-ins	47
U	
Unavoidable Conflicts and Unforeseen Emergencies Unrestricted On-Call	
V	
Vacation and Holiday Time	91
Vacations Off-call for Relief Staff	43
Shortage of PTO Due to Bereavement	48
Two Weeks Guaranteed	
Unit Equity	75, 91
Vision Plan	20
Voluntary Commitment - Floating Outside	
Region	43
Voluntary Requested Reduction in Scheduled	
Commitment	56

W

Wage Schedule	84
Weekend Differential	
Weekend Staffing	
Work Day	88
Work Outside Nurse Classifications	
Workers' Compensation	29
Working Leave Status	51
Written Warning (excludes other	
Disciplinary Notices)	59
Written Warnings and Discipline	59